# HIGH PERFORMANCE WORK PRACTICES IN SMALL AND MEDIUM-SIZED FIRMS

# NING LIU WU

Thesis submitted to the University of Nottingham for the degree of Doctor of Philosophy

**JULY 2011** 

# **Abstract**

Drawing on data from Workplace Employment Relations Survey 2004, this thesis assesses the uptake of High Performance Work Practices (HPWPs), the factors associated with the uptake of HPWPs and the relationship between HPWPs and performance (also termed the "HPWP-performance link") in small and medium-sized firms.

The findings show that medium-sized firms have a higher uptake of HPWPs compared to small firms in general. They also show that the extent of the use of HPWPs in small firms tends to be influenced more by internal than external factors, while the opposite holds true for medium-sized firms.

Although the widely reported HPWP-performance link holds in large firms, the findings suggest only a specific bundle of HPWPs seeking to develop opportunities for employees to participate in management decision making and use their skills and abilities at work are related to improved financial performance in small firms. However, neither the overall use of HPWPs nor any specific bundles of practices are associated with better performance in medium-sized firms. In medium-sized firms, some HPWPs are positively associated with performance and some others are negatively associated with performance. These findings suggest the HPWP-performance link in medium-sized firms is distinct from that in small and large firms.

Overall, the findings suggest small and medium-sized firms should be analyzed as two distinct groups and the HPWP-performance link is not universal. The lack of a consistent HPWP-performance link in medium-sized firms has important implications for HRM theory, the HR advice that medium-sized firms should be offered, and government support and employment policies targeted at medium-sized firms.

# Acknowledgements

I would like to express my gratitude to Professor Nick Bacon and Professor Kim Hoque for their constructive supervision, guidance and generous support throughout my PhD research study and the preparation of this thesis. Thank you also to Professor Cees van der Eijk at the Method and Data Institute for his instructive advice and encouragement.

I would like to pay tribute to Professor Mike Rose, who was the co-ordinator of the Work Attitudes Measurement Network, for introducing me to a series of seminars/mini-conferences on a range of debates and theories relating to issues concerning secondary survey data analysis.

On a more personal level, I would like to say thank you to my husband and my parents; without whose unwavering and unyielding support and encouragement this PhD research study would not have been possible.

Finally, I would also like to thank the WERS sponsors- the Department of Trade and Industry, the Economic and Social Research Council, the Advisory, Conciliation and Arbitration Service and the Policy Studies Institute as the originators of the WERS 2004 data. None of these organisations are responsible for any of the findings or claims made in this thesis.

# Contents

Abstract		2
Acknowled	dgements	3
Contents		i
List of Tal	bles	iv
Chapter 1	Introduction	1
1.1	Research aims	2
1.2	Thesis layout	3
Chapter 2	HPWPs and performance in small and medium-sized firms	6
2.1	Introduction	6
2.2	The use of HPWPs in small and medium-sized firms	9
2.2.1	The use of HPWPs in small firms	9
2.2.2	The use of HPWPs in medium-sized firms	11
2.3	Contextual factors associated with the use of HPWPs in small and med	dium-sized
	firms	13
2.3.1	External factors	14
2.3.2	Internal factors	16
2.4	The HPWP-performance link in small and medium-sized firms	18
2.4.1	The HPWP-performance link in small firms	18
2.4.2	The HPWP-performance link in medium-sized firms	21
2.4.3	High performance work practices in small and medium-sized firms	22
2.5	Discussion and conclusions	43
Chapter 3	The uptake of HPWPs in small and medium-sized firms	46
3.1	Introduction	46
3.2	Research method	47
3.2.1	Data/sample	47
3.2.2	Dependent variables: HPWPs	48

3.2.3	Independent variables: small, medium-sized and large firms53
3.2.4	Control variables
3.2.5	Weights
3.3 R	esults57
3.3.1	Assessing the uptake of HPWPs in small, medium and large firms – Chi <sup>2</sup> tests57
3.3.2	Comparing the uptake of HPWPs between small, medium and large firms using
	maximum likelihood multi-variate regression
3.3.3	Assessing the firm size effect on the uptake of HPWPs in small, medium and
	large firms61
3.4 L	Discussion and conclusions64
Chapter 4	The antecedents of the uptake of HPWPs in small and medium-sized firms67
4.1 In	ntroduction67
4.2 R	esearch method68
4.2.1	Dependent variable: HPWP intensity
4.2.2	Independent variables: contextual factors
4.2.3	Control variables
4.3 R	esults71
4.3.1	The relationship between external factors and the uptake of HPWPs in small,
	medium and large firms71
4.3.2	The relationship between internal factors and the uptake of HPWPs in small,
	medium and large firms73
4.4 D	Piscussion and conclusions75
Chapter 5	The relationship between HPWPs and performance in small and medium-
	sized firms79
5.1 Ir	ntroduction79
5.2 R	esearch method80
5.2.1	Dependent variables: performance measures
5.2.2	Independent variables: individual HPWPs, HPWP bundles and HPWP intensity
	82

5.2.	3 Control variables
5.3	Results84
5.3.	1 Assessing the relationship between individual HPWPs and performance in small,
	medium and large firms84
5.3.	2 Assessing the relationship between HPWP bundles and performance in small,
	medium and large firms90
5.3	3 Assessing the relationship between HPWP intensity and performance in small,
	medium and large firms95
5.4	Discussion and conclusions97
Chapter	6 Conclusions 105
6.1	The difference in the uptake of HPWPs between small and medium-sized firms105
6.2	The difference in the antecedents of the uptake of HPWPs between small and
	medium-sized firms
6.3	The difference in the HPWP-performance link between small and medium-sized
	firms112
6.4	Analytically splitting small and medium-sized firms as two distinct groups117
Bibliogr	aphy120
Appendi	x 1 Variable means140

# **List of Tables**

Table 2.1 H	PWPs used in studies on the relationship between HPWPs and
pe	erformance
Table 2.2	PWP "bundles" in selected studies
Table 3.1 Th	he uptake of HPWPs by firm size in the private sector
Table 3.2 Co	comparing the uptake of HPWPs between small, medium and large firms. 60
Table 3.3 Co	comparing the uptake of HPWPs between subcategories of firm size 63
Table 4.1 Th	he association between contextual factors and the uptake of HPWPs in
sn	mall, medium and large firms
Table 5.1 Th	he relationship between individual HPWPs and performance in small,
m	nedium and large firms
Table 5.2 Th	he relationship between HPWP bundles and performance in small, medium
an	nd large firms
Table 5.3 Th	he relationship between internal fit and performance in small, medium and
laı	irge firms
Table 5.4 Th	he relationship between HPWP intensity and performance in small, medium
an	nd large firms

# Chapter 1 Introduction

In the United Kingdom, small and medium-sized enterprises (SMEs) accounted for 99.9 per cent of all enterprises in 2008, 59.4 per cent of private sector employment and 50.01 per cent of private sector turnover (referring to the value of sales, work done and services rendered) (BERR, 2009). These figures clearly demonstrate the size and economic importance of the SME sector.

However, in comparison with the extensive literature exploring the relationship between high performance work practices (HPWPs) and performance in large firms (Huselid, 1995; McDuffie, 1995), relatively few studies have explored the impact of such practices in small and medium-sized firms. Given this, the relationship between HPWPs and performance in small and medium-sized firms is yet to be established.

A number of arguments might be made as to why one might anticipate a link between HPWPs and performance in small and medium-sized firms. The universalist argument, for example, states that 'the use of high-performance work practices should lead to positive outcomes for all types of firms' (Huselid, 1995: 644). This implies that the HPWP-performance link should hold in small and medium-sized firms as well as large firms.

There is considerable debate in the literature, however, over the uptake and impact of HPWPs in the SME sector. On the one hand, Sisson (1993) argues that SMEs use formal practices such as HPWPs very rarely and even in circumstances they do, they use them in an *ad hoc* way. Where the HPWP-performance link is concerned, a number of studies have failed to identify a positive relationship (see, for example: Storey, 1998; 2002; 2004).

On the other hand, a number of studies have suggested that small and medium-sized firms are using more HPWPs than is frequently suggested (Bacon et al., 1996; Forth et al., 2006) and some studies also report that HPWPs are associated with improved performance in small firms (de Grip and Sieben, 2005; Faems et al., 2005; Sels et al., 2006; Way, 2002). However,

although these studies appear to suggest the relationship between HPWPs and performance holds in small firms as well as large firms, variations in the definitions of small firms make overall conclusions difficult to reach.

A further difficulty in relation to the HPWP-performance link in the SME sector relates to the widespread practice of combining small and medium-sized firms into a single unit of analysis. This leads to an assumption that small and medium-sized firms are a homogeneous group. However, a few recent studies report that the nature of human resource management (HRM) in medium-sized firms could differ from small firms (Baron et al., 2007; Hoque and Bacon, 2008). Combining small firms and medium-sized firms together therefore could obscure the possibility that the HPWPs that potentially affect performance in medium-sized firms might be different from those that affect performance in small firms. This thesis, therefore, analytically divides small and medium-sized firms into two distinct groups and explores the HPWP-performance link in each firm sector independently.

## 1.1 Research aims

There are three research aims in this thesis. The first aim is to assess whether small and medium-sized firms use formal practices such as HPWPs, and if so, whether the uptake of HPWPs in small firms differs from that in medium-sized firms. The second aim of this thesis is to evaluate whether the factors associated with the uptake of HPWPs in small firms are different from those in medium-sized firms. Thirdly, this thesis aims to explore whether there is a relationship between HPWPs and performance in small and medium-sized firms and if so, whether the nature of the HPWP-performance link in medium-sized firms differs from that in small firms.

To conduct these assessments, this thesis draws on data from the Workplace Employment Relations Survey (WERS) 2004 (DTI, 2005). This is the first WERS that includes workplaces

with as few as five employees, which allows this thesis to explore the HPWP-performance link within a more representative sample of small firms.

### 1.2 Thesis layout

This thesis is structured as follows. Chapter 2 reviews debates on the use of HPWPs, the factors that influence the use of HPWPs and the HPWP-performance link in small and medium-sized firms. Chapter 3 introduces the WERS 2004 data to be used in the study, and draws on these data to evaluate the extent of and variation in the uptake of HPWPs between small, medium and large firms both within a bivariate analysis and also a multivariate analysis that controls for observable workplace characteristics that might affect this relationship. Chapter 4 assesses the extent to which a range of factors, both internal and external to the firm are associated with the uptake of HPWPs in small and medium-sized firms and examines whether different factors influence the uptake of HPWPs in small firms than in medium-sized firms. Chapter 5 explores the relationship between HPWPs and performance in small and medium-sized firms. The tests assess whether the HPWP-performance link holds in small and medium-sized firms, and if so, whether the link in medium-sized firms is different from that in small firms. Chapter 6 summarises the findings and discusses the contribution the results of the analysis make to knowledge with regard to HPWPs in the SME sector.

Within the thesis, the use of HPWPs is assessed from three perspectives: individual HPWPs; sets of closely-defined HPWPs (termed "HPWP bundles"); and the total number of HPWPs adopted (termed "HPWP intensity"). Each of the three approaches has been used in previous studies, with some focusing on specific HPWPs such as training (Storey, 2004; Storey and Westhead, 1997), others relying on the count measure of HPWP intensity, (de Kok and Uhlaner, 2001; Sels et al., 2006; Way, 2002), and others (Faems et al., 2005) looking at specific HRM domains<sup>1</sup> to assess the HPWP-performance link. None of the existing studies,

<sup>&</sup>lt;sup>1</sup> Faems et al. (2005) addressed six domains, which are selection, training, careers, compensation, performance management and participation, with each domain consisting of 3

however, have conducted an analysis from all three perspectives. Using all three approaches allows for an exploration of the HPWP-performance link in small and medium-sized firms from a number of different perspectives.

A particular focus within this thesis is to seek to address the use and effects of HPWPs in small and medium-sized firms separately. Combining the SME sector as a whole into a single category and assuming homogeneity between SMEs risks overlooking the possibility that there might be considerable variation within the sector. Hence, splitting small and medium-sized firms into two independent groups provides an opportunity to test whether the theory and principles of HRM developed largely in large firms is equally applicable in small and/or medium-sized firms. This has potentially important implications for government policy, particularly if the analysis suggests that policy initiatives need to differentiate between small and medium-sized firms in terms of the advice, guidance and support they are provided with. Successive UK governments have sponsored a number of initiatives to promote the awareness and use of practices such as HPWPs by encouraging SMEs to participate in advisory networks such as employers' associations, local chambers of commerce and Learning and Skills Councils (LSCs). The 'Ambition of Enterprise' strategy, for example, involves 'an increase in the proportion of SMEs making use of external business advice' (BIS, 2008:10). If the analysis conducted here finds differences in the antecedents of high performance in small firms and medium-sized firms, this will suggest that the advice provided through these networks needs to take into account the possibility that firms in these two size categories require very different advice and guidance. In particular, it would only be advisable to continue to encourage the adoption of HPWPs if it can be shown in the first instance that these practices are genuinely associated with higher performance.

It is worth mentioning here that estimates will be provided of the uptake and effects of HPWPs in large firms, for the purpose of comparison. HRM theory has been largely developed in the context of large firms. Hence, any discrepancies between small and/or medium-sized firms and

HRM practices identified by the authors and firms were ranked in line with the overall score of

large firms (as a comparison benchmark) will help to demonstrate whether or not HRM theory is universally applicable across all firm size categories.

The analysis conducted in this thesis may be of particular importance to government policies in the SME sector given the current severe economic climate. The British government recognises, for example, that access to finance is critical for businesses to survive and grow and that small and medium-sized companies face particular challenges. The government has recently produced a report on how business and the government can work together to produce a diverse, competitive and sustainable financial environment to financing a private sector recovery. Among the initiatives advocated, the report states that the government will continue 'expanding the range of support available for SMEs, including alternative sources of finance extending existing schemes that provide finance for viable small businesses; and helping banks and businesses work together more effectively' (BIS, 2010: 3-4). With specific regard to training in SMEs, the government 'will also increase its investment in the Leadership and Management Programme for SMEs from £4 million to £30 million per annum. Eligibility will be extended from April 2008 to SMEs with between ten and 250 employees (currently 20 to 250 employees)' (BIS, 2008: 45). It would only be advisable to continue to advocate and fund an increase in training (and other HPWPs) in SMEs if it can be demonstrated in the first instance that they are genuinely associated with improved performance in both small and medium-sized firms.

# Chapter 2 HPWPs and performance in small and mediumsized firms

## 2.1 Introduction

While a large body of evidence suggests that high performance work practices (HPWPs) are associated with better performance in large firms, there is still a dearth of similar studies in small and medium-sized firms. Calls to investigate the nature of human resource management (HRM) in the small and medium-sized enterprise (SME) sector suggest a need to test whether the principles and theory of HRM, developed largely in large firms, are also relevant to the small and medium-sized firm sector (Baron, 2003; Heneman et al., 2000). Several authors suggest HPWPs may be inappropriate in small and/or medium-sized firms (Marlow, 2006; Storey, 1998; 2002; 2004). However, there is some evidence that SMEs are indeed adopting HPWPs (Bacon et al., 1996; Forth et al., 2006) and that the use of such practices is also associated with better performance in small and medium-sized firms as well as in large firms. Although the number of studies of this kind is limited (see, for example, Faems et al., 2005; Sels et al., 2006), these studies suggest small and medium-sized firms may improve their performance by adopting HPWPs and that HPWPs for large firms are equally applicable for SMEs.

Such findings would seem to correspond with the universal notion that HPWPs will improve firm performance irrespective of context (see, for example, Huselid, 1995). Exemplified by Pfeffer's (1994) 16 practices, studies have drawn on various lists of HPWPs and demonstrated that the adoption of these practices is associated with higher levels of organisational performance. Among these studies, three terms are often used: high commitment management (HCM); high involvement management (HIM); and high performance work systems (HPWS). Authors addressing HCM argue that these practices can create the conditions for employees to display internally self-driven initiative and to take responsibility for self-regulating their own

behaviour (Walton, 1985). Given that a highly committed workforce is expected to have a lower absence rate, a higher labour retention rate, improved labour productivity and to deliver better quality of products or services. This will ultimately lead to higher firm performance (Guest, 1987).

By contrast, HIM emphasises job enrichment and/or self-managing teams that give individuals considerable autonomy to control and monitor their own work (Guthrie, 2001; Lawler, 1992). Practices such as problem-solving groups are seen as generating high involvement, and this in turn fosters employee self-management and helps to reduce layers of bureaucracy and expensive monitoring and control processes, thereby improving firm performance.

Beyond this, HPWS relates to how the organisational structure influences the employment structure and affects firm performance (Appelbaum et al., 2000). HPWS comprises two elements: first, production systems that focus on how work is organised; and secondly, the employment structures that provide the environment within which these production systems operate. Based on the premise that labour is a unique factor of production which yields the potential for efficiency gains that competitors find difficult to imitate, firms will seek to adopt new HRM techniques in order to secure competitive advantage.

Despite the fact that authors use different terms, the overlaps between these terms regarding the practices firms should adopt to improve performance are apparent. Teamwork, for example, is used as a tool in HCM to up-skill the workforce, provide work autonomy and elicit high commitment. HIM introduces teamwork as an approach to encourage workers to identify with organisational goals and utilise worker knowledge to bring about organisational improvement. HPWS views teamwork as crucial for enhanced performance as it allows firms to alter their organisational structure subject to the changes in the external market and also to enable quality to be maintained at a higher level. More importantly, HCM, HIM and HPWS emphasise performance effectiveness. This thesis chooses to use the term 'high performance work practices' (HPWPs) for three reasons. First, the term HPWPs explicitly addresses the essence of studies of HPWP-performance link, i.e. performance impact of HRM practices. Second, the

term HPWPs represents the universalist stance upheld by several authors (see, for example, Huselid, 1995), and the purpose of this thesis is to test whether the HPWP-performance link is universally applicable to small and medium-sized firms<sup>2</sup>. Third, the selection of the 18 practices is based on studies that have shown these practices, either individually or used in combination, were associated with improved performance.

Another growing consensus among the studies of the HPWP-performance link also notes that coherent HPWPs used in bundle can create synergistic effect and generate a greater impact on performance compared to these practices used in isolation (MacDuffie, 1995). Beyond this, Delery (1998: 294) argues that using HPWPs in bundles should avoid "deadly combination" when HRM policies or practices work in opposite directions such as strong training for teamwork but performance appraisal only rewards highly individualistic behaviours. In this sense, Boxall and Purcell (2000) argue that implementing many HPWPs simultaneously should ensure the internal consistency between these bundles (termed "internal fit") as well as between the HRM policies and HPWPs adopted. The operation of HPWPs is, therefore, a process that 'involves some tension among competing objectives in management and inevitably implies tensions among competing interests' (p. 188).

As such, this thesis reviews the aforementioned arguments and develops hypotheses concerning a) whether small and medium-sized firms use formal practices such as HPWPs as large firms do, if so, whether the extent of the uptake of such practices in small firms differs from that in medium-sized firms; b) what factors can affect the use of HPWPs in small and

-

<sup>&</sup>lt;sup>2</sup> This thesis has omitted a review of debates on contingency theory in relation to whether business strategy moderates the relationship between HPWPs and performance in small and medium-sized firms. This is because questions designed for business strategies in WERS 2004, i.e. price, quality and innovation strategies, do not generate sufficient number of observations to conduct a meaningful analysis of these issues. The three questions directly related to the business strategy in WERS 2004, i.e. Kprice for price strategy, Kqual for quality strategy, Krank for innovation strategy, which were also used as the base questions for such strategies in a few other studies (see for example, Michie et al, 2008), produce figures for each strategy as follows: 700 observations for quality strategy, 141 for price strategy and 600 for innovation strategy across all firm sectors in the private sector. This approach generates fewer than 20 observations for small firms using a price strategy. This figure reduces sharply after standardisation and thus makes it impossible to generate reliable estimates. Moreover, with over 80 percent of small firms claiming to be quality enhancers, the validity of the corresponding questions become somewhat doubtful in light of the evidence of poor performance within the sector (Storey, 1994). Hence, the plan to address the "best fit" issue in small and medium-sized firms was eventually abandoned.

medium-sized firms and whether the factors associated with the uptake of such practices in small firms differs from those in medium-sized firms; and c) whether the uptake of HPWPs is associated with performance in small and/or medium-sized firms, if so, whether the pattern of the HPWP-performance link in small firms differs from that in medium-sized firms.

This chapter is structured as follows. Analytically splitting small and medium-sized firms into two distinct groups, the next section (2.2) reviews debates on the uptake of HPWPs in small and medium-sized firms; the third section (2.3) describes a range of contextual factors that may influence management decision-making concerning the uptake of HPWPs in small and medium-sized firms; the fourth section (2.4) considers debates on the HPWP-performance link in small and medium-sized firms. The final section (2.5) summarises these debates and the hypotheses developed for further testing.

# 2.2 The use of HPWPs in small and medium-sized firms

This section reviews debates on the uptake of HPWPs in small and medium-sized firms respectively.

## 2.2.1 The use of HPWPs in small firms

It has been generally acknowledged that the uptake of HPWPs in small firms is much lower than in large firms (Marlow, 2006; Sisson, 1993; Wilkinson, 1999). There are a number of possible reasons for this. First, the resource-poverty argument indicates that small firms are unlikely to be able to afford the investment in HPWPs (Storey and Westhead, 1997). Second, the cost of investing in HPWPs is much higher in small firms than in large firms as aggregate costs incurred from implementing such practices in large firms can be distributed across a larger workforce and thus economies of scale can be achieved (Storey, 2002; Storey and Westhead, 1997). It is also argued that internal labour markets in large firms facilitate employee retention and increases the chance of return on HPWPs investment (Green, 1993).

Third, a lean firm structure and frequent face-to-face communication may make some formal procedures unnecessary in small firms. Given this, HPWPs in small firms as formal practices are often viewed as unnecessarily bureaucratic (Gibb, 2000). Fourth, small firms may be acting rationally in not investing in HPWPs in the absence of evidence demonstrating that investment in such practices improves business performance. As discussed earlier, Storey (2004) for example reports that greater investment in training does not improve performance in small firms.

Notwithstanding these arguments, several studies provide evidence of significant increases in the uptake of HPWPs in small firms in recent times (Bacon et al., 1996; Forth et al., 2006), with some authors arguing that the use of such practices in small firms is very similar to that in large firms (Golhar and Deshpande, 1997). Bacon et al. (1996), for example, report that a growing number of small firm owners are aware of HPWPs (termed as "new management practices" in their research) and a higher uptake of HPWPs has emerged as a result. Such an increase in the uptake of HPWPs in small firms may be due partly to an increased awareness of such practices as a result of government policies and government-sponsored advisory initiatives aimed at the SME sector (Goss et al., 1994). However, one might still argue that there will be a greater demand for formal practices in large firms given the greater likelihood of co-ordination problems (Blau, 1970). Nevertheless, it would appear that the uptake of HPWPs in the SME sector may no longer be as limited as argued by several authors (Marlow, 2006; Sisson, 1993; Wilkinson, 1999).

It is possible, however, that there is scope for considerable variation within the SME sector. A threshold for formality is conjectured in Roberts et al.'s (1992) study, in which they argue that it is crucial for small firms with 20 or more employees to introduce sufficient formal practices as beyond this number the informality that tends to dominate firms with fewer than 20 employees will no longer be sufficient to accommodate increasing levels of complexity. Kotey and Slade (2005) also note that as size expands, the adoption of such practices increases (at first rapidly but then more slowly). This has two implications. The first is that a considerable difference in the extent of the uptake of HPWPs may exist between very small firms and other

SMEs. The second is that differences between medium-sized firms and large firms in terms of the uptake of HPWPs might not be particularly great. This will be assessed in Chapter 3. Following on from this, if there has been a growth in the use of individual HPWPs in small firms, it might also be the case that the adoption of specific bundle(s) of closely-defined HPWPs used in combination is more widespread in small firms than previously appreciated.

Given the above, this thesis will assess the extent to which the use of HPWPs differs between small, medium and large firms. It will also assess whether formality starts at a particular size threshold.

#### 2.2.2 The use of HPWPs in medium-sized firms

Compared with the literature on the use of HPWPs in small firms, large firms or SMEs in general, studies focusing specifically on the uptake of HPWPs in medium-sized firms are rare. As alluded to above, although the widespread use of the term 'SMEs' seems to imply that small and medium-sized firms form a homogeneous group, it is quite possible that small and medium-sized firms have different HRM concerns and priorities, and will therefore adopt different HPWPs.

However, the limited research conducted on the extent to which medium-sized firms differ from small firms is inconclusive. Some of the literature points to similarities: for example, Storey (2002) argues that, as with small firms (Storey, 2004), medium-sized firms are less likely to adopt HPWPs such as training than are large firms, given that both small and medium-sized firms will view trained employees resigning from the firm as an unacceptable waste of resources. Examining HPWPs more broadly, existing debates tend to regard both small and medium-sized firms as suffering equally from resource constraints and therefore equally unable to afford expensive and complex HPWPs.

Against this, however, Kotey and Slade (2005) argue that medium-sized firms will have a greater tendency to introduce formal HPWPs than small firms given that, in comparison with

small firms, medium-sized firms have a greater division of labour, more hierarchical structures and require increased documentation and better developed administrative processes as a result of their larger workforce size. Also, Lynch and Black (1998) argue that while the economies of scale enjoyed by medium-sized firms are less than those enjoyed by larger firms, they nevertheless have greater economies of scale than small firms and therefore may be more inclined to engage in practices such as workforce training and development. Furthermore, medium-sized firms are more likely to have internal labour markets, thereby increasing employee retention and the chance of a return on training investments (Green, 1993).

Additionally, specific external pressures on medium-sized firms may also make it more likely that they will adopt HPWPs than small firms. As Frormann (2006) argues, medium-sized firms tend to view the development of formal practices as necessary to demonstrate enhanced reliability and accountability, as this is considered essential to increase their ability to obtain finance from venture capitalists, investment bankers and stock analysts. Given this, medium-sized firms may also hire chief executive officers (CEOs) with greater business experience who introduce formal rules as part of building a professional business (Baron et al., 2007).

The limited empirical evidence on the practices adopted by small and medium-sized firms also suggests differences may exist. Drawing on data from WERS 2004, Hoque and Bacon (2008) find overall levels of training to be higher in medium-sized firms than in small firms, and no lower than in large firms, thereby suggesting support for Lynch and Black's (1998) argument that small firms will view training as too costly to the organisation whereas medium-sized firms may not. Beyond this, a number of studies, while not directly evaluating differences between small and medium-sized firms, have identified the importance of size as a predictor of the uptake of HPWPs. For example, Forth et al. (2006) find the uptake of specific HPWPs to be correlated positively with firm size, and Kotey and Slade (2005) find variation within the small firm sector by size. As these studies point to the importance of firm size as a predictor of the approach taken to HRM, one might reasonably hypothesize that the uptake of HPWPs will be higher in medium-sized firms than in small firms.

# 2.3 Contextual factors associated with the use of HPWPs in small and medium-sized firms

Turning to the factors associated with the uptake of HPWPs in SMEs, a number of factors might be important. For example, successive government sponsored initiatives may have helped to bring awareness of HPWPs to small and medium-sized firms and thereby enhanced the uptake of such practices as a result. A further aim of the thesis is to identify what factors are likely to be associated with the uptake HPWPs in small and medium-sized firms, and also to indentify whether the factors associated with the use of HPWPs in small firms are different from those in medium-sized firms.

In considering these issues, it is worth keeping in mind Rainnie's (1989) argument that the use of HPWPs in SMEs can be largely externally imposed. Given this, Hendry et al. (1995) emphasise the importance of giving due attention to both external and internal factors in considering how small and medium-sized firms adapt and change. Hence, the factors that are potentially associated with the adoption of HPWPs in small and medium-sized firms are assessed from two perspectives: first, the external perspective, which emphasizes the impact of competition in the external market, government policy, employers' associations/networks and the presence of the largest customer; and second, the internal perspective, which focuses on the impact of factors such as firm size, firm age, family ownership, a single independent workplace, union recognition and the presence of an HR specialist. The analysis will also seek to identify whether the factors associated with the uptake of HPWPs in small firms are different from those in medium-sized firms, and thus shed further light on whether small and medium-sized firms should be viewed as distinct firm sectors. The next sections explore the external and internal factors that might be associated with the take up of HPWPs in SMEs in further detail.

#### 2.3.1 External factors

In understanding the factors that have the potential to influence the uptake of HPWPs in SMEs, it is essential to recognise potential pressures exerted by external factors. One such factor is product market competition, with SMEs being more vulnerable to a highly competitive market than is the case for larger firms (Edelman et al., 2005). The extent to which external factors<sup>3</sup> are likely to be associated with the uptake of HPWPs and whether such associations are likely to vary between small and medium-sized firms is discussed in detail below.

First, as mentioned above, highly competitive market may be particularly punishing for small and medium-sized firms than large firms, given that the latter may find it easier to access alternative markets (Storey, 1994). High competition can drive down price and reduce the finance available for HR investments, thus reducing the likelihood that HPWPs will be adopted. That said, high competition can also drive firms to identify and adopt the most efficient practices, thus resulting in new HPWPs being introduced. One might also expect the impact of market competition to vary between small and medium-sized firms. Arguably, the impact of market competition will be greater in small firms than in medium-sized firms, given the latter may have more resources at their disposal (Curran, 1996; Hill and Stewart, 1999).

Second, the provision of specific HPWPs may be affected by the relationship between SMEs and their larger customers. A quarter of SMEs in Hunter et al.'s (1996:245) research, for example, had changed their workforce training in response to their larger customers' requirements. Large customers also assessed and audited their SME suppliers by inspecting training records, budgets and plans to ensure that both management and workforce understood quality control procedures (p.245). Hence, one might expect a greater use of HPWPs to emerge in SMEs in instances where they have a large, dominant customer.

3

<sup>&</sup>lt;sup>3</sup>Several external factors that were reported to have an impact on the use of HPWPs in SMEs in previous studies are not included in this thesis for various reasons. For example, being part of a franchise chain – associated with a greater use of HPWPs (Hoque, 2000) – is dropped in the analysis due to the small number of observations in the sample. Though much of the literature argues that the use of HPWPs is strongly influenced by the type of business strategy adopted, this is not addressed here because the questions asked about in relation to business strategies in the WERS 2004 cannot produce a valid typology. For further details, see the discussion of this in Footnote 2 earlier in the chapter.

Against this, however, large customers may be in a position of market power over SME suppliers, thus enabling them to drive down prices. This in turn may result in SMEs having to adopt authoritarian management, tight supervision and a low wage strategy (Rainnie, 1989). Using WERS 1998, however, Hoque and Bacon (2006) reported there is no link between the presence of larger customers and the extent of uptake of training in SMEs. This finding may, however, mask considerable variation within the SME sector. In particular, what remains to be seen however is whether the impact of large, dominant customers is greater in small firms than in medium-sized firms. One might expect small firms to be more likely to be dominated by a single large customer than is the case for medium-sized firms. Also a medium-sized firm may have more resources at its disposal to be able to seek out new markets should a large customer seek to drive down prices. A further aim of this thesis is to address this issue.

Third, government-sponsored initiatives can exert an impact on the extent of the uptake of specific HPWPs. For example, Investors in People (IiP) recognition may provide a best practice blueprint for SMEs to work towards, hence one might anticipate the uptake of HPWPs to be higher in SMEs that have secured recognition than in those that have not (Goss et al., 1994). However, the evidence indicates that IiP is not associated with one key HPWP (workforce training) in SMEs (Hoque and Bacon, 2006). One possible explanation for this is that if SMEs have been compelled to seek IiP (as a requirement from a larger customer, for example) and they may view IiP accreditation as a marketing tool or as a procedural requirement that has no subsequent impact on employment practices once recognition has been secured (Hoque and Bacon, 2006; Ram, 2000). Thus, it remains unclear whether the IiP award will be associated with the overall uptake of HPWPs in either small or medium-sized firms.

Fourth, the UK government has encouraged the provision of business advice through business support networks such as employers' associations, local chambers of commerce and Learning and Skills Councils (LSCs). Hoque and Bacon (2006) found that SMEs joining three or more different employers' associations were more likely to provide training to their employees than those who were not a member of any such associations. However, a few studies report that

being a member of various business networks is only weakly associated with the use of management advice supported by government (Bennett and Robson, 2003). This may be due to the fact that SME managers are extremely reluctant to accept external advice given the imperative of survival over growth and their strong commitment to independence (Curran and Blackburn, 2000). Hence, it is not clear whether the extent of the uptake of HPWPs in small and medium-sized firms will be affected by membership of employers' associations, and if so, whether the impact of such associations is different in small firms than in medium-sized firms.

#### 2.3.2 Internal factors

A large body of literature reports that the extent of the uptake of HPWPs is influenced by a range of organisational factors within the firm such as firm size, the history of the firm, union recognition and industry sector (see, for example, Guthrie, 2001). Studies of HPWPs in SMEs also note that a range of further factors such as family ownership, the presence of HR specialists and whether an SME is a stand-alone establishment or is multi-sited predicts the use of HPWPs (Hoque and Bacon, 2006; Wager, 1998). This section reviews these factors in detail below.

First, it has been argued that it is less likely for family-owned small firms to use HPWPs because family-owned firms more often opt for personal networks rather than formal practices (Ram and Edwards, 2003). Empirical evidence also lends support to this argument (Hoque and Bacon, 2006). Given the majority of small firms are family-owned (Forth et al., 2006), this is likely to be a particularly important factor where small firms as opposed to medium-sized firms are concerned. Nevertheless, the extant literature frequently discusses the impact of family ownership on the use of HPWPs in SMEs as a whole (see, for example, Kotey and Slade, 2005). Hence, although medium-sized firms might be less likely to be family owned, one might still anticipate that family-owned medium-sized firms will be less likely to use HPWPs than non-family owned medium-sized firms.

Second, small firms are well known for their short lifecycles and high failure rates. A short lifecycle may in part explain the low uptake of HPWPs in small firms because such practices may be unlikely to lead to performance given the short time periods in which they are in operation (Storey and Westhead, 1997). Empirical studies report that small firms that have been in operation for many years tend to have more HPWPs in place compared to those with a short operation history (Faems et al., 2005; Wager, 1998). It would therefore be expected that the longer a small firm stays in business, the higher the chance of the firm introducing more practices. A positive relationship between the use of HPWPs and firm age would also be expected to emerge in medium-sized firms as the benefits from straying in business longer are likely to encourage such firms to invest in HPWPs (Faems et al., 2005).

Third, multi-sited firms may tend to install more formalised procedures to minimise the principal-agent problem (Knocke and Kalleberg, 1994). Findings based upon the 1998 WERS data revealed that small firms with multiple workplaces make greater use of formal procedures for employment relations issues (Cully et al., 1998). With a higher proportion of medium-sized firms being multi-sited, a greater adoption of HPWPs may also be anticipated in these firms than in single-sited medium-sized firms.

Fourth, the low uptake of HPWPs in small and medium-sized firms may be associated with a lack of HR specialists (Boxall and Purcell, 2008). Empirical research has reported that the presence of HR/personnel specialists is positively related to training in the SME sector (Hoque and Bacon, 2006), with HR experts being able to bring the knowledge and expertise a firm needs to enable it to make the most appropriate HR decisions. Empirical evidence also reveals that small firms/SMEs are less likely to employ a dedicated HR specialist (Cully et al., 1996; Forth et al., 2006). Nonetheless, while HR specialists are considerably less prevalent in small firms and/or medium-sized firms, in the instances where they are present, they may well be associated with a greater use of HPWPs.

Fifth, firm size (de Kok and Uhlaner, 2001; Forth et al., 2006) has been widely reported to be positively correlated with the extent of adoption of HPWPs as the cost of HRM activity can be

distributed across a large workforce in a large firm (Storey and Westhead, 1997). A greater use of these practices may occur in small and medium-sized firms that have a relatively large workforce size.

Sixth, union recognition has been found to be strongly associated with formal performance appraisals, higher compensation and more training to new recruits (Jackson et al., 1989). With specific regard to SMEs, Green (1993) reported that the level of training activity was higher in workplaces with union recognition. Both Jackson et al.'s (1989) and Green's (1993) findings suggest that unions may bring benefits to small and medium-sized firms in terms of encouraging the adoption of a more sophisticated approach to HPWPs. If a greater use of HPWPs is found in unionised small firms, one might argue that this suggests the threshold for statutory union recognition in the Employment Relations Act 1999 should be extended to small firms employing fewer than 21 workers (Ewing and Hock, 2003).

To summarize, one might anticipate that a range of factors, both internal and external to the firm might be associated with the adoption of HPWPs in SMEs, and that these factors might have different effects in small firms than in medium-sized firms. The analysis to be conducted in Chapter 4 will seek to address these issues using data from WERS 2004.

# 2.4 The HPWP-performance link in small and medium-sized firms

This section reviews debates on the HPWP-performance link in small and medium-sized firms separately.

# 2.4.1 The HPWP-performance link in small firms

As stated at the beginning of this chapter, while a large body of literature reports that HPWPs are positively associated with performance in large firms, studies of this relationship in small firms remain very limited. Of the studies that have been undertaken, several suggest that there

is no relationship between HPWPs and performance in small firms. Among these studies, training, as a single practice, has been widely reported to have no impact on business performance in the sector (Storey and Westhead, 1997; Storey, 2004). This is for a number of reasons. First, in the small firm sector resource poverty means the benefits of training may not outweigh the costs of implementing it, while in large firms the cost of training provision can be reduced via economies of scale (Storey and Westhead, 1997). Second, small firms often have difficulty in retaining their employees and are therefore unlikely to reap any benefits from the investment in training they have made in their employees (Storey, 2004). Third, the training activity in small firms is often informal and *ad hoc* and therefore is unlikely to have an impact on performance (Patton et al., 2000). Empirical evidence appears to reinforce such concerns. The implication of this is that any efforts to invest in training with the aim of improving performance in the small firm sector might be 'misguided' (Storey, 2004: 893).

However, a few studies report a positive relationship between HPWPs and performance in small firms. For example, de Grip and Sieben (2005) find a relationship between training and productivity in small pharmacy companies. Looking at HPWPs more broadly, Faems et al. (2005) combine several closely related HPWPs together, categorising HPWPs into six HRM domains (see Footnote 1 for the composition of these domains). Taking this approach, they find that some domains (training, compensation, career management and performance management) are individually related to labour productivity, although there is no such relationship where other HRM domains (selection and participation) are concerned.

Faems et al. (2005) also found that although several individual HRM domains were associated with improved productivity, they were not associated with improved financial performance. This is, of course, a critical issue in determining whether there is a genuine and meaningful relationship between HPWPs and performance (Guest and Conway, 2007). As Way (2002) argues, the absence of a relationship with financial performance is perhaps to be expected given that the gains from the implementation of such HPWPs may not exceed the labour costs associated with the use of these practices. Against this, Sels et al. (2006) found that the extensive use of HPWPs leads to increased productivity and an overall positive effect on small

firm profitability. This suggests that HPWPs can be valuable for small firms seeking to achieve better economic outcomes as the 'value-enhancing effect' of such practices outweighs the 'cost-raising effect' (Sels et al., 2006:319).

In contrast to the findings generated by studies looking at either individual HPWP (training) (de Grip and Sieben, 2005) or several closely defined HPWPs used in combination (Faems et al., 2005), Sels et al.'s (2006) study seems to suggest that small firms can maximize their profits by adopting a strong HRM system<sup>4</sup> (Bowen and Ostroff, 2004), which involves using as many HPWPs as possible. Although Sels et al. (2006) define small firms as having up to 99 employees, the study nonetheless casts doubt on whether the HPWP–performance link will also hold small firms with fewer than 50 employees in the private sector (the widely acknowledged definition for small firms used in official statistics in the OECD, European Union and the UK). Also, given none of the studies to date has examined the relationship between HPWPs and performance from all three perspectives, it remains unclear whether it is the different perspectives that leads to variation in the findings achieved.

Despite the lack of agreement within the limited studies conducted in small firms, as described above, the review of the uptake of HPWPs in Section 2.2 nevertheless suggests that individual HPWPs may be linked to performance in small firms as relatively higher transparency is likely to make the impact of such practices more direct than in medium and large firms. Beyond this, the increase in the use of HPWPs in small firms reported by a few recent studies (Bacon et al., 1996; Forth et al., 2006; Kotey and Slade, 2005) furthers the expectation that specific bundle(s) of HPWPs, i.e. a few closely defined HPWPs used together, may help to target particular HRM concern(s) and thus lead to better performance in these firms.

This thesis, therefore, seeks to explore the HPWP-performance link from all three perspectives. The analysis to be conducted in Chapter 5 will demonstrate whether there is a specific pattern to the HPWP-performance link in small firms.

<sup>4</sup> The strong HRM system has been frequently assessed using a count index in empirical studies (see for example, Appelbaum et al., 2000; Guthrie, 2001; Sels et al., 2006; Way, 2002).

20

#### 2.4.2 The HPWP-performance link in medium-sized firms

As with the literature on the uptake of HPWPs in medium-sized firms, the literature on the relationship between HPWPs and performance in medium-sized firms remains underdeveloped, both theoretically and empirically. Among the limited studies specifically undertaken on medium-sized firms, Storey (1998) argued that there was no evidence of an association between training and performance, and therefore there is little point in these firms adopting the training practices of larger firms.

This finding is similar to that reported for small firms (Storey, 2004; Storey and Westhead, 1997). Nonetheless, Storey's later work (Storey, 2002) reported a positive association between a set of practices relating to education, training and development, and performance when they were used in combination. Storey's (2002) study also revealed that there were high correlations between education, training and development practices, and any attempt to isolate one practice from the other two led to a significantly weakened relationship with performance. Although the three highly-correlated practices in combination only form an essential element in developing employees' ability, it nevertheless suggests that consistent and coherent HPWPs could generate a stronger impact on performance (compared to individual HPWPs used in isolation) in medium-sized firms.

This finding appears to support Delery and Doty's (1996) argument that synergistic effects are likely to occur when consistent HPWPs are used in bundles. As argued in the HPWP-performance link literature, practices that aim to develop employees' ability/competence (A), motivate them to use their ability (M) and provide them with the opportunity to contribute (O) (termed the "AMO framework") will elicit high commitment and result in enhanced performance (Boxall and Purcell, 2008: 5). If further evidence demonstrates that practices in connection with the AMO framework are also associated with high performance in medium-size firms, one implication would be that medium-sized firms are more likely to achieve better performance if they use closely-defined practices in combination rather than use specific

HPWPs in isolation. This would differentiate medium-sized firms from small firms where there might be more scope for individual HPWPs to be linked to enhanced performance.

A study of German medium-sized firms, however, suggests that medium-sized firms might adopt HPWPs as a result of external pressures, particularly to demonstrate enhanced reliability and accountability in order to increase the chance of obtaining finance (Frormann, 2006). If medium-sized firms have adopted HPWPs for this reason, they may distract attention away from developing employment practices aimed at ensuring co-ordination and control (Baron et al., 2007; Blau, 1970). Hence, one might argue that HPWPs adopted as a result of external factors (Frormann, 2006) will not necessarily lead directly to higher performance. If medium-sized firms are more likely than are other firms to have adopted HPWPs to impress external financers rather than to address specific co-ordination or workforce commitment/control issues, this in turn suggests there is a greater likelihood that the association between HPWPs and performance will be weaker in medium-sized firms than in small or large firms.

As such, this thesis also seeks to assess the HPWP-performance link in medium-sized firms from a number of perspectives, and hence evaluate whether the greater use of HPWPs in medium-sized firms than small firms also leads to higher performance.

#### 2.4.3 High performance work practices in small and medium-sized firms

The review in Sections 2.4.1 and 2.4.2 suggests that many existing studies on the HPWP-performance link in small and medium-sized firms focus on specific HPWPs such as training (de Grip and Sieben, 2005; Patton et al., 2000; Storey, 2002; Storey and Westhead, 1997) and it has also been recognised that a more transparent organisational structure in SMEs might allow the impact of an individual HPWP on performance to be more direct compared to a similar use in larger firms (de Kok and Uhlaner, 2001). The following section discusses the individual HPWPs one might expect to be associated with performance in the SME sector, and considers the differential impact one might anticipate they will have in small and medium-sized firms.

# Individual HPWPs and performance in SMEs

Following previous studies on the HPWP-performance link, this thesis selects 18 HPWPs from a list of HPWPs tested in previous studies and reported to have positive relationships with various performance outcomes (Combs et al., 2006; Guest et al., 2003; Hoque, 2000; Ramsey et al., 2000; Wall and Wood, 2005; Wood and de Menezes, 2008). Details on the HPWPs used in this selection of studies are presented in Table 2.1. A brief illustration of how each individual HPWP associated with high performance in large firms is relevant to small and medium-sized firms is provided below.

#### Recruitment/selection

Recruiting and selecting employees with the relevant competencies is the first critical step for firms pursuing competitive advantage through HRM (Wright and Snell, 1991). Texts on recruitment invariably recommend a systematic procedure that comprises stages including developing a job description, advertising via formal recruitment channels and conducting structured interviews. A careful and well-structured recruitment and selection process helps the firm to identify individuals with the sort of knowledge, skills, attitude and personal qualities that will enable them to fit into the firm and make a commitment to it (Carless, 2005). Potential applicants also benefit from such an approach as they gather information on the organisation's values and culture as they go through the process. The selection of individuals who possess superior skills and behavioural characteristics may enhance the firm's ability to develop a workforce capable of producing superior performance (Way, 2002).

In small and medium-sized firms, however, job vacancies are often filled in an informal and *ad hoc* way (Heneman and Berkley, 1999). Local newspaper advertisements, personal or employee reference and walk-ins, for example, are the most frequently used recruiting methods in small businesses, because these methods are convenient, inexpensive and provide managers with direct control over appointments (Cardon and Stevens, 2004). However, research suggests

Table 2.1 HPWPs used in studies on the relationship between HPWPs and performance

HPWPs	Combs et al., 2005	Guest et al., 2003	Hoque, 1999/2000	Ramsey et al., 2000	Wall & Wood, 2005	Wood & de Menezes, 2008
Sophisticated recruitment		$\sqrt{}$	$\sqrt{}$			
Induction		√ (combined with recruitment)	√ (combined with recruitment)	$\checkmark$		√
Off-the-job training	$\sqrt{}$	√ ·	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$
Internal labour market	$\checkmark$	$\checkmark$	$\checkmark$	$\checkmark$	√ (combined with performance related	$\checkmark$
Performance-related pay	V	$\sqrt{}$	$\sqrt{}$	$\sqrt{}$	pay) √	$\checkmark$
Profit-related pay	J	V	$\vec{\lambda}$	į	V	Ì
Performance appraisal	V	V	į.	V	$\dot{}$	V
Teamwork	$\sqrt{}$	V	V	V	V	V
Team briefing	√ (termed information sharing)	)	$\sqrt{}$	√ (termed downward communication)	√	$\checkmark$
Consultation committee	) .	√ (termed two-way communication)		√ ····································	√ (termed participation)	$\checkmark$
Employee attitude survey Quality circles	√ (termed participation)	)	N	(termed problem	r	1
•		.1	V	solving group)		-1
Functional flexibility		V		√ (included as part of training)	,	V
Benefits	$\checkmark$	√ (termed single status & harmonization)	$\sqrt{\text{(termed single status)}}$	√ (termed harmonization)	V	√ (termed single status)
Flexible working/family-friendly	1	1	√ .	$\checkmark$	√ (termed harmonization)	
Equal opportunities			√ (termed harmonization)	√	,	
Grievance procedures Job security	1	<b>√</b>	√	√ √	√	√

the adoption of more formal procedures and methods could reduce staff turnover and its associated costs in small firms (Carroll, 1999). Research also indicates that, by adopting formal recruiting practices reflecting industry norms, SMEs can gain legitimacy in their industries and increase the likelihood of potential applicants applying for posts (Williamson, 2000). As such, formal recruitment practices may lead to appropriate skills being developed and hence, may lead to improved performance in small and medium-sized firms.

#### Induction

Formal induction procedures for introducing new employees to a firm can generate better performance outcomes given that new staff will understand the work, the communication systems and the operational practices of the firm from the start of their period of employment. Induction, along with sophisticated recruitment and training, are key components for the formation of workers' competence that contribute to higher performance at both the level of the employee and the level of the firm (Guest, 1997). However, compared to the extensive attention paid to sophisticated recruitment and training, only a few studies include induction as an independent HPWP. This is confirmed in the sample of empirical studies presented in Table 2.1. Only two of these studies make reference to it, and these studies treat induction as a component of a more general sophisticated approach to recruitment (Guest et al., 2003; Hoque, 2000).

However, formal induction procedures may well be atypical within the SME sector, hence introducing such procedures may make small firms stand out from their competitors, not only in terms of developing new talent, but also in terms of subsequent firm performance. As with recruitment/selection practices, the adoption of formal induction procedures in small firms is notably low in comparison with large firms (Forth et al., 2006). This could be because, as Green (1993) argues, the flat organisational structure within small firms may make such a practice redundant. That said, one could also argue that first impressions of a firm may affect the commitment that a new employee will have towards his/her job. A properly managed induction can convey effectively, as in large firms, the firm's ethos and objectives so that

newly recruited employees can align their aims and objectives those of the organisation's and shape their behaviour in line with the firm's policy. As such, induction processes may well prove to be an important mechanism by which firms can develop a highly committed workforce that is willing and able to produce high levels of labour productivity.

#### Training

Training can help to improve firm performance by imparting relevant new knowledge and skills, based on employee and organisational needs, if it is effectively designed and delivered (Salas et al., 1999). When training results in improvements in relevant knowledge and the acquisition of relevant skills, it is likely that employee job performance will improve, provided that the skills learned in training can be transferred to the job (Salas et al., 1999). A meta-analysis (Tharenou et al., 2007) based upon 67 studies of large firms suggests that training is positively related to human resource outcomes and organisational performance but is only very weakly related to financial outcomes.

However, as argued above, the resource-poverty argument presumes that it is less likely that small firms will train their employees than large firms as the fixed costs cannot be distributed across a large workforce (Storey and Westhead, 1997). As also discussed above, Storey (1998; 2002; 2004) reports that neither small firms nor medium-sized firms benefit from the provision of training. Conversely, medium-sized firms may introduce more training for their workforce than small firms as they benefit from greater economies of scale than do small firms (Lynch and Black, 1998). Thus, a positive association between training and firm performance may be expected in medium-sized firms, if not in small firms.

#### Internal labour market

Firms that give preference to their internal candidates in filling job vacancies are likely to keep talent longer and may also increase the chance of a return on HPWPs investments (Green, 1993). Employees will recognise that these firms offer the potential for promotion and good

career prospects. Hence, it could be argued that a 'win-win' situation will emerge as employees secure benefits in terms of career progression while firms achieve high yields from a highly competent and committed workforce (Combs et al., 2006). Hence, where small and medium-sized firms are able to offer opportunities for career progression, this could lead to benefits in terms of improved retention, greater commitment and a better developed workforce.

However, small firms in particular face a dilemma with regard to this. While a lack of talent may be detrimental to their performance, at the same time the absence of internal labour markets within these firms makes talent difficult to recruit and retain (Cardon and Stevens, 2004). Given this, in instances where small firms have been able to implement an internal labour market one might expect performance improvements from this. However, at the same time, the scope for small firms to put an internal labour market in place might be limited.

#### Performance appraisal

Performance appraisal represents a formalised process of employee monitoring and is intended to be a management tool to improve productivity and performance (Cardy and Dobbins, 1994). It can be used for a wide range of different purposes (Bowles and Coates, 1993), all of which can work towards enhancing firm performance, although the probability of establishing such a relationship is increased in certain circumstances. For example, performance appraisal oriented to performance monitoring and rewards provision is found to be related to improved performance in firms with employees having shorter tenure, while appraisal with the purpose of aligning their efforts with organisational goals and providing career development plans is associated with greater performance in firms where the employees have substantial latitude to influence productivity (Brown and Heywood, 2005). Nonetheless, it is also acknowledged that the implementation of a formal performance appraisal involves substantial costs and will most likely be undertaken in circumstances in which the anticipated gains are greatest.

In SMEs the extant evidence suggests that many employees feel that performance appraisals are conducted too informally, and employers feel they lack the time to conduct appraisals more

effectively (Dundon et al., 2002). The informal manner in which performance appraisals are conducted in SMEs may be detrimental to employee commitment and harmful to performance (Cassell et al., 2002). As such, regular and thorough formal performance appraisals in small and medium-sized firms may well help to shape employees' attitude and behaviour and align these with the firm's goals. Hence, one might anticipate such appraisals to be linked to enhanced performance.

#### Performance-related pay

Performance-related pay can take various forms. For example, it can be an award based on judgements about an employee's performance made through a system of performance appraisal, or it can simply be decided in accordance with the quantity produced by individuals. Irrespective of the existence of different types, performance-related pay is argued to be able to motivate employees to perform their best at work and generate high productivity and increased economic return without creating significant wage pressure (Piekkola, 2005). Nonetheless, evidence also shows that performance-related pay schemes may lower morale and have a demotivational effect (Marsden and Richardson, 1994). This may be the result of suspicions among employees over the fairness of the appraisal process and the appropriateness of relying solely upon individual managers in making appraisal judgements. Nevertheless, advocates posit that by bringing rewards into line with results, a firm can directly monitor its financial performance, and employees can gain a better understanding of their individual performance against the explicit criteria for evaluation, thus encouraging them to amend and shape their behaviour accordingly (Keizer, 2009).

Given the relatively direct link between individual behaviours and firm performance in small and medium-sized firms, performance-related pay may have a stronger impact on motivating workers in these firms than in large firms. Hence, a positive relationship between performance-related pay and firm performance may be expected in small and medium-sized firms.

#### Profit-related pay

A competitive level of compensation is critical to retain competent employees (Lam and White, 1998), and profit or target-related pay schemes can have a direct impact on performance through the promotion of employee commitment to organisational goals (Fernie and Metcalf, 1996). Although it is contentious to use pay as an incentive to elicit intrinsic commitment from employees (Wood and de Menezes, 1998), the existence of utilitarian exchange (Cohen, 2007) between the employer and employees implies financial incentives, such as profit-related pay, can develop high 'instrumental' commitment (Meyer and Allen, 1997). In light of the resource-poverty argument, it is possible that small and medium-sized firms linking employee compensation with the firm's profitability would provide competitive leverage.

#### Teamwork

Job and work structures have the potential to enhance firm performance by allowing skilled and motivated employees to become more involved in determining what work needs to be carried out and how it should be achieved (Delaney and Huselid, 1996). Defined as an interdependent collection of individuals who share responsibility for specific outcomes of their organisation (Sundstrom et al., 1990: 20), teamwork is argued to be a work structure facilitating multi-skilled co-ordination, effective resource allocation, high quality, rapid innovation and high customer satisfaction (Boyett and Conn, 1991). A longitudinal study conducted by Banker et al. (1996) reports that both product/service quality and labour productivity improved in the years following the formation of work teams. Procter and Burridge's (2008) analysis drawing upon the WERS2004 data confirms that teamwork is positively associated with product/service quality and labour productivity. The debates over the performance effectiveness of teamwork remain unresolved, however, as Harley's (2001) analysis of the WERS 1998 data suggests that teamwork does not matter much in relation to firm performance. This casts doubt on whether a relationship is likely to exist in small and medium-sized firms. Critical accounts argue that teamwork can undermine employee discretion, intensify work and increase stress for employees (Findlay et al. 2000; Marchington, 2000). Thus, teamwork might not necessarily boost performance (Ramsey et al., 2000).

Although SMEs are less likely to use teamwork than large firms (Forth et al., 2006), the informality in small and medium-sized firms offers greater flexibility compared to large firms. Hence, SMEs appear to have the ideal structure for teamwork systems given that individual's work is often dependent upon others and these firms offer an atmosphere that is naturally conducive to teamwork. However, teamwork also means individual employees need to take on multiple tasks that may require a breadth of skills and knowledge. Given the particularly low incidence of training in SMEs compared to large firms in general (Forth et al., 2006), it is possible that SME employees will lack of the skills and knowledge required in order to function effectively within a teamworking environment that requires multi-skilling. Hence, one might argue that there is no particular reason to expect that teamwork within small or medium-sized firms will lead to improved performance.

#### Team briefings

Communication channels such as team briefings enable employers to communicate organisational goals and the business position of the firm to employees and may increase employee understanding of the reasons behind business decisions. As a result, high commitment could develop as individual aims become better aligned with the firm's goals. Effective information flows can shape employee perceptions of fairness and their behaviour, even if employees do not expect to have an impact on decisions (Way, 2002). Greater information flows might also increase employees' perceived trust in management, hence leading to higher motivation and higher firm performance (Guthrie, 2001).

Employees in small firms are more likely to report that their managers are good or very good at informing their employees about changes in various operational issues than are those in medium-sized and large firms. However, they are less likely to inform their employees about financial issues (Forth et al., 2006). Arguably, this is because performance in small firms is generally poor so conveying such a message would not impact positively on the workforce. In comparison with small firms, medium-sized firms are more likely to discuss financial

information with their employees (Forth et al., 2006). Hence one might expect that team briefings to be linked to better performance in medium-sized firms than in small firms.

#### Consultation committees

Consultation committees of managers and employees or their representatives in the workplace are primarily concerned with the exchange of views. These committees are sometimes called joint consultative committees, work councils or representative forums, and it is argued that the establishment of such committees can improve managerial systems by tapping into employees' ideas, knowledge and experience, promoting greater diffusion of information and facilitating improved relations with trade unions (Wilkinson et al., 2004). The existence of consultation committees could also be viewed as proof that management recognises employees' contributions and is willing to listen to them and discuss key issues, which may give rise to improved employee attitudes and behaviours, loyalty, commitment, more cooperative relations and, ultimately, superior performance.

It is not a surprise that, given the scope for direct communication in small firms, use of formal consultation channels such as joint consultative committees is lower than elsewhere (Forth et al., 2006). However, questions could be raised over whether or not informality is sufficient as a viable mechanism for independent employee participation in the absence of formal structures, particularly as evidence suggests there has been much more information than consultation in SMEs after the introduction of the Information and Consultation of Employees (2004) Regulations (Wilkinson et al., 2007). This issue is made more problematic by the fact that the Regulations do not encompass firms with fewer than 50 employees and, thus, there is likely to be a very low uptake of such practices in small firms. However, those small (and medium-sized) firms that are able to develop formal consultation procedures may be better placed to convince the workforce of the management's integrity, trustworthiness and consistency (Cassell et al., 2002). In return, these firms may gain trust and elicit high workforce commitment. The benefits of such procedures might be particularly great in medium-sized firms given they are more likely than small firms to experience co-ordination problems.

#### Employee attitude surveys

Understanding how the workforce perceives workplace policies and procedures may help managers understand employees' needs and to make changes to policies and procedures in light of this. This may result in motivated employees who demonstrate commitment. It might also facilitate employee retention and lead to superior firm performance (Mirvis and Lawler, 1977).

It is rare, however, for SMEs to carry out formal employee attitude surveys because informal relationships within these firms may be seen by managers as sufficient (Forth et al., 2006). However, once these firms start to use such a practice to access employees' views on various management issues, this may result in improved employee satisfaction, higher work commitment and overall improved performance.

#### Quality circles

Known as a key attribute of total quality management, quality circles or continuous improvement groups require employees to understand the production process and take responsibility for solving problems and improving the quality of products or services (Legge, 1995). Many organisations are looking for a competitive advantage through improved product or service quality (Porter, 1998). Hence, quality programmes are often used as a strategy to increase employee participation in decision-making (Lawler et al., 1998). Indeed, quality circles are seen as marking the commitment of organisations to changing organisational culture (Hill, 1991: 541). Longitudinal studies report that quality circles can lead to improved job satisfaction (Griffin, 1988) and small but statistically significant benefits for productivity, absenteeism and intentions to quit (Marks et al., 1986). Recent studies report similar findings. Pereira and Osburn (2007), for example, report that quality circles have a stronger impact on job performance than on employee attitudes. However, others argue that quality circles are part of a strategy by managers to bypass trade unions and create an individualistic relationship with

employees in order to increase the legitimacy of management in employees' eyes (Batstone and Gourlay, 1986: 117-20). Hill (1991:566) further argues that quality circles were bound to fail given that they were introduced within conventional organisational structures that comprise bureaucratic and Taylorist principles. Although benefits might occur in the short term, in the longer term performance is likely to deteriorate as a result of the failure to integrate quality circles into broader firm structures. Quality circles, therefore, need to be part of a broader strategy to change the organisational culture and structure throughout the organisation as a whole.

The relatively transparent organisational structure in small firms may make it easier to implement quality circles that are embedded within and complementary to other systems within the organisation. It may be possible, therefore, for small firms to benefit from such practices in terms of improved quality of product or service. In contrast, the increased layers of management in medium-sized firms resulting from a much larger workforce may result in more bureaucratic structures and hence a greater risk of an incompatibility between quality circles and existing organisational structures. It remains to be seen therefore whether using quality circles as an individual practice is associated with performance in medium-sized firms.

#### **Functional flexibility**

Training employees to carry out jobs other than their own is recognised as a way of increasing an organisation's flexibility and thereby developing sustainable capabilities that are inextricably embedded in organisational experience, learning and practice. Such training can lead to a multi-skilled workforce, enable employees to work in teams and empower employees to participate in decision-making. It can also help firms to adjust in fast-changing environments (Wood, 1999). Functional flexibility is often analysed as a component of a specific set of practices, for example, HPWS, and has been demonstrated to be associated with performance (Appelbaum et al., 2000).

Following on from the argument relating to potential impact of teamwork in small and medium-sized firms, training employees in a broad range of skills may enhance the capability of employees to contribute to teamworking. Although there is no research conducted on functional flexibility exclusively in small or medium-sized firms, the need for employees in small firms to cover a wide range of tasks and the limited scope for task specialisation in small firms (Storey and Westhead, 1997) suggests a need to train SME employees to be multiskilled. Given SMEs are less likely to provide such training to their employees, in instances that they do, better performance would be expected.

#### Benefits

Benefits in the form of non-pay compensation are designed to improve the quality of working and personal lives of workers (Milkovich et al., 2005). Benefits such as employer pension schemes, private health insurance, more than four weeks of paid annual leave and sick pay in excess of statutory requirements are said to help firms attract new candidates, retain current employees and produce effective performance (Bohlander et al., 2001). The evidence also shows that benefit plans that tend to increase in value over time also encourage employees to stay longer with their employer (Gomez-Mejia et al., 2001). This line of argument implies that firms providing benefits to their workforce might be expected to have high job satisfaction, sustained loyalty, high retention of employees and improved product or service quality.

However, small and medium-sized firms are less likely to offer benefits compared to their larger counterparts (Forth et al., 2006). This could be because many small and medium-sized firms operate in hyper-competitive markets, where low or non-existent profit margins do not leave room to readily absorb extra costs (Ram and Edwards, 2003). Nevertheless, although the offer of a competitive benefits package remains a rarity within the SME sector, small and medium-sized firms that are able to offer these kinds of benefits would be expected to increase their chance of achieving superior performance compared to other firms.

#### Flexible working/family-friendly practices

The aim of flexible working/family-friendly schemes is to correct the adverse effects of work—life conflicts by assisting individuals so that improved performance can be achieved (Glass and Fujimoto, 1995). Schemes to help families care for children or older adults, a workplace nursery or a nursery linked with the workplace, work at home or paid parental leave, enable employees to focus on and remain in work (Glass and Fujimoto, 1995). In addition, enhanced loyalty, high satisfaction and productivity can be achieved if flexible working/family-friendly practices are in place (Davis and Kalleberg, 2006). Apart from the potential benefits gained from flexible working/family-friendly plans from the employers' point of view, it is also recognised that employers feel pressure to use such schemes in order to attract skilled labour. This is particularly the case where labour markets are tight (White and Rogers, 2000). Hence there are a number of reasons why flexible working/ family friendly practices might lead to high labour productivity and/or high labour retention.

Although the available evidence indicates that SMEs are less likely to report the use of flexible working/family-friendly practices, employees in these firms usually report that they would have access to such practices if they were required (Forth et al., 2006). There is a suggestion, therefore, that there is a greater availability of informal flexible working/family-friendly practices in small and medium-sized firms, although this is hard to measure. Nonetheless, it can possibly be hypothesised that small and medium-sized firms with such practices in place are likely to have better employment relations and better firm performance than are firms lacking these practices.

#### Equal opportunities

It has been long argued that there is a business case for promoting equal opportunities in employment. Harnessing talent from a diverse background (for example, age, gender, race, religion and/or sexual orientation) can help firms to achieve high labour productivity and increased economic return (Dickens, 1986). Wilson and Iles (1999) report that managing a

diverse workforce effectively and providing equal opportunities in various management channels, such as recruitment, development and promotion, could develop co-operative behaviour and improve operational efficiency and firm profitability. In addition to the full utilisation of the skills and potential of all employees, equal opportunities and diversity management can contribute to organisational success by enabling access to increasingly diverse markets and improving corporate image (Kandola and Fullerton, 1994). Where empirical evidence is concerned, there is limited evidence of an association between the use of equal opportunities alone and enhanced firm performance (Shen et al., 2009), although a count measure of HPWPs that includes equal opportunities has been demonstrated to have a positive association with, for example, increased labour productivity (Ramsey et al., 2000).

Small and medium-sized firms are not only less likely to have a formal written policy on equal opportunities, but are also less likely to monitor or review their recruitment and promotion procedures to identify indirect discrimination (Forth et al., 2006). Either way, it remains to be seen whether the extent of the application of equal opportunities practices is associated with better performance in small and medium-sized firms.

#### Grievance procedures

Formal grievance procedures that aim to resolve potential differences between employers and employees in a reasonable, fair and consistent manner can reduce employee dissatisfaction and increase motivation, thereby potentially reducing labour turnover and increasing retention rates (Renwick and Gennard, 2001). A business case for adopting formal grievance procedures exists in that potential damage to corporate image and costs incurred by industrial tribunals will inevitably result in low morale and mistrust among employees and loss of confidence among customers, which can subsequently harm a firm's business performance. This has been empirically verified by a meta-analysis suggesting that the application of formal grievance procedures is positively associated with firm performance (Combs et al., 2006).

Some recent research reports that the majority of small firms have adopted formal grievance procedures (Forth et al., 2006), although they are less likely to have adopted them than are larger firms. Arguably, small firm managers may view formal grievance procedures as overly bureaucratic and they may prefer to deal with grievances in an informal manner. Hence, one might argue that the application of formal grievance procedures could bring about additional costs to small and medium-sized firms and thus harm performance. Conversely, it can also be argued that the provision of such procedures in small and medium-sized firms could have a positive impact on workforce commitment and firm performance given that the workforce might see such procedures as a demonstration of trustworthiness, integrity and goodwill on the part of the employer.

#### Job security

It has been argued that guaranteed job provision can develop trust between employees and employers, which may elicit commitment to firm performance (Preuss and Lautsch, 2002). Evidence also demonstrates that a job security policy is related to higher labour productivity and enhanced market returns (Combs et al., 2006). This does not suggest that job security is necessarily a requirement for better performance as employees in the current climate no longer expect unconditional job security given that volatility in the competitive environment places complete security outside management control (Rousseau and Tijoriwala, 1998). Employees instead may now expect only that management will engage in good faith efforts to secure jobs. However, firms that do provide job security or a non-compulsory redundancy policy may well be distinct from their competitors, thus helping them to attract talent and elicit high commitment, ultimately leading to superior performance. The impact might be particularly apparent in small and medium-sized firms as such provision is comparatively rare (Forth et al., 2006).

#### **HPWP** bundles

While many of the individual HPWPs discussed above may be associated with enhanced performance, the existing evidence also shows that HPWP 'bundles' demonstrate much stronger associations with performance outcomes (MacDuffie, 1995), given the synergistic effects that can emerge from a mutually reinforcing package of HPWPs (Delery, 1998; Huselid, 1995). Specific HPWP bundles that comprise several HPWPs are therefore viewed as more effective than individual HPWPs. On the contrary, 'deadly combinations' can also occur whereby HRM policies or practices work in opposite directions, such as teamwork alongside a performance appraisal system that only rewards highly individualistic behaviours (Delery, 1998:294). Hence, the strategically designed bundling of various HPWPs is considered essential to deliver high levels of firm performance.

Nonetheless, there is no general agreement on what the bundles are or what their components should be. Within the literature it is argued that enhanced performance can be achieved through three HPWP bundles: managing the workforce's ability; motivation; and the opportunity to participate (AMO) (Boxall and Purcell, 2008). Many empirical studies have tried to validate this conceptual framework. Examples of previous empirical studies of HPWP bundles are presented in Table 2.2.

Some of these studies did not find any of the HPWP bundles frequently described in the HRM literature. For example, drawing data from WERS 2004, Guest and Conway (2007)<sup>5</sup> found no evidence that the choice of a range of HPWPs follows a specific pattern. Instead, they allocated the HPWPs into four bundles in line with the AMO conceptual framework and added a commitment bundle comprising practices such as equal opportunity, flexible working, fringe benefits and job security. The results suggest that the ability/competence bundle is associated with lower labour turnover and superior financial performance, while the motivation bundle is

<sup>&</sup>lt;sup>5</sup>The sample used in Guest and Conway's (2007) study excludes all independent workplaces with fewer than 25 employees.

HPWP "bundles" in selected studies Table 2.2

Empirical Studies		HPWP	Allocation	Database		
	Bundle 1	Bundle 2	Bundle 3	Bundle 4	techniques	Datavase
Guest & Conway, 2007	Competence	Motivation	Opportunity to contribute	Commit- ment	AMO & authors' self-justification	WERS 2004 <sup>1</sup>
Huselid, 1995	Employee skills and organization- al structure	Employee motivation			Principle component factor analysis	Compact Disclosure <sup>2</sup>
MacDuffie, 1995	Use of buffers	Work systems	HRM policies		Cluster analysis	Questionnaire survey <sup>3</sup>
Michie et al., 2008	Cluster 1	Cluster 2	Cluster 3	Cluster 4	Cluster analysis	WERS 2004
Wood & de Menezes, 1998	Low HCM	Medium- low HCM	Medium- high HCM	High HCM	Latent classic modelling	WIRS 90 & EMSPS⁴

Notes:

associated with higher productivity. Guest and Conway (2007) conclude that the association between individual HPWP bundles and performance is somewhat weak.

In line with MacDuffie's (1995: 204) argument that cluster analysis appears to be an ideal technique for detecting HPWP bundles as it 'groups observations that lie in close proximity in multidimensional space for a given set of variables', Michie et al. (2008: 42) conducted cluster analysis also drawing on data from WERS 2004 and reported four clusters (see Table 2.2). They did not give these clusters specific titles, however, on the basis that none of them 'particularly match the sort of HR bundles generally described in the literature'. While the differentiation between these four clusters was not clear enough to identify which HPWP

<sup>&</sup>lt;sup>1</sup> The British Workplace Employment Relations Survey 2004.

<sup>&</sup>lt;sup>2</sup> A database containing comprehensive financial information from 10-K reports (informational documents filed with the Securities and Exchange Commission in the U.S.A.) on nearly 12,000 publicly held U.S. firms.

The International Assembly Plant Study was sponsored by the International Motor Vehicle Program (IMVP) at M.I.T

U.S.A.

<sup>4</sup> The British Workplace Industrial Relations Survey (WIRS) 1990 and the Employers' Manpower and Skills Practices Survey (EMSPS).

bundles they potentially relate to, they did nevertheless find overlaps between each cluster, similar to those reported in MacDuffie's (1995) study. Michie et al. (2008) justified their use of the four clusters mainly in accordance with the extent to which a range of HPWPs were used between firms. In other words, all workplaces that 'are highest on selection rigour, training opportunities, job autonomy, teamwork, communications, equal opportunities, flexible working/family-friendly practices, job security, consultations with employees, and HR strategy' should go to the same cluster, i.e. Cluster 1, which would seem to be 'the closest of the four to what is generally characterized in the literature as a 'high commitment work system'' (Michie at al., 2008: 43). Meanwhile, workplaces in Cluster 3 scored the lowest on the counts of the set of HPWPs used, and Clusters 2 and 4 fell somewhere in between. The ambiguity involved in this framework, however, limits the extent to which the findings may be interpreted in line with the theory of HRM.

Different techniques aside, pooling data from both the Workplace Industrial Relations Survey 1990 and the Employers' Manpower and Skills Practices Survey (EMSPS), Wood and de Menezes (1998) report that the use of HPWPs in British workplaces does follow specific patterns. However, the pattern implies that it is the extent of the use of HPWPs that might make a difference between firms rather than which of the HPWPs have been adopted. Indeed, both the results produced by cluster analysis (MacDuffie, 1995; Michie et al., 2008) and latent classic modelling (Wood and de Menezes, 1998) would appear to suggest that categorising firms into different groups according to the extent of the uptake of HPWPs in such firms can be used to evaluate whether a firm uses HPWPs as a complex system (in terms of using many HPWPs). Although these findings lend support to Bowen and Ostroff's (2004) argument that a complex HRM system will have a greater impact on performance, it is still not clear which HPWPs used together would lead to better performance.

However, a few studies have reported that HPWP bundles exist and that these bundles are associated with enhanced performance. For example, using principle component factor analysis, Huselid (1995) noted that two HPWP bundles (termed "HR systems"), i.e. employee skills/organisational structure and employee motivation, are associated with lower labour

turnover, higher labour productivity and enhanced financial performance. Huselid's (1995) identification of HPWP bundles may be the closest to the AMO theoretical framework, though the bundle representing 'employee skills and organizational structure' is in fact a combination of the ability/competence bundle, the opportunity to perform/contribute bundle and the commitment bundle in Guest and Conway's (2007) study. Either way, the principle component factor analysis used in this study would appear to be a useful technique for detecting latent patterns of HPWPs.

Similar studies of small and medium-sized firms are, however, rare. Although Storey (2002) reports a relationship between a combined measure of training activities, employee development and education events (none of which was individually associated with performance), this is far from suggesting that medium-sized firms tend to use specific HPWP bundles. As such, a systematic assessment and understanding of whether the use of HPWPs follows specific patterns and whether such bundles lead to better performance in small and/or medium-sized firms is yet to be developed. Nonetheless, a growing body of studies shows that HPWPs are used by small firms (Bacon et al., 1996; Forth et al., 2006; Golhar and Deshpande, 1997) suggesting that specific HPWP bundles may well be used in some small firms.

It would appear, therefore, that identifying which HPWPs are used together in small and medium-sized firms is essential in helping to understand whether the increase in the adoption of HPWPs in small firms follows specific patterns distinct from medium-sized firms and whether the adoption of such bundles leads to distinct relationships between HPWPs and performance in small and medium-sized firms.

#### **HPWP** intensity

A majority of studies demonstrating positive associations between HPWPs and performance use a count/cumulative measure of such practices (Appelbaum et al., 2000; Guest et al., 2003; Guthrie, 2001). The logic for such a measure of HPWPs is that it can capture the complementary effect between these practices (Delaney and Huselid, 1996) and it may also

represent a complex HRM system which can generate a greater impact on performance compared to either individual HPWPs or specific HPWP bundles (Bowen and Ostroff, 2004). As 'the exact mechanisms creating that positive association may vary somewhat across practices' (Delaney and Huselid, 1996: 952), studies that adopt HPWP intensity to assess the HPWP-performance link generally assign equal weights to each individual HPWP so that any significant associations established should be interpreted as the result of the aggregation of those HPWPs.

A few empirical studies lend support to the view that the intensity of use of HPWPs can enhance small firm performance. Sels et al. (2006), for example, report that HPWP intensity significantly increased financial performance in small firms. However, the analysis contains firms with up to 100 employees. Hence, it is not possible to know from their results whether the reported HPWP-performance link holds in small firms alone or whether the results are driven by the inclusion of medium-sized firms in the analysis.

Alongside the benefits created by a count measure of HPWPs, an analysis solely relying on the count measure of HPWPs may nonetheless risk misinterpreting the results for small and medium-sized firms. This is because the adoption of HPWPs is likely to be subject to several objectives that are not always consistent with each other. In medium-sized firms, for example, the extant evidence implies that, as stated in Section 2.2.2, potential tensions might emerge between the demand for more HPWPs to manage a larger workforce (Baron et al., 2007) and the external pressure to introduce specific HPWPs to satisfy potential financers (Frormann, 2006). If such tensions do occur, the use of more HPWPs may not necessarily be associated with enhanced performance in medium-sized firms. However, such a finding would not rule out the possibility that some of the HPWPs may be individually associated with better performance in medium-sized firms. This means attention should simultaneously be paid to individual practices as well as specific clusters/sets of practices. Hence, a comprehensive understanding of the nature of HRM in small and medium-sized firms requires a consideration of the impact of individual HPWPs as well as the impact of HPWP bundles and HPWP intensity.

#### 2.5 Discussion and conclusions

This chapter has explored issues surrounding the use and impact of HPWPs in small and medium-sized firms from three perspectives. First, this chapter reviewed debates on the uptake of HPWPs in small and medium-sized firms. Second, it identified and described factors that may exert an influence on the use of HPWPs in small and medium-sized firms. Third, it discussed debates on the relationship between HPWPs and performance in small and medium-sized firms.

The review of debates on the uptake of HPWPs suggests that small firms rarely use formal practices such as HPWPs compared to large firms, and in instances where these firms use such practices, they use them in an *ad hoc* way (Sisson, 1993; Storey and Westhead, 1997). Nevertheless, a few studies report that HPWPs are found in small firms (Bacon et al., 1996; Forth et al., 2006), though they are not as prevalent as in medium-sized firms, which tend to adopt more HPWPs in order to cope with the increasing intensity of HR problems such as coordination and control when the size of workforce expands (Baron et al., 2007; Greiner, 1972).

The review of the debates on the uptake of HPWPs in small firms also suggest that there may be significant variation within the small firm sector in relation to the extent to which HPWPs are used. Roberts et al. (1992) reported that small firms with more than 20 employees tend to use as many HPWPs as large firms, suggesting that there may be a firm size threshold beyond which informality becomes untenable. In addition, differenced use of HPWPs between medium and large firms may be minimal, given Kotey and Slade's (2005) argument that as size expands, the adoption of such practices increases at first rapidly but then more slowly. Hence, the differences between small and medium-sized firms may well be greater than the differences between medium-sized and large firms in terms of the uptake of HPWPs.

The review of debates on the factors associated with the uptake of HPWPs (i.e. antecedents of HPWPs) indicates that a range of contextual factors may influence the extent to which HPWPs are adopted. Among these factors, some are expected to be strongly associated with the use of

such practices in small firms. For example, a highly competitive market may be linked to a lower use of HPWPs in small firms as they tend to find it is too costly to introduce HPWPs when severe competition drives down the price of the products or services they supply (Smallbone et al., 1999). Some might be strongly associated with a greater use of HPWPs in medium-sized firms, given the need to introduce additional HPWPs for co-ordination purposes as the workplace grows in size and operations become more complex (Baron et al., 2007; Forth et al., 2006; Kotey and Slade, 2005); and others are expected to be associated with a higher uptake of HPWPs in both small and medium-sized firms, such as firm age (Wager, 1998). Overall, the use of HPWPs in small firms would be expected to be more sensitive to external factors, while medium-sized firms may introduce more HPWPs to deal with internal factors that are possibly related to specific HRM concerns such as co-ordination and control (Baron et al., 2007; Greiner, 1972).

The review of the HPWP-performance link suggests that, in small firms, individual HPWPs might be positively associated with performance in small firms given that within the sector as a whole HPWPs are rarely used (Storey, 2004). Thus adopting a small number of practices in a piecemeal fashion might be sufficient to differentiate the firm from its competitors. Nevertheless, one might still expect specific HPWP bundles to be linked to improved performance. In medium-sized firms, however, one might expect HPWP bundles to be more likely to be associated with better performance than individual HPWPs used in isolation (Storey, 2002). Nonetheless, doubt has been cast on whether an overall relationship between HPWP intensity and performance would be expected in medium-sized firms given that HPWPs may be introduced mainly to satisfy the external requirements of potential financers (Frormann, 2006). This may distract managers away from central HR concerns (Baron et al., 2007).

Assessments to be conducted in Chapter 3 and onwards will seek to address these issues. If it can be shown that there are significant differences between small and medium-sized firms in terms of the extent of uptake of HPWPs, the factors associated with the uptake of HPWPs and the relationship between HPWPs and performance, this will add to the argument that small and

medium-sized firms should be treated as two distinct groups of firms (Hoque and Bacon, 2008), rather than banded together as a homogenous group.

# Chapter 3 The uptake of HPWPs in small and medium-sized firms

#### 3.1 Introduction

As discussed in Chapter 2, while most research findings indicate that informality tends to dominate small firms and the appropriateness of introducing HPWPs into these firms should be questioned (Marlow, 2006), other evidence suggests that such practices do exist in small firms (Bacon et al., 1996; Forth et al., 2006) and the variation in the adoption of these practices between small and large firms is not as significant as expected (Golhar and Deshpande, 1997).

Compared to the amount of literature on small firms, SMEs or large firms in general, understanding of the uptake of HPWPs specifically in medium-sized firms is particularly underdeveloped. Existing evidence, although limited, suggests that the use of HPWPs in medium-sized firms could be significantly different from that in small firms (Baron et al., 2007; Forth et al., 2006; Hoque and Bacon, 2008). In addition, the positive correlation widely reported between the adoption of specific HPWPs and firm size (Forth et al., 2006) implies that the uptake of HPWPs in medium-sized firms may be higher than in small firms. If this is shown to be the case, the implication will be that the conventional approach of combining small and medium-sized firms into a single unit of analysis will be rendered somewhat questionable.

This chapter addresses these issues in two steps. The first step is to assess the extent to which the uptake of HPWPs in medium-sized firms differs from that in small and large firms. The second step is to evaluate whether firm size explains the bulk of the variation in the uptake of HPWPs between these firms and whether there is a threshold at which point informality becomes untenable.

This chapter is constructed as follows. The next section (3.2) describes the estimation techniques involved in the assessments. The second section (3.3) reports the results. The final section (3.4) discusses the findings and concludes.

#### 3.2 Research method

The assessment of the extent to which the uptake of HPWPs differs between small and medium-sized firms is conducted in three steps. First, a set of 18 HPWPs are cross-tabulated with a categorical firm size variable, thereby allowing  $Chi^2$  analysis to be performed on the differences in the uptake of these practices between small firms and medium-sized firms (large firms being used as a comparator group). Second, a maximum likelihood multi-variate analysis is conducted to hold a range of factors constant that might bias estimates of differences between small and medium-sized firms. Third, using the same set of control variables as in the second step, maximum likelihood multi-variate regression is carried out by further categorizing firm size to detect whether there is a threshold of formality.

#### 3.2.1 Data/sample

The analysis draws on data from the Workplace Employment Relations Survey (WERS) 2004 management survey. WERS 2004 is designed to be nationally representative of British workplaces comprising five or more employees within Standard Industrial Classification major groups D to O (excluding agriculture, hunting, forestry and fishing, and mining and quarrying). It is the first time that workplaces with as few as five employees have been included in WERS, which allows the research to evaluate the nature and impact of HPWPs in small and medium-sized firms in a more representative cohort of small firms than was the case previously. The survey comprises 2,295 observations with a response rate of 64 per cent (Kersley et al., 2005). Respondents to the survey are the most senior managers within the workplace with responsibility for employee relations matters.

The estimation techniques used to address the issues under consideration in this chapter are described in detail below.

#### 3.2.2 Dependent variables: HPWPs

The uptake of HPWPs is considered in three ways: 1) individual HPWPs; 2) the 18 HPWPs were grouped into four categories representing practices that aim to develop employee ability, motivation, opportunity to perform and commitment to work, i.e. AMOC (Guest and Conway, 2007)<sup>6</sup>; 3) a cumulative index of the total number of HPWPs used i.e. HPWP intensity.

#### Individual HPWPs

The construction of the 18 HPWPs utilised in the assessment is presented below.

#### Sophisticated recruitment

Sophisticated recruitment is in place where either personality/attitude tests or performance/competency tests are conducted in filling vacancies within the largest occupational group (LOG).

#### Induction

A formal induction procedure is present if a standard induction programme normally lasting for at least 2 days (if counted in days) or at least 16 hours (if counted in hours) designed to introduce new non-managerial employees in the largest occupational group to the workplace.

<sup>&</sup>lt;sup>6</sup>Given that the results generated by both Principle Component Factor Analysis and Mokken Scale Procedure estimation showed that almost all HPWPs loaded on one factor across all firm sectors, it was therefore an arbitrary decision of the author over which HPWP should be allocated into each of the defined category.

#### Off-the-job training

A workplace is seen as providing its workforce with off-the-job training if over 60 per cent<sup>7</sup> of experienced employees in the LOG have been given time off from their normal daily work duties to undertake training over the past 12 months.

#### Performance appraisal

Over 60 per cent of non-managerial employees in the workplace have their performance appraised at least annually.

#### Performance-related pay

Over 60 per cent of non-managerial employees in the workplace are paid according to results or receive merit pay.

#### Profit-related pay

Over 60 per cent of non-managerial employees in the workplace receive profit-related payments or profit-related bonuses.

#### Internal labour market

An internal labour market exists in workplaces where preference is given to internal applicants, other things being equal, over external applicants. It also exists where internal applicants are the only source for filling vacancies.

#### Functional flexibility

Over 60 per cent of the LOG is formally trained to do jobs other than their own.

<sup>&</sup>lt;sup>7</sup>It has been a convention in WERS data user group that the presence of a practice in a workplace means most of the targeted workforce should take part in the practice asked about (see for example, Guest & Conway, 2007; Michie et al., 2008). The cut-off point for most of the targeted workforce in WERS is over and above 60 per cent.

#### **Teamwork**

Over 60 per cent of the LOG in the workplace is working in formally designated teams and either of the following conditions are met: team members depend on each other's work to be able to do their jobs; or team members jointly decide how the work is to be done.

#### Team briefing

Meetings held at least weekly between line managers or supervisors and all the workers.

#### Consultation committee

There are committees of managers and employees in the workplace, primarily concerned with consultation, rather than negotiation. These could include joint consultative committees, works councils or representative forums.

#### Employee attitude survey

A formal survey has been conducted regarding employees' views or opinions during the past two years.

#### Quality circles

Non-managerial employees are involved in problem-solving or quality circles or continuous improvement groups.

#### Benefits

A benefit scheme is in place if three or more of the following non-pay terms and conditions apply to the largest occupational group: employer pension scheme; private health insurance; more than four weeks of paid annual leave (excluding public holidays); sick pay in excess of statutory requirements.

#### Equal opportunity

Equal opportunity practices are present in the workplace if recruitment and selection have been monitored or reviewed to identify indirect discrimination by at least three of the following

characteristics: gender, ethnic background, disability and/or age; or if promotion procedures have been monitored or reviewed to identify indirect discrimination by at least three of the following characteristics: gender, ethnic background, disability and/or age.

#### Flexible working/family-friendly practice

Flexible working/family-friendly practices are deemed to be present if three or more of the following are practised in the workplace: working from home in normal working hours; job sharing schemes; flexitime where an employee has no set start or finish time, but has an agreement to work a set number of hours per week or month; any female employees taking maternity leave from the workplace receive their normal full rate of pay; working only during school term-time; workplace nursery or nursery linked with workplace; financial help with child-care and/or the care of older adults; and a specific period of leave for carers of older adults.

#### Grievance procedures

Formal grievance procedures are adopted if all of the following four items are in place: a) there is a formal procedure for dealing with an individual grievance raised by any employee at the workplace; b) employees are required to set out in writing the nature of the grievance procedures; c) employees are asked to attend a formal meeting with a manager to discuss the nature of their grievance; and d) employees have a right to appeal against a decision made under the procedures.

#### Job security

There is a recognised job security policy at a workplace if such policies apply to the largest occupational group.

#### **HPWP** bundles

To assess whether HPWPs have been introduced as coherent bundles within small, medium and large firms, principle component factor analysis was conducted. In the event, the results

showed that almost all of the 18 HPWPs demonstrated high loadings on only one factor, this being very similar to Guest and Conway's (2007) finding. As such, reflecting the growing consensus on the AMO framework underpinning the HPWP-performance link (Boxall and Purcell, 2008), this thesis adopts Guest and Conway's (2007) approach and allocates HPWPs into four different bundles, i.e. ability, motivation, opportunity to perform and commitment (the 'AMOC' framework). The reason for including another HPWP bundle representing commitment is because commitment is associated with lower labour turnover (Meyer and Allen, 1997) and 'organizations that have invested in their human capital will normally wish to retain that capital, and promoting employee commitment to the organization can contribute to retention' (Guest and Conway, 2007). The inclusion of the fourth bundle of HPWPs helps to capture the extent to which such an investment makes a difference to organisational performance.

The choice of which HPWPs are allocated to the same HPWP bundle is described as follows.

- The ability bundle includes practices that aim to improve the competence of the workforce via sophisticated recruitment, formal induction procedures and off-the-job training.
- The motivation bundle entails practices that are introduced as incentives to elicit high
  productivity among the workforce. These practices are an internal labour market,
  performance-related pay, profit-related pay and performance appraisal.
- The opportunity bundle comprises practices seeking to provide employees with the
  opportunity to perform and participate. These practices are teamwork, team briefing,
  consultation committees, employee attitude surveys, quality circles and the provision
  of training to enable workers to perform jobs other than their own (termed functional
  flexibility<sup>8</sup>).

<sup>&</sup>lt;sup>8</sup> In line with the configuration of the AMO framework (Boxall and Purcell, 2008) that employees should be provided with the opportunity to use their various skills and knowledge, many authors (Arthur, 1994; Huselid, 1995; Guthrie, 2001; Lawler, 1992) have included practices that seek to equip employees with multiple skills such as cross training and training focusing on future skill requirements. Hence, functional flexibility is allocated to the opportunity set.

 The commitment bundle consists of practices such as benefits, flexible working/family-friendly practices, equal opportunities, grievance procedures and job security.

#### **HPWP** intensity

A count measure evaluating the overall uptake of HPWPs is created by cumulating all the HPWPs used by a firm. High scores on this measure indicate a more intensive use and investment in these practices. Measures of this nature have been used in research on high commitment management (Arthur, 1994; Wood and de Menezes, 1998), high involvement management (Lawler, 1992; Wood and de Menezes, 2008) and high performance work practices (Appelbaum et al., 2000).

#### 3.2.3 Independent variables: small, medium-sized and large firms

In the maximum likelihood multi-variate analysis, the independent variables are three dummies representing small, medium and large firms respectively.

As mentioned in Chapter 1, there is no general consensus on the definition of small firms worldwide. Some previous studies define small firms as having up to 100 employers (Sels et al., 2006). A number of studies have adopted the US government stance which defines small firms as comprising up to 500 employees (Hendry et al., 1995; Heneman et al., 1999). However, in official statistics in the Organisation for Economic Co-operation and Development (OECD), European Union (EU) and the United Kingdom (UK), small firms refer to private sector firms employing fewer than 50 employees.

There is also no generally acknowledged definition for medium-sized firms within the limited research conducted on this sector. However, the OECD (2005: 17), the EU and the UK's Department for Business Innovation and Skills (Enterprise Directorate, 2006), define medium-

sized firms as private sector firms with between 50 and 249 employees. This definition has been used by a few recent studies (See, for example, Hoque and Bacon, 2008).

For the purposes of this thesis, small firms are defined as private sector firms employing 5–49 employees; medium-sized firms are defined as private sector firms with 50–249 employees; and large firms are defined as private sector firms consisting of 250 or more employees. Taking into account missing data, small firms account for 48.02 per cent of the weighted data, medium-sized firms account for 12.38 per cent and large firms account for 39.6 per cent. It is worth noting that workplaces within this categorisation can either be single independent workplaces (whereby the whole firm consists of a single workplace) or they can be part of a larger organisation. In these instances, it is the size of the larger organisation that is important in determining whether the workplace is classified as being part of a small, medium or large firm. For example, a small workplace with 10 employees is defined as a workplace within a small firm if the organisation of which it is a part has fewer than 49 employees. However, it will be classified as a workplace within a medium-sized firm if the organisation of which it is a part has between 50 and 249 employees, and a workplace within a large firm if the organisation of which it is a part has 250 or more employees.

#### 3.2.4 Control variables

Variables that may account for an association between HPWPs and performance, sometimes termed "third factors" (Wall and Wood, 2005: 443), should be controlled in order to generate reliable results, particularly in cross-sectional studies. The choice of which third factors are included is largely dependent on the nature of the research. A full list of control variables (dummies) included in this research is described as follows (See Appendix 1 for variable means).

#### Industry/sector

Many empirical studies on the relationship between HPWPs and corporate performance draw their data from manufacturing industries (Huselid, 1995; MacDuffie, 1995). However, one

might argue that the nature of the HPWP-performance relationship will be different in the service sector than in the manufacturing sector. This is because less task interdependence and high work-flow uncertainty might make specific HPWPs (such as training staff to develop good communication skills and to build up trust with customers) more appropriate for services than for manufacturing. Conversely, other HPWPs might be less appropriate in services (teamwork, for example). The impact on performance outcomes of the same set of HPWPs may therefore differ between industry sectors. A few previous studies report that the impact of HPWPs on firm performance in manufacturing firms is greater than that in services (Combs et al., 2006). As such, two dummy variables are created for the analysis in line with the 12 coded industries according to 2003 Standard Industrial Classification- manufacturing and non-manufacturing industries. Service industries are included in the non-manufacturing industries as any further categorisation would lead to a significant reduction in the number of observations in each industry.

#### Family ownership

Evidence from SME studies shows that it is less likely for a family-owned small firm to use HPWPs because these firms often opt for personal networks rather than formal practices (Ram and Edwards, 2003). Considerable diversity is also found in the adoption of HPWPs within family-owned small and medium sized firms (Duberley and Walley, 1995). The extensive literature on SMEs (see, for example, Westhead and Storey, 1996) suggests that the dominant role of personal links in most family-owned small and medium-sized firms can be a key determinant of low performance in these firms. As such, three dummy variables are generated around family ownership. Those firms in which a single individual or family owns at least 50 per cent of the firm and the controlling owners are actively involved in day-to-day management of the workplace on a full-time basis are defined as family-owned and owner-involved. Those in which a single individual or family owns at least 50 per cent of the firm but the controlling owners are not actively involved in day-to-day management of the workplace on a full-time basis are defined as family-owned but not owner-involved. Those firms in which a single individual or family owns less than 50 per cent of the firm are defined as not family-owned.

#### Firm age

Firm age is controlled for given that the extent of use of HPWPs may increase over time (Faems et al., 2005; Guthrie, 2001). Small firms are well known for a short life cycle and high failure rate, and this may in part explain the low uptake of HPWPs in small firms. It would therefore be expected that the longer a small firm stays in business, the higher the chance of the firm introducing more practices. Hence, firm age needs to be controlled to generate appropriate estimation of differences in the uptake of HPWPs in small and medium-sized firms in the analysis of the HPWP-performance link. The age of the firm is, therefore, categorised into four subgroups: 0–4, 5–9, 10–19, 20 and more years.

#### Union recognition

While trade union presence has been found to be associated with a greater use of HPWPs (Green, 1993; Jackson et al., 1989), extensive empirical studies also indicate that trade unions are associated with lower profitability or financial performance. It is argued that higher wages secured from collective bargaining result in the extraction of a greater share of rents at unionised establishments (Clark, 1984; Machin and Stewart, 1996). Although small firms normally have very low union recognition rates (Forth et al., 2006), the negative effect of trade unions on financial performance in small firms would be much more salient than in large firms given most small firms are short of alternatives to offset higher wages. The analysis therefore controls for union recognition. The union recognition variable is derived from three questions in the survey: a firm has union recognition if a union is recognised by management for negotiating pay and conditions for any sections of the workforce in the workplace; apart from the unions that have members in this workplace, there are other trade unions or staff associations that are recognised by management for negotiating pay and conditions for any sections of the workforce; or, if none of the workforce at this establishment are members of trade unions, there are trade unions or staff associations that are recognised by management for negotiating pay and conditions for any sections of the workforce.

#### Competitive environment

A highly competitive market appears to be more punishing for SMEs than for large firms. The constant erosion of margins resulting from high levels of competition makes SMEs less likely to use expensive practices such as HPWPs (Raymond and St-Pierre, 2005). Hence, a dummy variable for the degree of market competition is included in the analysis. Firms are viewed as operating in a relatively highly competitive market if they assess the degree of competition in the market as either very high or high; otherwise, firms are seen as having a relatively lower degree of competition in the market.

#### 3.2.5 Weights

The data are weighted throughout the analysis in order to account for the complex design of the WERS 2004 surveys. This weighting, which accounts for the probability of selection of the workplace into the main management sample, is essential if unbiased population estimates are to be obtained, as large workplaces are over-represented within the WERS 2004 stratified survey design.

#### 3.3 Results

The results generated by each of the assessments are described in sequence in this section.

### 3.3.1 Assessing the uptake of HPWPs in small, medium and large firms - Chi<sup>2</sup>tests

The results relating to individual HPWPs in Table 3.1 suggest that medium-sized firms are more likely to use 8 of the 18 HPWPs than small firms (at the 10, 5 or 1 per cent significance level). In contrast, medium-sized firms are less likely to use 9 of the 18 HPWPs than large firms. This suggests considerable differences in the approach taken to HRM between small, medium and large firms.

Table 3.1 The uptake of HPWPs by firm size in the private sector

	Small	Medium	Large	Full sample	Small vs.	vs.	Medium vs.
	(n=223)	(n=127)	(n=601)	(n=951)	Medium Chi <sup>2</sup>	Large Chi <sup>2</sup>	Large Chi <sup>2</sup>
Sophisticated recruitment	28.57		45.07	35.11	.9871	.0027	
Induction	36.67	44.82	61.64	47.55	.3041	.0000	.0367
Off-the-job training	26.73	52.44	53.01	40.31	.0007	.0000	.9436
Internal labour market	13.56	29.84	26.94	20.87	.0102	.0025	.7010
Performance-related pay	19.43	18.11	38.96	26.98	.8190	.0002	.0027
Profit-related pay	9.39	20.1	16.82	13.66	.0608	.0403	.6232
Performance appraisal	48.68	3 60.4	77.13	61.38	.1636	.0000	.0244
Teamwork	23.88	3 25.96	27.95	25.75	.7646	.1901	.7802
Team briefing	22.94	30.41	29.19	26.34	.2705	.3978	.8616
Consulting committee	4.03	3 10.3	12	7.96	.0310	.0018	.6208
Employee attitude survey	17.83	3 23.77	57.79	34.36	.3188	.0000	.0000
Quality circles	17.0	8.99	20.03	17.24	.0611	.4640	.0125
Functional flexibility	20.3	1 13.67	25.75	21.64	.3116	.2710	.0893
Benefits	16.8	4 36.36	58.28	35.64	.0027	.0000	.0063
Flexible working/family-friendly	17.93	3 23.19	22.08	20.22	.4032	.3411	.8647
Equal opportunity	10.1	4 24.04	25.25	17.84	.0143	.0004	.8681
Grievance procedures	17.3	9 33.17	52.05	33.05	.0130	.0000	.0195
Job security	12.7	7 6.6	12.49	11.9	.2374	.9420	.2499
Ability bundle	25.3	9 36.29	54.94	38.42	.1237	.0000	.0205
Motivation bundle	22.1	1 42.14	50.44	35.79	.0057	.0000	.3139
Opportunity bundle	8.8	4 15.68	25.26	16.18	.1203	.0000	.1100
Commitment bundle	5.1	4 13.51	24.82	13.96	.0383	.0000	.0956

Note: All estimations are weighted.

In relation to the uptake of specific bundles of HPWPs, the results suggest that small firms are less likely to use either the motivation or the commitment bundle than medium-sized firms, while large firms are more likely to adopt both the ability bundle (at the 5 per cent significance level) and the commitment bundle than medium-sized firms (at the 10 per cent significance level). Using the uptake of specific HPWPs as an indicator of primary HRM concerns, this would seem to suggest that, in comparison with small firms, medium-sized firms tend to be more likely to invest in practices that can either motivate the workforce or elicit greater

commitment from the workforce. Compared with medium-sized firms, large firms appear keen to develop the ability/competence of their workforce. Hence, the difference between small and medium-sized firms in relation to the specific bundles of HPWPs adopted seems not only greater than that between medium-sized firms and large firms, but there may also be specific reasons for these distinctions.

Overall, the *Chi*<sup>2</sup> test reveals that small firms differ from medium-sized firms in two ways in terms of the uptake of HPWPs: a) medium-sized firms are more likely to use 8 of the 18 practices than small firms; and b) medium-sized firms are significantly more likely to be using two specific bundles of HPWPs (the ability and the commitment bundles) than small firms. Hence, the question is whether firm size explains such variations in general across different firm sectors.

It is possible, however, that the differences between small, medium-sized and large firms will disappear once a range of observable characteristics are controlled for. As argued earlier in this section, identifying a true firm size effect requires the findings to remain statistically significant after a set of contextual factors are entered as control variables. As such, the next section of this chapter will assess the differences in the uptake of HPWPs between small, medium-sized and large firms within a multivariate analysis that controls for a range of observable characteristics. If the difference in the uptake of HPWPs between small and medium-sized firms or medium and large firms reduces, this will suggest that the differences identified above are the result of differences in the non-size related characteristics of small, medium and large firms and are not the result of size effects per se.

### 3.3.2 Comparing the uptake of HPWPs between small, medium and large firms using maximum likelihood multi-variate regression

The results of the multivariate analysis of the differences in the uptake of HPWPs in small, medium and large firms controlling for industry sector, firm age, firm size, family ownership, union recognition and degree of competition are reported in Table 3.2.

Table 3.2 Comparing the uptake of HPWPs between small, medium and large firms

Medium

Small

LargeFull sample

4.62

3.88

4.18

4.40

.0000

.0000

.0000

.0000

	(n=223)	(n=127)	(n=601)	(n=951)
Mean of HPWP intensity	3.642	4.909	6.625	4.978
	Small	Large	F	Prob.>F
HPWP intensity <sup>1</sup>	303(.096)***	.167(.080)**	15.01	.0000
Sophisticated recruitment	024(.217)	.286(.215)	2.61	.0039
Induction	108(.212)	.286(.215)	2.99	.0010
Off-the-job training	663(.208)***	204(.211)	3.97	.0000
Internal labour market	642(.231)***	042(.230)	2.30	.0113
Performance-related pay	.225(.211)	.530(.212)**	3.36	.0003
Profit-related pay	527(.274)*	162(.235)	1.87	.0457
Performance appraisal	296(.218)	.239(.225)	5.27	.0000
Teamwork	076(.223)	.092(.224)	.70	.7206
Team briefing	295(.211)	088(.212)	.67	.7565
Consultation committee	562(.215)***	090(.185)	5.50	.0000
Employee attitude surveys	398(.206)*	.815(.203)***	9.67	.0000
Quality circles	.339(.209)	.443(.212)**	1.15	.3199
Functional flexibility	.165(.254)	.329(.252)	1.09	.3673
Benefits	551(.213)**	.333(.208)	9.48	.0000
Flexible working/family-friendly	196(.222)	254(.223)	1.52	.1254
Equal opportunities	545(.244)**	099(.241)	4.74	.0000
Grievance procedures	625(.206)***	.384(.205)*	7.29	.0000
Job security	.360(.300)	.351(.316)	.41	.9431

Commitment bundle

Ability bundle

Motivation bundle

Opportunity bundle

-.262(.215)

-.544(.213)\*\*

-.438(.223)\*

-.542(.263)\*\*

.237(.214)

.120(.210)

.219(.210)

.304(.259)

By holding medium-sized firms as a reference category, the results in Table 3.2 in relation to HPWP intensity demonstrate that the absolute value of the coefficient (-.303) that represents the association between the HPWP intensity and small firms is larger than the coefficient (.167) representing the association between the HPWP intensity and large firms. This suggests

All estimations are weighted. Survey probit analysis for all individual HPWPs and HPWP bundles.

Survey Poisson analysis. Coefficients given. Standard errors in brackets.

All equations control for the industry sector, firm size, firm age, family ownership, union recognition, and the degree of market competition described in Chapter 2.

<sup>\*</sup> significant at 10 per cent. \*\* significant at 5 per cent. \*\*\* significant at 1 per cent.

the difference in the extent of HPWP intensity between small and medium-sized firms is greater than the difference between medium and large firms.

-

The findings relating to the adoption of individual HPWPs by firm size lends further support to the findings where the HPWP intensity is concerned. The results in Table 3.2 show that small firms are less likely to have adopted 8 of the 18 practices they were asked about than medium-sized firms. Large firms, in contrast, are more likely to use only 4 out of the 18 practices than are medium-sized firms. Given that a half of the differences between medium-sized and large firms disappear when the controls are added into the equation, the results of the multi-variate analysis indicate that some of the prima facie differences between medium and large firms are explained by differences in observable workplace characteristics that than by size *per se*.

This finding is further confirmed when the uptake of HPWPs is measured using the four HPWP bundles. As reported in Table 3.2, while no difference emerged between medium-sized and large firms in relation to their use of the HPWP bundles, the coefficients for both the motivation bundle and the commitment bundle indicate that medium-sized firms are significantly more likely to use these sets than small firms, and slightly more likely to use the opportunity bundle than small firms.

The limited differences between medium-sized and large firms also suggest that there may be a certain size threshold at which there is a step-increase in the number of HPWPs used but above which the extent of use of HPWPs does not increase dramatically. This issue is tested in further detail in the following section.

## 3.3.3 Assessing the firm size effect on the uptake of HPWPs in small, medium and large firms

To assess whether there is a threshold of formality in terms of the uptake of HPWPs across various firm sectors, the small, medium and large categories are further subdivided to provide a broader range of sub-categories. Within this analysis, only small firms operating on a single

site<sup>9</sup> are kept in the sample as there are no further cut points for firm size for small firms within the Workplace Employment Relations Survey (WERS) 2004<sup>10</sup>. Hence it is not possible to subdivide the small firm category in instances where the firm is operating on more than one site. However, for single independent small firms it is possible to subdivide further using the workplace size measure (given that in these instances workplace size is equivalent to firm size, or given that the firm is operating on a single site). Hence, small firms were divided into three subgroups: '5–9', '10–24', '25–49'; medium-sized firms were split into three subgroups: '50–99', '100-149' and '150–249'; and large firms were categorised into 4 subgroups: '250–1,999', '2,000–9,999', '10,000–49,999' and '50,000+'. Using this categorisation, the analysis will seek to demonstrate whether or not informality dominates all small firms, whether any variation in the uptake of HPWPs also occurs in medium-sized firms and whether there is a threshold above which there is a step-increase in the number of HPWPs adopted.

<sup>&</sup>lt;sup>9</sup> 84 per cent of small firms in the sample are operating on a single establishment.

Alternatively, all workplaces owned by small firms can be classified into different categories according to the size of the workplace and, thus, all valid cases for small firms would be kept in the sample, i.e. small firms operated on either a single workplace or multiple workplaces. Following the same logic as described in Section 3.2.3, those workplaces owned by either small, medium or large firms were categorised as "small firm workplaces", "medium-sized firm workplaces" or "large firm workplaces". Hence, small firm workplaces were divided into three subgroups: '5-9', '10-24', '25-49'; medium-sized firm workplaces were split into five subgroups: '5-9', '10-24', '25-49', '59-99' and '100-249'; and large firm workplaces were categorised into 7 subgroups, i.e.in addition to the five categories created for medium-sized firm workplaces, two more categories were created, i.e.'250-499' and '500+'. This analysis demonstrated similar results to those generated by the analysis based purely on firm size, i.e. the analysis presented in the thesis. This suggests categorisation based on workplace size is valid when it is done within the defined firm sector.

Another test was conducted on whether workplace size predicted the uptake of HPWPs as well as firm size. In this test, all workplaces were classified into different size categories purely based on the size of the workplace. This meant, for example, workplaces with up to 9 employees were grouped together irrespective of whether it was part of a small, medium or large organisation. Analysing these categories by firm size indicated that workplaces with "5-9" employees owned by large firms were likely to have a higher uptake of HPWPs than were similarly sized workplaces owned by small firms. Analysis of the other workplace size categories demonstrated similar variation driven by the size of the firm of which the workplace was a part. This would suggest that assessment purely based on workplace size could be misleading.

Table 3.3 Comparing the uptake of HPWPs between subcategories of firm size

	No. of Employees		1	2		3	4	ŀ
	5-9			089 (.131)	365	(.158)**	248 (	(.156)
Small	10-24	.089 (	.131)		276	(.138)**	159 (	(.130)
	25-49	.365 (	.157)**	.276 (.138)**			.118	(.158)
	50-99	.248 (	.156)	.159 (.130)	118	(.158)		
	100-149	.395 (	.149)***	.306 (.121)**	.030	(.155)	.148	(.145)
	150-249	.470 (	.197)**	.380 (.177)**	.104	(.203)	.222	(.191)
	250-1999	.491 (	.130)***	.402 (.092)***	.126	(.135)	.244	(.123)**
Large	2000-9999	.545 (	(.137)***	.456 (.097)***	.180	(.140)	.297	(.128)**
Large	10000-49999	.492 (	.129)***	.403 (.088)***	.127	(.135)	.244	(.122)**
	50000+	.750 (	.142)***	.661 (.106)***	.385	(.148)***	.502	(.136)***
		5		6	7		8	
	5-9	395 (	(.149)***	470 (.197)**	491	(.130)***	545	(.137)***
Small	10-24	306 (	(.121)**	380 (.177)**	402	(.092)***	456	(.097)**
	25-49	030 (	(.155)	104 (.203)	126	(.135)	180	(.140)
	50-99	148 (	(.145)	222 (.191)	244	(.123)**	297	(.128)**
Medium	100-149			074 (.182)	096	(.112)	150	(.115)
	150-249	.074 (	(.182)		022	(.165)	075	(.164)
	250-1999	.096 (	(.112)	.022 (.165)			053	(.080)
I	2000-9999	.150 (	(.115)	.075 (.164)	.053	(.080.)		
Large	10000-49999	.097 (	(.108)	.022 (.158)	.001	(.074)	053	(.071)
	50000+	.355	(.119)***	.280 (.166)*	.259	(.092)***	.205	(.087)**
		9		10	%	F	Prob.>F	N
	5-9	492	(.129)***	750 (.142)***	22.07			
Small	10-24	403 (	(.088)***	661 (.106)***	13.81			
	25-49	127	(.135)	385 (.148)***	3.86			
	49-99	244	(.122)**	502 (.136)***	5.04			
	100-149	097	(.108)	355 (.119)***	4.55	9.22	0	92
	150-249	022	(.158)	280 (.166)*	3.72	1.44	U	, ,
	250-1999	001	(.074)	259 (.092)***	13.2			
	2000-9999	.053	(.071)	205 (.087)**	9.14			
	10000-49999			258 (.082)***	13.69			
	50000+		(.082)***	•	6.41			

Note:
Survey Poisson analysis
Coefficients given. Standard errors in brackets.
All equations control for the industry sector, firm age, family ownership, union recognition, and the degree of market competition described in Chapter 2.

\* significant at 10 per cent. \*\* significant at 5 per cent. \*\*\* significant at 1 per cent.

The results of the assessment of the relationship between the uptake of HPWPs and firm size are presented in Table 3.3. Holding category constant in turn (5–9 in equation 1, 10–24<sup>11</sup> in equation 2 and so on), the results demonstrate that: first, small firms with 25 or more employees are more likely to have a greater use of HPWPs than other small firms (see equation 3). This suggests that there may be a threshold of 25 employees above which informality becomes untenable. There is no evidence that firms in the 25–49 employees size category use fewer HPWPs than do firms in the 10000-49999 category. Indeed, the uptake of HPWPs stays reasonably consistent across all firm size categories until it reaches the point where the number of employees increases to 50,000. Hence the differences between medium-sized and large firms may be driven by the greater uptake of HPWPs in very large firms rather than by a greater uptake of HPWPs across all large firms.

Second, the results also point to significant variation within the small firm sector (5-49 employees). This calls into question how analytically distinct this sector is. By contrast, there is relatively greater consistency in the extent of the uptake of HPWPs within the medium-sized firm category.

#### 3.4 Discussion and conclusions

The aim of this chapter was to assess the extent to which the uptake of HPWPs in small firms differs from that in medium-sized firms. The assessment was conducted in three steps. First, a bi-variate  $Chi^2$  analysis was conducted to ascertain whether any of the HPWPs are more widely used in medium-sized firms than in small firms. Second, a multi-variate analysis was undertaken to evaluate whether the differences in the uptake of HPWPs between small and medium-sized firms identified by the  $Chi^2$  test held when a set of controls for observable workplace characteristics were introduced. Third, dividing firms in each sector into several subcategories a multi-variate analysis was conducted to evaluate whether the uptake of HPWPs

<sup>&</sup>lt;sup>11</sup> Using the number of 24 employees rather than 20 employees (as suggested by Roberts et al., 1992) as a cut-off point in small firms is due to significant reduction of observations in the subcategory of 10–20 employees.

is consistent within firm size categorisations and whether there is a threshold at which there is a step-increase in the number of HPWPs adopted.

Taking the results in this section together, the assessments of the association between the uptake of HPWPs and firm size produced three notable findings. First, medium-sized firms tend to have a greater use of HPWPs than small firms in general, and there is a higher uptake of the motivation bundle and commitment bundle in medium-sized firms than in small firms. However, there is considerable variation between small firms, with small firms with 25 or more employees using as many HPWPs as do medium and large firms, suggesting formality starts to take root in firms that employ over 25 employees. Second, there is little variation in the uptake of HPWPs within the medium-sized firm sector. This homogeneity in the use of such practices may be an indication that medium-sized firms tend to have similar HRM concerns. Third, the differences in terms of the uptake of HPWPs between large and medium sized firms are less than the differences between small and medium-sized firms. In addition, the difference between large and medium-sized firms would appear to be driven largely by very large firms pursuing a high number of HPWPs rather than by differences between medium-sized firms and large firms as a whole.

Therefore, the variation in the uptake of HPWPs within the small firm sector suggests that formality, defined by the number of HPWPs adopted by a firm (Bacon et al., 1996; Storey et al., 2010), starts taking root once the size of the workforce reaches 25 employees. This suggests that for firms above this size threshold, the development of formal policies and procedures may be a natural response to the increase in size and the requirement to co-ordinate activities among employees within the firm (Greiner, 1972; Kotey and Slade, 2005; Roberts et al., 1992).

The results are also perhaps notable in that they do not endorse arguments in the literature that firms with fewer than 25 employees will not introduce any HPWPs whatsoever, or that they will find these practices to be of no value to them. On the contrary, small firms with 25 or more employees seem to use as many HPWPs as medium-sized firms and most large firms (except

those with over 50,000 employees). This may be because small firms with 25 or more employees need formal practices such as HPWPs to accommodate the increased complexity in their organisational structure (Roberts et al., 1992). Either way the results point to considerable heterogeneity within the small firm sector, with small firms with fewer than 25 employees adopting fewer HPWPs than small firms with 25 to 49 employees.

In contrast, there is a great degree of consistency in the uptake of HPWPs in medium-sized firms that appears to indicate that the medium-sized firm category may constitute a group of firms with very similar HRM concerns. These HRM concerns could be derived from the relatively large size of the workforce in these firms requiring a greater division of labour, hierarchy, formal policies and documented procedures (Kotey and Slade, 2005). It could also be the result of the need to engage in impression management as medium-sized firms need to adopt some HPWPs to convince external financers that the firm is being professionally managed.

Another notable finding with regard to the uptake of HPWPs is the apparent step-change in the number of HPWPs adopted once organisations cross a very large (50,000 employees) threshold. Large multi-national companies tend to operate a much more complex system in terms of distance and cross-culture management (Chang et al., 2009; Parry et al., 2008). The results presented here would seem to endorse this argument. It may be the case that a broader range of HPWPs is introduced to deal with the co-ordination problems involved in managing HR issues across different business units located worldwide.

# Chapter 4 The antecedents of the uptake of HPWPs in small and medium-sized firms

#### 4.1 Introduction

The reviews in Chapters 2 and 3 in relation to the factors that may potentially exert an influence on the uptake of HPWPs suggests that the increase in the use of these practices reported in small and medium-sized firms (Bacon et al., 1996; Forth et al., 2006) could be the result of a range of factors. For example, many government policies and government-sponsored initiatives aim to make SMEs aware of formal HR practices that may help them manage their employees more effectively (Goss et al., 1994), though whether such an increase would also lead to enhanced performance in these firms is not clear. In contrast, there are also factors that may be associated with a lower use of such practices in SMEs. For example, a highly competitive market is viewed as more punishing for small firms (probably medium-sized firms as well) than for large firms and may prevent SMEs from investing in HPWPs (Storey, 1994).

It is not however known to what extent these factors affect the use of HPWPs in small and medium-sized firms separately. As such, this chapter aims to analyse the extent to which certain observable factors are associated with the uptake of HPWPs in small and medium-sized firms. The assessment seeks to further the understanding of why variation in uptake exists between firms and to what extent these factors are able to predict the uptake of HPWPs in each firm sector.

This chapter is structured as follows. The next section (4.2) describes the research method. The ensuing section (4.3) reports the results. The final section (4.4) discusses the findings.

#### 4.2 Research method

Using the same data (i.e. the Workplace Employment Relations Survey (WERS) 2004 management questionnaire) and the definition for small, medium-sized and large firms described in Chapter 3, the assessment of the association between contextual factors, both external and internal to the firm, and the extent of adoption of HPWPs was conducted using survey Poisson estimation techniques. This is because the dependent variable is a count measure of all the HPWPs adopted by a firm, and Poisson analysis offers an ideal estimation (see Walters' (2007) discussion on why ordinary least squares (OLS) estimation is sub-optimal to Poisson analysis where a count measure is concerned). The remainder of this section illustrates the construction of variables used in the estimation.

#### 4.2.1 Dependent variable: HPWP intensity

The use of HPWPs in each firm was measured on a cumulative scale, as described in Section 3.2.2. This chapter seeks to examine whether the variation in the use of such practices between small, medium and large firms reported in Chapter 3 is the result of the contextual factors embedded in each of these firm sectors. It will evaluate whether the factors associated with the uptake of the overall number of HPWPs in medium-sized firms differs from that in small and large firms.

#### 4.2.2 Independent variables: contextual factors

The contextual factors to be assessed here include factors that are external and internal to the firm as described below.

#### External factors

Small and medium-sized firms are generally viewed as more vulnerable if located in a highly competitive market compared with their larger counterparts, as they have fewer resources to

manage such pressures. It is therefore important to assess whether certain observable factors external to a firm are associated with the uptake of HPWPs in small and/or medium-sized firms and to what extent such associations in small firms vary from those in medium-sized firms. The external factors included here are detailed below.

The external environment for a firm is regarded as competitive (coded as 1) if they answered either "Competitive" or "Highly competitive" to the question "How would you assess the degree of competition in this market?"; it was coded 0 if they answered otherwise.

Three dummies were created to explore whether providing products or services to a large and potentially dominant customer exerted potential influence on the adoption of HPWPs. The "Less than 5 per cent" category refers to those firms where only less than 5 per cent of the total annual value of goods or service was supplied to its large customers. This category also included firms that answered 'not applicable' to the question relating to the establishment's largest customer in terms of the value of goods or services supplied. The remaining firms fell into either of the following two groups categorised according to the proportion of the total annual value of goods or services supplied to its largest customer: 5-25 per cent and 26 per cent or more. According to the supplier-development view, larger and dominant customers may encourage their suppliers to invest in HPWPs (Hunter et al., 1996). The contrasting costminimisation view is that large customers will drive down the price of their suppliers and prevent them from investing more in HPWPs (Rainnie, 1989). As such, the relationship between the use of HPWPs in a small or medium-sized firm and whether it supplies a large proportion of its products or services to a single customer is unclear.

One dummy variable was created to evaluate whether the award of IiP accreditation in small and medium-sized firms is associated with a greater use of HPWPs. In line with Goss et al.'s (1994) finding, more HPWPs should be found in SMEs with IiP recognition as IiP may provide guidance and advice that may encourage firms to adopt HPWPs. However, it is also possible

<sup>&</sup>lt;sup>12</sup>These two categories were combined given no observations in the small firm sample fell into the "Less than 5 per cent" category and the number of observations in the medium-sized firm sample in this category is also low; it is therefore not possible to have this as a separate dummy.

that some SMEs may use IiP recognition as a marketing tool so that the IiP award may not necessarily lead to a higher use of HPWPs in such firms (Hoque and Bacon, 2008). It therefore remains to be seen whether IiP recognition will affect the uptake of HPWPs in small and medium-sized firms.

Establishments were grouped into three categories according to the number of business networks in which they participated: membership of one employers' network; membership of two or more employers' networks<sup>13</sup>; or not being a member of any of the employers' networks asked about. The five different groups of employers' networks included in the survey are: Employers' Association; Industry or Trade Association; Chamber of Commerce; Federation of Small Businesses; or some other similar group. Given active engagement with such business advisory networks is viewed as a catalyst to increase awareness of employment practices and can potentially improve efficiency among small and medium-sized firms (Goss et al., 1994), a positive association would be expected between the membership of these business networks and the uptake of HPWPs. However, it is also possible for those small and medium-sized firms to have no significant change in their use of such practices if firms are members of such networks but are not actively seeking advice or acting on the advice offered (Bennett and Robson, 2003). Thus, the association between participation in business networks and the uptake of HPWPs in small and medium-sized firms is currently unclear.

#### Internal variables

Apart from firm size, there are other contextual factors internal to the firm that may be related to a greater use of HPWPs. In addition to internal variables such as firm size, firm age, HR specialists, union recognition and corporate governance/proprietary control, which are constructed in the same way as described in Section 3.2.4 (in Chapter 3), there is one other factor internal to the firm that may also be associated with a greater use of HPWPs in small and/or medium-sized firms: firms with multiple workplaces may use more HPWPs than firms that operate on a single establishment. The construction of this measure is described below.

Only a small number of firms in either the small firm sample or medium-sized firm sample had participated in more than two of the employers' networks asked about in the WERS 2004 and they were grouped into the category 'membership of two or more employers' networks'.

A greater use of HPWPs would be expected in small and/or medium-sized firms with multiple workplaces in order to reduce potential principal-agent problems in such firms (Knocke and Kalleberg, 1994). In the analysis, while firms operating as a single independent establishment were defined as single-sited, those owning more than one establishment were defined as multisited.

#### 4.2.3 Control variables

All equations controlled for industrial sector.

#### 4.3 Results

The results of the assessment relating to the extent that contextual factors were associated with the uptake of HPWPs are described under the headings of "External factors" and "Internal factors" as follows.

### 4.3.1 The relationship between external factors and the uptake of HPWPs in small, medium and large firms

The results of the assessment of the relationship between external factors and the uptake of HPWPs in small firms are presented in Table 4.1. It shows that only IiP recognition was significantly linked to the uptake of HPWPs in small firms. There is no evidence that employers' networks have significant impact on the uptake of HPWPs in small firms, except the slightly greater use of HPWPs in those firms participating in two employers' networks (at the 10 per cent significance level). There is no evidence to suggest that the uptake of HPWPs in small firms is affected by the degree of market competition. In addition, having a large, dominant customer is not strongly associated with a higher uptake of HPWPs, though slightly greater use of these practices was found in those small firms where the largest customer

Table 4.1 The association between contextual factors and the uptake of HPWPs in small, medium and large firms

	Small	Medium	Large
External factors			
Degree of competition	124 (.116)	359 (.103)***	054 (.073)
Ref. Category: Largest customer <5%			
Largest customer ≥5-25%	.233 (.134)*	.226 (.131)*	.075 (.071)
Largest customer ≥26%	.017 (.139)	.038 (.124)	015 (.077)
IiP	.415 (.134)***	.382 (.139)***	.138 (.063)**
Ref. Category: Not a member of any emp	oloyers' network		
Member of 1 employers' network	.195 (.119)	.280 (.135)**	.002 (.072)
Member of 2 employers' networks	.297 (.176)*	.229 (.168)	.171 (.077)**
Internal factors			
Ref. Category: Small firm workplaces: 5	-9 employees		
10-24 employees	.095 (.124)		
25-49 employees	.271 (.150)*		
Ref. Category: Medium-sized firms 50-9	9 employees		
100-149 employees		.146 (.128)	
150-249 employees		.089 (.160)	
Ref. Category: Large firms 250-1,999 en	nployees		
2,000-9,999 employees			.062 (.086)
10,000-49,999 employees			.049 (.075)
50,000 employees			.231 (.100)**
Ref. Category: firms 0-4 years old			
5-9 years old	.298 (.229)	.210 (.271)	157 (.111)
10-19 years old	.176 (.210)	.099 (.259)	117 (.115)
20+ years old	059 (.212)	.039 (.253)	099 (.113)
Ref. Category: Not family owned			
Family owned/owner not involved	.063 (.247)	297 (.179)	205 (.099)**
Family owned/owner involved	084 (.130)	117 (.140)	049 (.114)
Single independent workplace	.318 (.159)**	093 (.114)	.033 (.098)
Union recognition	.411 (.177)**	.201 (.171)	.136 (.069)**
HR specialist	159 (.143)	.122 (.100)	.055 (.065)
F	6.140	5.710	3.370
Prob. > F	.0000	.0000	.0000
N	217	120	513

Notes: Survey Poisson analysis. Coefficients given, standard errors in brackets.
All equations control for industry sectors.
\*significant at 10%, \*\*significant at 5%, \*\*\*significant at 1%.

accounted for 5-25% of the total annual value of goods or service supplied (significant at the 10 per cent level). This suggests that the use of HPWPs in small firms is more likely to be driven by internal than external factors, although such a finding contradicts the widely accepted view presented in the literature.

The results for medium-sized firms produce a very different picture. In addition to a strong association between IiP recognition and the extent of uptake of these practices, a significantly higher level of uptake of HPWPs also emerged in medium-sized firms that had membership of only one employers' network. The uptake of HPWPs decreased dramatically in medium-sized firms where the external market was perceived to be highly competitive. This suggests that the uncertainties caused by market competition could have a significant impact on the HRM activities in medium-sized firms, possibly because managerial attention to the market diverts attention away from managing employees.

Turning to the results in large firms, two associations (both at the 5 per cent significance level) indicate that large firms with an IiP award were likely to have a greater use of HPWPs; as were those firms participating in at least two different employers' networks. There is, nonetheless, no evidence to suggest that either the degree of market competition or trading with the largest customer were associated with the uptake of HPWPs in large firms.

### 4.3.2 The relationship between internal factors and the uptake of HPWPs in small, medium and large firms

The results in Table 4.1 also demonstrate that small firms operating at single independent workplaces (sometimes termed "stand-alones") are likely to adopt more HPWPs than those with multiple workplaces. According to the positive correlation between firm size and the uptake of HPWPs in particular in the small firm sector, as reported in Chapter 3, this suggests that these stand-alones might have a relatively larger firm size than those operating at multiple

workplaces<sup>14</sup>. Second, there is a strong association between union presence and a greater use of HPWPs. However, there is no evidence that firm age, family ownership, or the presence of an HR specialist is associated with the uptake of HPWPs in small firms.

Where medium-sized firms are concerned, the results in Table 4.1 reveal that none of the internal factors were associated with the uptake of HPWPs. Linking this finding to the associations established between external factors and the uptake of HPWPs in medium-sized firms, this would appear to suggest that medium-sized firms are more sensitive to factors external to the firm than to internal factors.

In large firms, the uptake of HPWPs appears to be higher in firms with union recognition. In contrast to the results in small and medium-sized firms, there is a negative association between family ownership and the use of HPWPs in large firms, suggesting that family-owned large firms tend to use fewer HPWPs compared with those firms that are not family owned in general. Note however that although the association between family-owned large firms with family members involved in the management of the firm was not statistically significant, the negative sign of the coefficient nevertheless implies the use of HPWPs in such firms may also be lower than that in large firms that are not family-owned. Moreover, a much higher uptake of HPWPs was also reported by very large firms with at least 50,000 employees.

These findings apparently support the argument that the manner of HRM in small firms differs from that in medium-sized firms not only in terms of the great variation in the uptake of HPWPs, but the factors associated with such an uptake in each firm sector also differ.

<sup>14</sup> It is impossible to test whether it is the case given there are no cut-off points among small firms in the WERS 2004 data.

74

#### 4.4 Discussion and conclusions

The aim of this chapter was to assess which contextual factors, both external and internal to the firm, are associated with the use of HPWPs in small and medium-sized firms. In particular, the assessment sought to demonstrate whether the uptake of HPWPs in small and medium-sized firms is associated with different factors, because if the factors differ, this would further call into question the conventional wisdom of combining small and medium-sized firms into a single unit of analysis in both academic and policy circles.

The analysis demonstrated that the use of HPWPs in small firms was affected by different contextual factors from those in medium-sized firms. The only exception was the strong association between IiP recognition and a greater use of these practices, which applies to both firm sectors. This finding is in line with that reported by Goss et al. (1994), suggesting that the structure or blueprint IiP provides results in small and medium-sized firms adopting more HPWPs.

It is a notable finding that IiP recognition was associated with the overall uptake of HPWPs in small firms, while the engagement with various employers' advisory networks did not lead to significant increases in HPWPs in small firms. Arguably, small firms are usually less experienced and have a limited capacity to act on new knowledge (Nooteboom, 1993). Therefore, clearer direction from an accreditation institution such as IiP may assist them in introducing HPWPs. The IiP recognition process may well offer them a comprehensive and formal management framework compared to the relatively loosely organised information normally provided by various employers' associations and business advisory networks. This finding would appear to suggest that, although IiP recognition may not necessarily lead to an overall increase in training activities in the SME sector (Hoque and Bacon, 2008)<sup>15</sup>, the systematic procedure involved in the IiP accreditation may nevertheless provide an effective

<sup>15</sup> This thesis also conducted a similar analysis and found no relationship between IiP recognition and off-the-job training in either small or medium-sized firms.

route to access general practices aiming to improve employment relations and performance in the small firm sector.

Apart from the association between IiP recognition and the uptake of HPWPs, the findings also indicate that the uptake of HPWPs in small and medium-sized firms is associated with different factors. The rest of this section will discuss the implications of these findings.

In small firms, the associations between three internal factors and a greater use of HPWPs would seem to suggest that the extent of the uptake of such practices is more likely to be determined by the characteristics embedded within these firms rather than external pressures. These factors will be discussed in detail below.

The higher uptake of HPWPs in small firms with 25 or more employees compared to those with fewer than 25 employees lends support to the finding in Chapter 4, indicating that formality starts to take root when firm size becomes large enough to accommodate a complex organisational structure and requires more sophisticated HPWPs to help manage increased coordination problems (Barron et al., 2007; Roberts et al., 1992).

There is also a higher uptake of HPWPs in small firms that operate on a single establishment compared to small firms with multiple workplaces, suggesting that it is not necessary for small firms with multiple workplaces to introduce more HPWPs in order to deal with potential principal-agent problems (Knocke and Kalleberg, 1994). This result, nonetheless, suggests that in a small firm with 3 workplaces and 49 employees in total, for example, the numbers employed in each workplace would be small and hence we may anticipate more face-to-face communication in each of these workplaces compared to a small firm that operates on a single establishment but also has 49 employees. And thus, the former is less likely to require additional formal practices.

Another finding in small firms that is worth noting is the positive association between union recognition and the uptake of HPWPs in small firms. This would challenge the threshold for

statutory union recognition in the Employment Relations Act 1999, suggesting that if it was to be extended to small firms employing fewer than 21 workers (Ewing and Hock, 2003), the use of HPWPs in small firms is likely to increase, and to the extent such practices are associated with performance, thus improve performance in the small firm sector. While the government has to date resisted this move, such a finding may be of use to the union movement in making arguments for lowering this threshold. Given that small firms have the lowest rate of union recognition (3.41 per cent) compared to medium-sized (5.72 per cent) and large firms (32.67 per cent), the implication for government policy becomes far-reaching.

In contrast to the findings demonstrated in small firms, the association between a high degree of competition in the market and a lower use of HPWPs nonetheless suggests that medium-sized firms tend to be more responsive to their competitive environment than small firms. One possible explanation for this finding is that, with more resources at their disposal compared to small firms, medium-sized firms are likely to develop a buffer zone that enables them to adjust their HRM activities in line with such external changes. This would also suggest that medium-sized firms tend to have more resources compared to small firms that allows them to increase the use of HPWPs when the degree of competition in the market is low; a developing firm structure in which bureaucracy has yet not in shape in medium-sized firms compared to large firms provides medium-sized firms with a reasonable scope for being flexible in response to changes in the degree of competition in the market. Arguably, medium-sized firms may route these resources to meet other requirements when competition increases, while small firms lacking necessary resources are not able to do so.

Another finding that also differentiates medium-sized firms from small firms can be seen from the greater use of HPWPs in medium-sized firms that were a member of one employers' business network; but there was no such association where membership had been obtained from two or more different networks. In contrast, the linkage between the use of these practices and membership of one or two employers' associations was very weak in small firms (at the 10 per cent significance level). The discrepancy between small and medium-sized firms would seem to suggest that medium-sized firms are more experienced at seeking and receiving advice

and capable to absorb the advice that may be offered on employment relations issues compared to small firms, and the greater resources available are likely to help medium-sized firms to implement the advice and suggestions obtained from their participation in a business network. Hence, it is not how many business networks they have taken part in that appears to matter, it is the capability of transforming the kind of information they have gathered from such business networks into their operational framework that matters. However, small firms often lack the knowledge and experience that can help them to adapt these practices efficiently (Nooteboom, 1993), and thereby their participation in such business networks alone would hardly make any impact on the adoption of HPWPs in these firms.

What makes medium-sized firms distinct from both small and large firms is that none of the internal factors assessed were associated with the extent of uptake of HPWPs in medium-sized firms. Coupled with the strong associations demonstrated by three external factors, i.e. the degree of competition in the market, IiP recognition and participation in business networks, the findings seem to indicate that medium-sized firms tend to be more ready to act on the information gathered from outside of the firm rather than focusing on internal issues such as a de-motivated workforce with lower commitment.

Overall, the analysis in this chapter suggests that the factors associated with the uptake of HPWPs in small firms differ from those in medium-sized firms. While the use of HPWPs in small firms is more likely to be driven by internal factors, the opposite is true for medium-sized firms. The results partly explain the greater use of HPWPs in medium-sized firms than in small firms reported in Chapter 3. This could be because the use of such practices in medium-sized firms is greatly influenced by external factors. Hence, the results reinforce the suspicion that the uptake of HPWPs may not necessarily lead to better performance in medium-sized firms if such an uptake is mainly a response to external pressure rather than to address internal issues. The assessment of the relationship between HPWPs and performance to be conducted in the next chapter (Chapter 5) explores whether the HPWP-performance link holds in small and medium-sized firms separately.

# Chapter 5 The relationship between HPWPs and performance in small and medium-sized firms

#### 5.1 Introduction

To recap, the review of the debates on the relationship between HPWPs and performance offers three observations as illustrated in Chapter 2. First, the universal approach suggests the use of HPWPs will lead to improved performance irrespective of the context in which these practices are operating (Huselid, 1995). Second, firms adopting HPWP bundles (i.e. ability, motivation, opportunity and commitment, or termed "AMOC") perform better than those using individual HPWPs in isolation, as synergies are likely to occur where coherent HPWPs reinforce each other (Delery and Doty, 1996). Third, a strong HRM system in terms of the total number of HPWPs in place will have a great impact on performance (Bowen and Ostroff, 2004).

The review of the debates on the HPWP-performance link in small firms in Chapter 2 indicates that: a) individual HPWPs are more likely to be associated with improved performance than HPWP bundles/HPWP intensity in small firms – given the resource poverty faced by small firms they may be able to distinguish themselves from their competitors via the adoption of just a few HPWPs; b) in medium-sized firms the relationship between HPWP bundles and performance may be stronger than that between individual HPWPs and performance as the larger workforce in medium-sized firms may require several HPWPs in place to accommodate the complexity involved in managing greater numbers of employees; c) the relationship between HPWP intensity and performance in small and medium-sized firms is unclear for several reasons. First, not many small firms are able to afford such an investment. Second, in medium-sized firms, the potential tension between the internally driven demand for more HPWPs (to improve co-ordination, for example) and the introduction of more HPWPs due to

external pressure may combine to produce no positive impact on performance from the adoption of more HPWPs in medium-sized firms.

Turning to the assessment of the uptake of HPWPs conducted in Chapter 3, the results show that medium-sized firms tend to have a greater use of HPWPs than small firms in general, suggesting that a much stronger relationship between HPWPs and performance might be found in medium-sized firms compared to small firms. However, the findings generated by the assessment of the antecedents of HPWPs in Chapter 4 indicate that such a higher uptake of HPWPs in medium-sized firms could well be due to external pressures, suggesting that installing HPWPs solely in response to external requirements may distract managers from internal HRM concerns and thus result in no significant improvement in performance. Hence, the aim of this chapter is to explore the relationship between HPWPs and performance in small and medium-sized firms separately.

Thus, the next section of this chapter (5.2) describes the method used in assessing each hypothesis. The third section (5.3) reports the results. Finally, Section 5.4 discusses and summarises the findings.

#### 5.2 Research method

As the hypotheses developed for this chapter involve assessment of the HPWP-performance link in small and medium-sized firms from multiple perspectives, i.e. individual HPWPs, HPWP bundles, synergistic effect from the co-existence of different bundles of HPWPs and the overall relationship between HPWP intensity and performance, variables constructed for the assessment of each hypothesis also vary. The construction of specific variables used in each analysis is presented under the corresponding heading below.

#### 5.2.1 Dependent variables: performance measures

In order to assess the relationship between HPWPs and performance and the extent to which this relationship may differ in medium-sized firms compared to small and large firms, five performance measures are used as dependent variables. Of these, two are human resource (HR) outcome measures and three are firm-level performance measures.

#### **HR** outcomes

The two HR outcome measures are absence rate and voluntary labour turnover. Absence rate is measured by the percentage of work days lost through employee sickness or absence. Labour turnover is measured by voluntary resignations on the part of employees. Both of these HR outcome measures are continuous variables that are naturally bound between 0 and 1, thus ordinary least squares (OLS) is not the best linear unbiased estimator for them (Michie et al., 2008). Instead, these two variables are tabulated by quartiles and ordered probit is used. For absence rate the quartiles are: 1 = high absenteeism (>5.47%);  $2 = \text{medium-high absenteeism } (\ge 3.5 \text{ and } \le 5.47\%)$ ;  $3 = \text{medium-low absenteeism } (\ge 2 \text{ and} < 3.5\%)$ ; and 4 = low absenteeism (<2%). For labour turnover, the quartiles are: 1 = high labour turnover (>17.39%);  $2 = \text{medium-high turnover } (\ge 8.77\% \text{ and } \le 17.39\%)$ ;  $3 = \text{medium-low turnover } (\ge 2.5\% \text{ and} < 8.77\%)$ ; and 4 = low labour turnover (<2.5%).

#### Firm performance measures

Three firm-level performance outcome measures are used: labour productivity; quality of product or service; and financial performance. These are all subjective performance measures where respondents are asked to evaluate the performance of their workplace in comparison with other establishments in the same industry on a five-point scale (where 1='a lot below average' and 5='a lot better than average'). See Wall and Wood (2005) for a discussion on the appropriateness of using subjective performance measures. Given the nature of these measures, ordered probit analysis is used.

#### 5.2.2 Independent variables: individual HPWPs, HPWP bundles and HPWP intensity

The constructions of independent variables vary among the three approaches to assess the HPWP-performance link. Different sets of independent variables are used in each test as follows.

#### Testing the relationship between individual HPWPs and performance

The impact of individual HPWPs is examined to explore whether specific HPWPs are related to performance. By way of entering all of the 18 HPWPs simultaneously in each model, the association for each can be seen as the effect of that HPWP independent of the co-existence of others.

### Testing the relationship between HPWP bundles and performance in small, medium and large firms

This test includes two assessments. The first assessment focuses on whether small and medium-sized firms adopting specific HPWP bundle(s) perform better than other firms. The second assessment evaluates whether firms using more than one HPWP bundle, i.e. the achievement of internal fit, have a better chance of achieving superior performance compared to other firms. In the second assessment, small and medium-sized (as well as large) firms were further split into five subcategories, i.e. firms using none, one, two, three or all four HPWP bundles<sup>16</sup>. If positive associations were found between the greater use of such HPWP bundles and performance, this will provide support to the view that small (or medium-sized) firms adopting coherent HPWPs are more likely to have improved performance than other firms.

<sup>&</sup>lt;sup>16</sup> No small or medium-sized firms claimed the use of all four HPWP bundles.

### Testing the relationship between HPWP intensity and performance in small, medium and large firms

This test aims to evaluate whether the adoption of more HPWPs is associated with higher performance in small, medium and/or large firms. This is because it is argued that performance can be improved when HPWPs are used in a more comprehensive and widespread manner (Bowen and Ostroff, 2004; Youndt et al., 1996).

#### 5.2.3 Control variables

The set of control variables that might influence the relationship between the adoption of HPWPs and performance includes firm size, industry sector, firm age, family ownership, union recognition and the degree of competition in the present market respectively (see details of the construction of these control variables in Section 3.2 4).

Among these control variables, it is important to pay particular attention to firm size. As firm size has been widely reported to be positively correlated with the extent of the adoption of specific HPWPs (Forth et al., 2006) and large firms generally perform better than small firms (Machin and Stewart, 1991; Michie and Sheehan, 2001), one would expect unbiased estimation should be achieved by controlling a firm size variable based on the categorization between small, medium and large firms. Further evidence, nonetheless, suggests that differences in firm size also matters within the broadly defined size categories. For example, Roberts et al. (1992) report that informality prevails in firms with fewer than 20 employees, while elements of formality start to appear only when informal management dries up in a firm where the number of employees increases to 20 or more. This suggests that firm size should also be controlled for in an analysis that purely focuses on evaluating the relationship between HPWPs and performance in small firms. This also suggests that each firm sector needs to be further subcategorised. In line with this argument, small firms were split into three categories of 50–99, 10-24, 25-49 employees; medium-sized firms were split into three categories of 50-99,

100-149, 150-249 employees; and large firms were split into four categories of 250-1,999, 2,000-9,999, 10,000-49,999, 50,000 and more employees.

#### 5.3 Results

The results regarding whether individual HPWPs, HPWP bundles and/or HPWP intensity are associated with performance in small, medium and large firms, are reported separately below. The results testing for internal fit will be presented alongside the results for HPWP bundles in Section 5.3.2.

## 5.3.1 Assessing the relationship between individual HPWPs and performance in small, medium and large firms

The results in Table 5.1 suggest that only two HPWPs in small firms are significantly associated with performance (at the 5 per cent level): induction is associated with better quality of product or service, indicating that formal induction could help new recruits to manage the quality of product or service much better than those in small firms that had no formal induction. Similarly, those small firms that had an internal labour market were likely to have higher labour productivity, suggesting that employees in such firms could be highly motivated by a clear career structure and the potential for promotion as a reward for high performance.

Although some of the findings suggest the use of specific HPWPs may have a negative impact on performance in small firms, for example, the presence of a consultation committee was associated with high absenteeism and functional flexibility was associated with high labour turnover and high labour productivity, such associations were only significant at the 10 per cent level. Hence, the overall observation of the results in small firms is that very few individual HPWPs are positively associated with performance.

Turning to the results in large firms, however, more HPWPs (six in total) were positively associated with different performance measures than in small firms. Among these practices, three HPWPs were linked strongly to more than one performance measure simultaneously. These are sophisticated recruitment, off-the-job training and functional flexibility. By way of example, off-the-job training was associated with lower labour turnover, higher labour productivity (though at the 10 per cent significance level) and enhanced financial performance. The positive association between off-the-job training and high labour retention suggests that employees who received formal off-the-job training were likely to stay. One can speculate that this could be because trained employees are more committed to the firm or it is also possible the skills gained from such training were so firm-specific that they would be locked into the firm. Similarly, functional flexibility was positively associated with both lower labour turnover and better quality of product or service, indicating that multi-skilled employees were likely to stay in these large firms and the versatile knowledge and skills obtained from such training would generate better quality of product or service.

In addition to this, three other HPWPs were positively associated with different performance measures. Team briefing, for example, was associated with lower labour turnover, suggesting that updating employees with a wide range of operational or management information through regular team briefings increases workforce commitment and helps large firms to retain employees. The other two HPWPs are quality circles and flexible working/family-friendly practices, with the former associated with higher labour productivity and the latter with better quality of product or service.

Apart from these positive associations in large firms, there are also several occasions where individual HPWPs are linked to poor performance (each with only one performance measure). For example, induction is associated with lower labour productivity, while grievance procedures and profit-related pay are related to a higher absence rate; but profit-related pay is also associated with higher labour productivity and enhanced financial performance. Therefore, it is not yet clear whether such negative associations would affect the overall relationship

Table 5.1 The relationship between individual HPWPs and performance in small, medium and large firms

Small firms	Absen	Absence rate		Labour turnover		Labour productivity		Quality of product or service		Financial performance	
Sophisticated recruitment	.046	(.218)	.115	(.246)	.021	(.179)	115	(.185)	.053	(.184)	
Induction	.169	(.192)	.038	(.202)	.097	(.183)	.500	(.214)**	.093	(.197)	
Off-the-job training	.381	(.202)*	.046	(.214)	053	(.172)	.119	(.217)	098	(.199)	
Internal labour market	.355	(.231)	.419	(.268)	.561	(.255)**	.503	(.261)*	012	(.247)	
Performance-related pay	.390	(.216)*	.029	(.239)	.064	(.215)	060	(.226)	.080	(.222)	
Profit-related pay	075	(.306)	.295	(.307)	099	(.229)	059	(.314)	091	(.288)	
Performance appraisal	165	(.201)	314	(.224)	.175	(.221)	.185	(.207)	.236	(.203)	
Team working	035	(.228)	.088	(.225)	.144	(.205)	.092	(.214)	.078	(.187)	
Team briefing	125	(.210)	161	(.216)	.015	(.227)	085	(.219)	192	(.227)	
Consulting committee	804	(.424)*	226	(.485)	095	(.377)	276	(.603)	.434	(.453)	
Employee attitude survey	196	(.238)	133	(.259)	084	(.228)	.140	(.229)	.219	(.220)	
Quality circle	.419	(.228)*	.098	(.265)	.089	(.252)	334	(.253)	211	(.212)	
Functional flexibility	189	(.221)	445	(.238)*	.453	(.265)*	226	(.195)	.112	(.224)	
Benefits	001	(.253)	.097	(.261)	027	(.240)	.044	(.250)	168	(.239)	
Flexible working/family-friendly	176	(.252)	.270	(.286)	039	(.224)	004	(.259)	092	(.269)	
Equal opportunity	206	(.312)	.158	(.287)	.435	(.374)	.285	(.372)	.416	(.353)	
Grievance procedures	302	(.234)	.320	(.243)	.146	(.202)	083	(.223)	.157	(.207)	
Job security	284	(.279)	044	(.327)	.139	(.302)	258	(.250)	.392	(.250)	
F	1.320		1.150		1.070		1.050		1.750		
Prob.>F	.143		.280		.375		.410		.015		
N	223		223		223		223		223		

### (Continued)

Medium firms		Absence rate		Labour turnover		Labour productivity		Quality of product/service		Financial performance	
Sophisticated recruitment	119	(.354)	.230	(.338)	552	(.325)*	944	(.400)**	263	(.283)	
Induction	.181	(.328)	182	(.320)	.375	(.302)	.168	(.349)	.050	(.254)	
Off-the-job training	234	(.287)	.874	(.321)***	.766	(.307)**	.670	(.284)**	.483	(.320)	
Internal labour market	.288	(.331)	.675	(.365)*	.109	(.370)	044	(.362)	.439	(.364)	
Performance-related pay	.465	(.366)	.342	(.328)	.004	(.381)	.477	(.296)	206	(.343)	
Profit-related pay	.266	(.355)	.350	(.360)	.222	(.378)	128	(.408)	.289	(.316)	
Performance appraisal	356	(.306)	015	(.357)	.531	(.342)	250	(.344)	.027	(.401)	
Teamwork	.891	(.371)**	484	(.348)	232	(.372)	762	(.371)**	.100	(.352)	
Team briefing	605	(.295)**	063	(.295)	506	(.347)	.951	(.306)***	.163	(.319)	
Consultation committee	361	(.436)	298	(.359)	-1.239	(.503)**	696	(.501)	333	(.368)	
Employee attitude survey	.008	(.355)	.129	(.309)	004	(.377)	.117	(.348)	.041	(.304)	
Quality circles	.218	(.412)	987	(.432)**	625	(.405)	.229	(.507)	.481	(.393)	
Functional flexibility	941	(.389)**	216	(.398)	378	(.410)	501	(.339)	859	(.333)**	
Benefits	1.049	(.320)***	930	(.343)***	912	(.367)**	.288	(.375)	174	(.322)	
Flexible working/Flexible working/family-friendly	.546	(.395)	040	(.393)	.801	(.416)*	1.417	(.440)***	.380	(.412)	
Equal opportunities	011	(.378)	289	(.403)	711	(.373)*	403	(.352)	-1.200	(.372)***	
Grievance procedures	594	(.326)*	400	(.334)	.486	(.337)	.066	(.338)	099	(.393)	
Job security	450	(.470)	.520	(.542)	.341	(.652)	.466	(.594)	100	(.497)	
F	2.520	•	1.400		1.870	•	1.450		1.750	-	
Prob.>F	.000		.118		.013		.094		.024		
N	127		127		127		127		127		

#### (Continued)

Large firms	Absence rate		Labour	Labour turnover		Labour productivity		duct or service	Financial performance	
Sophisticated recruitment	.077	(.185)	.517	(.191)***	042	(.160)	.365	(.189)*	.194	(.183)
Induction	033	(.197)	040	(.196)	380	(.189)**	.168	(.179)	174	(.200)
Off-the-job training	110	(.186)	.506	(.179)***	.333	(.173)*	.292	(.182)	.390	(.186)**
Internal labour market	.096	(.185)	.192	(.198)	350	(.187)*	.108	(.189)	.049	(.164)
Performance-related pay	.304	(.185)	007	(.196)	.242	(.185)	303	(.176)*	.088	(.190)
Profit-related pay	481	(.240)**	095	(.267)	.610	(.210)***	.292	(.251)	.655	(.258)**
Performance appraisal	.116	(.219)	149	(.273)	.304	(.189)	.230	(.220)	104	(.197)
Team working	.113	(.183)	.082	(.196)	076	(.194)	031	(.208)	234	(.171)
Team briefing	.206	(.182)	.502	(.180)***	.107	(.163)	.243	(.216)	.190	(.177)
Consulting committee	284	(.185)	.001	(.191)	022	(.182)	.058	(.197)	230	(.215)
Employee attitude survey	114	(.201)	319	(.206)	.179	(.188)	<b>-</b> .191	(.212)	067	(.187)
Quality circle	105	(.194)	192	(.176)	.539	(.187)***	071	(.227)	.256	(.169)
Functional flexibility	.126	(.224)	.429	(.197)**	.226	(.206)	.417	(.195)**	.224	(.196)
Benefits	007	(.194)	.212	(.204)	339	(.183)*	130	(.208)	068	(.191)
Flexible working/family-friendly	150	(.211)	256	(.224)	.106	(.196)	.405	(.200)**	176	(.204)
Equal opportunity	.224	(.187)	.187	(.234)	.282	(.186)	049	(.180)	.064	(.168)
Grievance procedures	372	(.177)**	.077	(.176)	018	(.167)	037	(.177)	.113	(.159)
Job security	.000	(.260)	210	(.309)	.174	(.180)	.087	(.195)	.288	(.240)
F	1.590		3.460	-	2.500	•	2.280		1.200	• •
Prob.>F	.027		.000		.000		.000		.219	
N	601		601		601		601		601	

Notes:

Survey ordered probit analysis. All estimates are weighted.

Coefficients given (standard errors in brackets).

\*significant at 10 percent, \*\* significant at 5 percent, \*\*\* significant at 1 percent,

Control variables entered into each equation are dummies representing industries, firm age, firm size, family ownership/involvement, union recognition and competitive environment.

between HPWPs and performance in large firms. The assessment of HPWP intensity will be able to illuminate this.

The results for medium firms demonstrate a rather striking picture. While a few of these practices are consistently and positively associated with two or more different performance measures, several other HPWPs are consistently and negatively associated with two or more different performance measures. In medium-sized firms, off-the-job training was consistently associated with lower labour turnover, higher labour productivity and better quality of product or service; flexible working/family-friendly practices are associated with both higher labour productivity and better quality of product or service. These appear to suggest that medium-sized firms may enjoy better performance through the introduction of specific HPWPs, whereas small and large firms do not.

Furthermore, in contrast to small and large firms, a significant number of individual HPWPs (5 out of 18) are linked to poor performance in medium-sized firms. Specifically, three HPWPs are negatively associated with two or more performance measures: sophisticated recruitment is associated with lower labour productivity and poor quality of product or service; functional flexibility is associated with a higher absence rate and lower financial performance; and equal opportunities are linked to lower labour productivity and poor financial performance. Such negative associations appear to offset the positive associations established by other individual HPWPs, thus leading to a non-significant impact on performance in medium-sized firms.

In addition, there are also more HPWPs (3 out of 18) in medium-sized firms than in small (1 out of 18 practices significant at the 10 per cent level) and large firms (1 out of 18 practices) that are simultaneously associated with both improved performance in one relevant measure and lower performance in other measures. In medium-sized firms, teamwork is associated with a lower absence rate but poor quality of product or service; team briefing is associated with a higher absence rate and poor quality of product or service; and benefits are positively associated with a lower absence rate, but higher labour turnover and lower labour productivity. These relationships imply that no association between HPWPs and performance in medium-

sized firms would be discovered with respect to the measure of HPWP intensity as the positive impact produced by some HPWPs may well be offset by the negative impact generated by a few other practices. Overall, these findings suggest that certain unique characteristics in medium-sized firms may well influence the operation of HPWPs in a different way to those in either small or large firms, and thus would eventually affect their potential to improve performance in medium-sized firms.

Such results clearly cast doubt on whether the use of HPWP intensity would improve the performance in medium-sized firms. This will be tested in Section 5.3.3.

### 5.3.2 Assessing the relationship between HPWP bundles and performance in small, medium and large firms

The aim of assessing the relationship between HPWP bundles (i.e. ability, motivation, opportunity and commitment) and performance is to ascertain whether firms adopting specifically defined set(s) of HPWPs perform better than other firms. This section will, first, assess which HPWP bundle(s) is/are associated with improved performance in each firm sector; and second, assess whether firms using more HPWP bundles always perform better than those using fewer/no HPWP bundles, i.e. internal fit.

### i. Testing the relationship between HPWP bundle(s) and performance in small, medium and large firms

The results in Table 5.2 clearly demonstrate that the opportunity bundle is associated with both higher labour productivity and enhanced financial performance in small firms, although the use of the opportunity bundle is also linked to higher absenteeism rates. Given that none of the associations established between these individual HPWPs was statistically significant at the 5 per cent level or lower in Table 5.1, such strong associations developed by the opportunity bundle would seem to suggest that using these practices together reinforces the potential effect

to be produced by these practices, either in a positive way (i.e. enhanced financial performance) or a negative way (higher absence rates). What still remains to be seen is whether synergistic effects would occur when these opportunity practices are used in conjunction with other defined sets of practices. If more associations were found to be significant when multiple bundles of practices are used together, support would be lent to the argument that coherent HPWPs can generate synergy and yield higher outcomes over and above individual practices. This will be assessed later in the test of internal fit.

In the results concerning large firms, Table 5.2 shows that the motivation bundle is linked to enhanced performance while the ability bundle is associated with lower labour turnover. In contrast to the associations established between the opportunity bundle and performance in small firms, individual HPWPs that constitute either of these two HPWP bundles are themselves also related to improved performance in large firms. By way of example, sophisticated recruitment and off-the-job training are associated with lower labour turnover in large firms (see Table 5.1), both of which are components of the ability bundle. In contrast to small firms, the situation in large firms makes it difficult to summarise whether it is simply because these individual practices were in place or because a synergy had been generated by using such practices in combination that results in better performance. Given that, as shown in Table 5.1, more HPWPs had demonstrated strong associations with different performance measures individually in large firms than in small firms, this may imply that the pattern of the HPWP-performance link established in large firms differs from that in small firms. One implication is that large firms may benefit more from adopting a strong HRM system, i.e. HPWP intensity, rather than using specific HPWP bundle(s). Section 5.3.3 will test this.

Table 5.2 The relationship between HPWP bundles and performance in small, medium and large firms

	Abser	nce rate	Labour 1	turnover	Labour prod	luctivity	ity Quality of product or service		Financial performance		
Small firms											
Ability	.249	(.208)	.106	(.213)	040	(.174)	.151	(.227)	.117	(.194)	
Motivation	.337	(.487)	.566	(.334)*	.441	(.265)	.214	(.355)	173	(.421)	
Opportunity	788	(.270)***	<b>-</b> .273	(.285)	.464	(.232)**	.278	(.257)	.657	(.236)***	
Commitment	.134	(.491)	268	(.453)	.425	(.317)	.580	(.401)	.299	(.335)	
F	2.680		1.840		1.260	, ,	1.030	, ,	2.670	. ,	
$Prob. \ge F$	.001		.035		.232		.421		.001		
N	223		223		223		223		223		
Medium firms											
Ability	.089	(.269)	.012	(.298)	.474	(.304)	107	(.297)	.013	(.221)	
Motivation	1.041	(.321)***	.281	(.321)	997	(.428)**	.744	(.494)	424	(.353)	
Opportunity	115	(.327)	190	(.282)	366	(.290)	044	(.312)	.313	(.319)	
Commitment	526	(.503)	764	(.550)	926	(.442)**	-1.123	(.477)**	-1.574	(.380)***	
F	2.450	` ,	1.250	( )	2.300	( ' ' - '	1.810	(*****)	2.200	()	
Prob. > F	.005		.253		.008		.045		.012		
N	127		127		127		127		127		
Large firms											
Ability	198	(.191)	.387	(.193)**	.204	(.183)	.282	(.180)	.230	(.180)	
Motivation	121	(.263)	.049	(.241)	.341	(.199)*	.292	(.255)	.457	(.206)**	
Opportunity	.033	(.201)	.140	(.205)	.313	(.184)*	.085	(.193)	.004	(.177)	
Commitment	195	(.216)	.154	(.243)	.025	(.176)	021	(.197)	151	(.225)	
F	1.110	Ç <b>/</b>	3.920	( )	2.230	( )	1.750	· · · /	1.790	()	
- Prob.≥F	.344		.000		.005		.038		.033		
N	601		601		601		601		601		

Note: See estimation techniques in Table 5.1.

Ability bundle includes practices such as sophisticated recruitment, formal induction, off-the-job training; motivation bundle includes practices such as internal labour market, performance-related pay, profit-related pay, performance appraisal; opportunity bundle includes practices such as teamwork, team briefing, consultation committee, employee attitude survey, quality circles and functional flexibility, and commitment set includes practices such as benefits, flexible working/ family-friendly, grievance procedure and job security.

Turning to the association between specific HPWP bundles and performance in medium-sized firms, the results in Table 5.3 suggest that, apart from the positive association between the motivation bundle and a lower absence rate, the rest of the statistically significant associations established between specific HPWP bundles and performance are all negative. This suggests that using HPWP bundles could do more harm than good to performance in medium-sized firms. It is also notable that potentially there are more negative associations between HPWP bundles and performance in medium-sized firms than reported for small and large firms, although a majority of associations are not statistically significant. Linking these results to those generated by the individual HPWPs reported in Table 5.1, it becomes even more doubtful that HPWP intensity would be linked to better performance in medium-sized firms. This will be tested in Section 5.3.3.

## ii. Testing the relationship between internal fit and performance in small, medium and large firms

The test of internal fit aims to assess whether the progressive use of different HPWP bundles would improve performance in each firm sector.

The results of the internal fit test are presented in Table 5.3. It suggests that using more specifically defined HPWP bundles has a greater impact on enhancing performance in small firms than in large firms. By way of example, compared to those small firms adopting the highest number of HPWP bundles (i.e. three different HPWP bundles<sup>17</sup>, used as the reference group), small firms adopting either only two or none of the four HPWP bundles were significantly more likely to have higher absenteeism rates. Similarly, lower labour turnover and higher labour productivity were found in small firms making the greatest use of the four HPWP bundles compared to other small firms. Although some of the associations are not statistically significant, the unanimously negative signs of the coefficients all indicate that the progressive use of HPWPs would increase the chance of achieving higher performance in small firms.

.

<sup>&</sup>lt;sup>17</sup> See Footnote 16.

Table 5.3 The relationship between internal fit and performance in small, medium and large firms

Absenteeism rates		Labour turnover	Labour productivity	Quality of product/service	Financial performance		
Small firms							
Two-HPWP-bundle <sup>1</sup>	-1.008 (.350)***	875 (.405)**	544 (.369)	090 (.477)	006 (.395)		
One-HPWP-bundle	297 (.303)	348 (.387)	563 (.332)*	188 (.402)	095 (.338)		
Non-HPWP-bundle	669 (.276)**	568 (.376)	837 (.337)**	522 (.400)	253 (.341)		
F	2.450	1.420	1.250	1.320	2.460		
<i>Prob.&gt;F</i>	.004	.151	.246	.201	.004		
<i>N</i>	223	223	223	223	223		
Medium firms							
Two-HPWP-bundle <sup>2</sup>	.475 (.483)	.456 (.413)	.023 (.383)	674 (.478)	.130 (.507)		
One-HPWP-bundle	.560 (.365)	.079 (.355)	.493 (.445)	.012 (.366)	.802 (.508)		
Non-HPWP-bundle	.004 (.349)	.022 (.288)	.198 (.367)	.323 (.267)	.325 (.502)		
F	1.630	1.250	1.420	1.120	1.220		
Prob.>F	.089	.258	.164	.347	.277		
<i>N</i>	127	127	127	127	127		
Large firms							
Three-HPWP-bundle <sup>3</sup>	233 (.605)	088 (.348)	.321 (.287)	048 (.263)	.487 (.297)		
Two-HPWP-bundle	193 (.601)	126 (.341)	063 (.227)	.162 (.217)	.058 (.258)		
One-HPWP-bundle	419 (.617)	482 (.332)	.006 (.266)	210 (.269)	.213 (.273)		
Non-HPWP-bundle	066 (.621)	641 (.316)**	542 (.285)*	264 (.317)	087 (.281)		
F	1.610	3.420	2.090	1.590	1.440		
$Prob. \ge F$	.067	.000	.009	.071	.122		
N	601	601	601	601	601		

Note: See estimation techniques in Table 5.1.

1.2. Reference category="Three-HPWP-bundle"; Reference category="Four-HPWP-bundle".

Compared to a few strong associations in small firms, there is only one association that is significant at the 5 per cent level, and it suggests that large firms using all four HPWP bundles are more likely to have a lower labour turnover than those large firms using no such sets. One implication is that the greatest use of such HPWP bundles did not always bring about the best outcomes in large firms. Moreover, the positive sign of a few associations further casts doubt on the view that large firms would benefit the most from adopting a large number of different HPWP bundles.

Turning to the results for medium-sized firms, none of the associations in Table 5.3 is statistically significant. This suggests that the progressive use of HPWP bundles would barely make any difference on the performance in medium-sized firms when compared with a much lower use of such bundles. Such a result is not totally out of expectation in light of the negative associations developed between HPWP bundles and performance in Table 5.2. It nonetheless reinforces the previous suspicion that no relationship would be established between HPWP intensity and performance in medium-sized firms.

### 5.3.3 Assessing the relationship between HPWP intensity and performance in small, medium and large firms

The assessment of the relationship between HPWP intensity and performance is to ascertain whether a strong HRM system (as described by Bowen and Ostroff, 2004) will improve the performance in each firm sector.

The results in Table 5.4 demonstrate a strong association between HPWP intensity and higher labour productivity in small firms. HPWP intensity is also weakly associated with better quality of product or service and enhanced financial performance. In comparison with that established by the opportunity bundle as shown in Table 5.2, the marginally significant association with financial performance nonetheless suggests that simply pooling a few HPWPs from different HPWP bundles together could cancel out the positive effect on financial

Table 5.4 The relationship between HPWP intensity and performance in small, medium and large firms

	Absen	Absence rate		Labour turnover		Labour productivity		Quality of product or service		Financial performance		
Small firms												
HPWP intensity	.008	(.035)	.033	(.034)	.097	(.031)***	.063	(.035)*	.062	(.036)*		
F	1.740		1.810		1.600	•	1.150		3.290			
$Prob. \ge F$	.066		.054		.102		.327		.000			
N	223		223		223		223		223			
Medium firms												
HPWP intensity	.038	(.045)	053	(.046)	011	(.046)	.019	(.052)	003	(.053)		
F	1.780		1.450		1.620		2.350		.290			
Prob.>F	.066		.160		.103		.012		.987			
N	127		127		127		127		127			
Large firms												
HPWP intensity	022	(.035)	.066	(.027)**	.088	(.034)**	.083	(.027)***	.059	(.027)**		
F	1.240	•	4.830	•	2.730	•	2.630		1.790	•		
Prob.>F	.252		.000		.001		.002		.046			
N	601		601		601		601		601			

Notes:

Survey ordered probit analysis. All estimates are weighted.

Coefficients given (standard errors in brackets).

\*significant at 10 percent, \*\* significant at 5 percent, \*\*\* significant at 1 percent,
Control variables entered into each equation are dummies representing industries, firm age, firm size, family ownership/involvement, union recognition and competitive environment.

performance created by the opportunity bundle. Thus, small firms only appear to benefit little from just adopting more HPWPs. Alternatively, this would suggest that internal fit between HPWPs may have limited impact on performance in small firms.

The results of similar tests in large firms reveal some interesting findings. In Table 5.4, HPWP intensity is strongly associated with all the performance measures except absenteeism rates, suggesting that the adoption of more HPWPs is linked to enhanced performance in large firms. Compared to the evidence that HPWP bundles are not associated with performance (Section 5.3.2), the results generated by the HPWP intensity measure here would seem to imply that performance in large firms would improve when more HPWPs (i.e. a strong HRM system) are adopted.

The results of the assessment in medium-sized firms demonstrate that none of the associations are statistically significant. Another striking finding here is that the sign of the coefficients of these non-significant associations are all negative except for absenteeism rates and quality of product or service, indicating that a greater use of HPWPs might be linked to potentially poor performance in medium-sized firms. This contrasts with the results relating to small and large firms. Although the rest of the associations in small firms are not significant, the signs of the coefficients of the other performance measures are in the anticipated direction, i.e. a greater use of HPWPs is linked to better performance. There are significant differences, therefore, in the association between HPWP intensity and performance in medium-sized firms, compared to small and large firms.

#### 5.4 Discussion and conclusions

This chapter had three aims. Its first aim was to assess and identify which HPWP(s) is/are associated with performance in small, medium and large firms separately. The second aim was to evaluate whether firms that adopt specific HPWP bundle(s) or firms with a progressive use of these HPWP bundles were more likely to report higher performance than other firms. The

third aim was to assess whether a strong HRM system (i.e. HPWP intensity) is associated with higher levels of performance in every firm sector.

The testing of individual HPWPs suggests that, in small firms, there are very few HPWPs (specifically, only two) that are associated with improved performance. In medium-sized firms, some HPWPs are positively associated with performance, some are negatively associated with performance and a few others are simultaneously associated with both better performance in one performance measure and poor performance in other performance measures. In large firms, there are more HPWPs that are individually associated with improved performance than in small and medium-sized firms, though the proportion of such practices among the total HPWPs asked about remains low (6 out of 18).

The testing of HPWP bundles indicates that small firms with internal fit are more likely to achieve performance superior to the other small firms (to be specific, improved performance has been reported in performance measures such as absence rates, labour turnover and labour productivity). Small firms using the opportunity bundle are more likely to improve their financial performance; in medium-sized firms, performance would not benefit from HPWP bundles; and in large firms, the benefit from HPWP bundles is very limited.

The testing of HPWP intensity (i.e. a strong HRM system) suggests that HPWP intensity can mildly improve performance in small firms in general (to be specific, significantly improved labour productivity but marginally improved quality of product and financial performance). There is no relationship between HPWP intensity and performance in medium-sized firms. In stark contrast, HPWP intensity is strongly associated with all performance measures in large firms except for the absence rate.

It is clear, therefore, some similarities between small and large firms with regard to the HPWP-performance link exist, while medium-sized firms somehow differentiate themselves from both small and large firms. The rest of this section will first discuss these results in

relation to small and large firms and then move on to deliberate on the results in medium-sized firms.

#### Small firms

Small firms appear able to benefit the most from adopting a specific HPWP bundle (i.e. the opportunity bundle) in terms of higher economic return, and the economic return can only be improved mildly when HPWP intensity is used. Compared to these, the performance effectiveness of individual HPWPs in small firms is very limited. Only two (out of 18) HPWPs are associated with improved performance in small firms: a formal induction programme is associated with better quality of product or service, and an internal labour market is associated with higher labour productivity. This finding is in marked contrast to that in large firms where six HPWPs were associated with enhanced performance, given that more individual HPWPs associated with performance should have been expected in small firms than in large firms. This contradicts the argument that the impact of individual HPWP on performance would be more direct in small firms than in large firms as they are more likely to be engaged in frequent direct communication and have flatter hierarchies (Bacon et al., 1996; de Kok and Uhlaner, 2001).

The strong associations demonstrated between the opportunity bundle and higher labour productivity, enhanced financial performance and higher absenteeism rates are notable findings. Given that none of the HPWPs allocated to the opportunity bundle is associated individually with financial performance, the enhanced financial performance would seem to suggest that using a few HPWPs belonging to the opportunity bundle together may assist in targeting and tackling a specific HRM problem in a reinforcing manner and thus subsequently increasing the economic return in small firms.

That said, small firms need to be aware of the fact that negative associations may also occur and may eventually reduce the benefit gained. Although the precise reason for such a negative association is beyond the scope of this analysis, the negative signs of associations between the individual HPWPs belonging to the opportunity bundle (except for the quality circles) and absenteeism rates, although not statistically significant, nonetheless suggest that these practices

potentially might have similarly negative associations with performance. Such associations would become significant when the set of HPWPs were used together. It could be possible that practices such as training multi-skilled employees to perform jobs other than their own and/or work in teams may well stretch the available resources in small firms and thus lead to more employees being absent from work. If informality prevails in the small firm sector, intensive overlaps between jobs/tasks may create tension on the shop floor, make jobs much more stressful and thus result in greater sickness leave. Indeed, it would seem to indicate that the adoption of HPWPs always comes with a cost and it is whether the gains from such an adoption would eventually compensate the cost that matters (Sels et al., 2006; Way, 2002).

In testing whether small firms would benefit more from adopting a strong HRM system, i.e. many different HPWPs, both of the high absence rates and enhanced financial performance linked with the opportunity bundle disappeared when it was used in conjunction with other specifically defined HPWP bundles and combined into a single index of HPWPs, i.e. HPWP intensity. Given that the HPWP intensity measure is only strongly associated with labour productivity (at the 1 per cent significance level), it would nevertheless imply that, constrained by resource poverty, small firms probably need to select HPWPs to deal with HRM priorities so that HR investment would mainly target such problems (Hoque and Bacon, 2006; Wager, 1998). By doing so, small firms may be able to improve their management and develop a committed workforce and eventually yield higher economic returns.

#### Large firms

Compared to the findings in small firms, the results for large firms suggest that large firms can benefit more from individual HPWPs (6 out of 18 in large firms compared to 2 out of 18 in small firms) and thus combining these practices would generate a complementary effect and improve performance in large firms (the HPWP intensity measure is consistently associated with all performance measures except the measure of absenteeism rates in large firms).

What also makes the HPWP-performance link in large firms differ from that reported in small firms is that using HPWP bundles can improve performance in large firms, but in a very mild way. With only the ability bundle associated with lower labour turnover and the motivation

bundle associated with enhanced financial performance, the performance effectiveness of HPWP bundles in large firms is rather limited in comparison with that in small firms. Nonetheless, this seems to suggest that large firms may be highly concerned about overall impact of HPWPs on performance. A possible interpretation for this is that the widespread use of these HPWPs may well have already constructed routine procedures and thus balancing the development of practices in different HRM domains would be the priority, thereby it is likely to be a case of whether enough practices like these have been adopted (Boxall and Purcell, 2008). This could be because large firms are more experienced in providing their employees with opportunities to participate in decision-making in relation to employment relations issues. Hence, the workforce would be more likely to perceive the presence of sufficient policies and practices as a message conveying genuine trust and fairness from the top, which in turn would affect individual performance and subsequent performance at the firm level as a whole. It could be, therefore, that the overall coverage of these practices is essential for large firms to achieve better performance than others.

#### Medium-sized firms

In marked contrast to the HPWP-performance link established in small and large firms, using more HPWPs in medium-sized firms appear to barely improve performance in these firms, it is using the right HPWP that matters. Although medium-sized firms have a few HPWPs that are positively associated with enhanced performance across various performance outcomes, medium-sized firms also have the largest number of HPWPs linked to poor performance across different performance measures and the largest number of HPWPs simultaneously associated with both better performance in one performance measure and poor performance in other performance measures. Although little is known about these processes, the results seem to suggest that in medium-sized firms the performance benefits of HPWPs to develop the company could be offset by the negative impact on the performance of other practices introduced to manage difficult HRM problems. No matter whether such a finding could be an indication of 'deadly combination' (Delery, 1998), it would definitely offer great potential for future research.

Indeed, an overall negative effect on performance would occur if the positive impact is outweighed by the negative impact. It could also be the case that using these HPWPs together may reinforce the potential negative effect and subsequently damage performance in medium-sized firms. These would indicate that the adoption of HPWPs *per se* is unlikely to solve all of the problems medium-sized firms face; it is adopting the right HPWPs that matters.

An explanation of why the use of HPWPs can barely improve performance may reside in a dilemma that most medium-sized firms are likely to face. That is, medium-sized firms may stretch their resources to introduce new HPWPs if they also suffer from resource poverty as small firms do (Storey, 2002). In addition, medium-sized firms also face increasing problems of co-ordinating work between an expanded workforce (Greiner, 1972). As medium-sized firms may require external funding to grow, these firms may demonstrate their credibility and accountability to external financers by showing plans for formalising and professionalising their management either through installing more HPWPs (Frormann, 2006) or hiring new CEOs who are experienced in developing formal HRM practices and are keen to adopt more HPWPs (Baron et al., 2007). It is also possible that medium-sized firms would attempt to adopt such practices as a universal approach to improve their performance, and would be able to convince potential investors that their investment would guarantee high returns. Nevertheless, the former strategy is likely to result in tension where HPWPs are adopted to impress external funders rather than to address the need to deal effectively with coordinating the work of employees and managing their motivation and commitment.

Another explanation could be that medium-sized firms may be experiencing growing pains. The reason for such a conjecture is that medium-sized firms are more likely to confront coordination problems given the expanded size of their workforce in comparison to small firms (Baron et al., 2007; Greiner, 1972; Kotey and Slade, 2005). Although they are larger than small firms, their size is still not large enough to justify adopting a wider range of HPWPs. However, frequent face-to-face communication may no longer be effective or sufficient to manage employees in most medium-sized firms.

Taking these discussions together, this chapter details three notable findings. First, the relationship between individual HPWPs and performance is not universal. Although more individual HPWPs are associated with improved performance in large firms than those in small and medium-sized firms, only a small proportion of such practices are associated with better performance across different firm sectors. Moreover, in medium-sized firms, some HPWPs are positively associated with performance and some others are negatively associated with performance, implying the HPWPs that medium-sized firms adopt do not necessarily help managers to resolve a complex range of problems The results do not necessarily support Delery's (1998: 294) argument that "deadly combination" of HPWPs would damage firm performance, they suggest that choosing the right practices is essential for medium-sized firms. Such choices may be vital because medium-sized firms are in a development stage where specific concerns emerge and HR practices are required to target these particular problems to assist medium-sized firms to evolve from one development stage to the next (Greiner, 1972).

Second, enhanced financial performance is reported by small firms when the set of HPWPs that seek to enhance the opportunity for employees to participate are used together. Small firms using different HPWP bundles in combination (i.e. internal fit) also have a better chance to achieve higher performance in a number of performance measures compared to other small firms, although such a use does not necessarily lead to higher financial performance. This could be because HRM problems are more readily identified as relating to specific HRM domain(s) in small firms, given the less complex structure in these firms compared to that in medium and large firms.

Third, while large firms require a strong HRM system to improve their performance, the relationship between HPWP intensity and performance in small firms is unclear (except for higher labour productivity) and there is no relationship between HPWP intensity and performance in medium-sized firms. This could be an indication that encouraging small firms to use as many HPWPs as found in most large firms may stretch the already limited resources in small firms and thus result in no significant improvement in financial performance (Storey,

2004), while in medium-sized firms, potential tensions from conflicting HPWPs in operation could well cancel out any positive impact some practices would have on performance.

Overall, the findings in this chapter lend support to the argument that medium-sized firms are not simply 'scaled-down' versions of large firms (Storey, 2002: 250), or scaled-up versions of small firms. The characteristics embedded in medium-sized firms appear to result in a unique pattern of HPWP-performance linkages distinct from that demonstrated by other firms.

This provides further support to the argument that small and medium-sized firms are likely to have different HRM concerns (Hoque and Bacon, 2008). The different pattern of HPWP-performance links in small compared to medium-sized firms presented in this chapter suggests that in future, research on small and medium-sized firms should be analytically split into two distinct groups in order to generate a better understanding of whether and how the theory and principles of HRM developed in large firms would apply to different firm sectors. The implications for government policies that aim to offer effective business support to SMEs will be discussed in the next chapter.

### Chapter 6 Conclusions

As discussed in Chapter 2, this thesis had three aims. Its first aim was to examine the extent to which the uptake of HPWPs in small firms differs from that in medium-sized firms. The second aim was to assess whether the contextual factors associated with the uptake of HPWPs in small firms differ from those in medium-sized firms. The third and final aim of the thesis was to explore whether there is a relationship between HPWPs and performance in small and medium-sized firms and, if so, whether such a relationship follows a different pattern in small firms compared to medium-sized firms. Any differences to be revealed in these analyses would add to the argument that small and medium-sized firms should be analytically split into two independent groups in future research. The analysis in the thesis generated several key findings.

## 6.1 The difference in the uptake of HPWPs between small and medium-sized firms

As stated in Chapter 2, several authors suggest that HPWPs are used in a piecemeal manner in SMEs (Storey, 2002; 2004; Storey and Westhead, 1997). However, the positive correlation between the uptake of specific HPWPs and firm size reported in a few studies (Forth et al., 2006; Kotey and Slade, 2005) suggests that medium-sized firms are likely to use more HPWPs than small firms and firm size thereby appears to be the key determinant of the variation in the use of such practices between small and medium-sized firms. Nonetheless, the assessment in Chapter 3 yielded several findings that contradict what the literature would lead us to believe.

First, in the assessment of the uptake of HPWPs between small, medium and large firms where individual HPWPs are concerned, medium-sized firms were more likely to use eight HPWPs than small firms and were less likely to use eight HPWPs compared to large firms. However, the number of HPWPs that were more often used by large firms compared to medium-sized

firms reduced to four after controlling for a range of variables. This suggests that there are factors other than firm size that may also explain the lower uptake of such practices in medium compared to large firms. Firm size appears to account for the variation in the uptake of HPWPs between small and medium-sized firms to a much greater degree than it does between medium and large firms.

Turning to the assessment where HPWP bundles are concerned, the findings suggest that medium-sized firms compared to small firms were more likely to use two HPWP bundles, i.e. the motivation bundle and the commitment bundle, and had a slightly greater use of the opportunity bundle. On the contrary, no such difference was found between medium and large firms. These findings would seem to suggest that medium-sized firms may have difficulties particularly in these two HRM domains compared to small firms, but that they do not have many distinct HR concerns from those of large firms.

The evidence provided on HPWP intensity suggests the difference in the extent of the uptake of HPWPs between small and medium-sized firms is greater than that between medium and large firms. This implies firm size is more likely to explain the differentiation in the extent of uptake of HPWPs between small and medium-sized firms than it does between medium and large firms.

Overall, these findings suggest that, although medium-sized firms have fewer employees than large firms, the available resources may still allow them to benefit from economies of scale compared to small firms, thereby leading to a higher uptake of such practices than small firms (Lynch and Black, 1998). Hence, the evidence presented here does not suggest that small and medium-sized firms suffer equally from resource poverty.

Second, there is considerable formality in the small firm category and the threshold of formality in small firms is firms with 25 employees. The evidence shows a step-increase in the uptake of HPWPs first emerges when the size of the firm passes the threshold of 25 employees and the second step-increase does not appear until firm size becomes very large, i.e. 50,000 or

more employees. Hence, firms with as few as 25 employees may well have as many HPWPs as are often found in many large firms in order to accommodate a more complex organisational structure than smaller firms (Roberts et al., 1992). One possible conclusion is that formality starts to take root once the size of the firm passes this threshold of 25 employees. Such results also indicate that the difference in the use of HPWPs between medium and large firms is mainly driven by very large firms, i.e. firms with 50,000 or more employees. Overall, the results suggest the use of HPWPs in small firms is not completely *ad hoc* (for example, formality develops in small firms with 25–49 employees), but it is in part a response to workforce size. Nevertheless, these findings need to be interpreted with caution as the analysis was based on the sample of small firms that were standalones. (See justifications for this analysis in Footnote 8 and Footnote 9.)

Finally, there is a great degree of consistency in the uptake of HPWPs within the medium-sized firm sector and a notable heterogeneity within the small firm sector. The consistently higher uptake of HPWPs in medium-sized firms, coupled with the much greater use of the motivation bundle and commitment bundle compared to small firms (but not lower than large firms), would suggest that medium-sized firms may have similar HRM concerns and therefore have a tendency to use similar numbers of HPWPs and also probably adopt very similar HPWPs (Greiner, 1972), i.e. a set of HPWPs that seeks to motivate the workforce and a set of HPWPs that seeks to elicit high commitment among employees.

## 6.2 The difference in the antecedents of the uptake of HPWPs between small and medium-sized firms

In reviewing factors potentially associated with the uptake of HPWPs in Chapter 2, previous literature suggests that several external factors, for example, a competitive market, would be associated with a lower uptake of HPWPs in small firms, while some other external factors, such as the presence of larger customers would be associated with a higher uptake of HPWPs

in both small and medium-sized firms. Meanwhile, a range of internal factors, for example, firms with HR specialists in place and firms not owned by a family, were assumed to be associated with a higher uptake of HPWPs in both small and medium-sized firms. The analysis in Chapter 4 has generated findings in stark contrast to these hypotheses.

Results in Chapter 4 suggest that factors associated with the uptake of HPWPs in small and medium-sized firms are different in most circumstances. Apart from IiP recognition which demonstrates a significant association with the use of HPWPs in both small and medium-sized firms, all the other factors associated with a greater use of HPWPs in small firms were not associated with HPWPs in medium-sized firms. These findings will be discussed below.

#### Small firms

Apart from IiP recognition, no other external factors were associated with a greater use of HPWPs in small firms.

The association between IiP recognition and the uptake of HPWPs in small firms supports Goss et al.'s (1994) finding that IiP accreditation can help small firms to increase their uptake of HPWPs. It also suggests that IiP accreditation may well offer them a comprehensive and formal management framework compared to the relatively loosely organised information normally provided by various employers' associations and business advisory networks.

In contrast to the findings in relation to external factors, the use of HPWPs in small firms was more likely to be influenced by several internal factors. The positive association between union recognition and the uptake of HPWPs in small firms suggests that encouraging unionisation in the small firm sector could also enhance the uptake of HPWPs. Given the threshold for statutory union recognition in the Employment Relations Act 1999 is small firms employing at least 21 workers, it would be advisable for government to extend the threshold if it wished to increase the use of HPWPs (Ewing and Hock, 2003).

There is also a positive association between stand-alone workplaces and a greater use of HPWPs in small firms, suggesting that small firms with multiple workplaces do not encounter significant principal-agent problems as frequent face-to-face communication in small firm workplaces tends to make the use of HPWPs redundant.

That said, no evidence suggests that small firms with HR specialist in place had a greater use of HPWPs compared to those firms with no HR expert in place. This seems to imply that, although HR specialists have the knowledge, skills and experience related to effective employment relations management, HR specialists in small firms may find the scope for them to implement effective HR practices such as HPWPs is very limited given resource poverty. There is also no evidence that family-owned small firms tended to use fewer HPWPs compared to small firms that are not family-owned. This finding contradicts much of the small firm literature. One possible interpretation of the lack of a family-ownership effect is that resource poverty simply prevents small firms from adopting more HPWPs regardless of ownership.

### Medium-sized firms

In contrast to the findings that no external factors were associated with the use of HPWPs in small firms, in medium-sized firms several external factors were associated with the uptake of HPWPs.

First, a high degree of market competition is associated with a lower use of HPWPs, suggesting market uncertainty may increase the risk of medium-sized firms investing in HPWPs.

Second, there is a relationship between participation in one business network and the uptake of HPWPs, suggesting that medium-sized firms may be more capable at adapting to the advice that may be offered compared to small firms. In part, this may occur because they have more resources to devote to HR issues, they may have higher levels of management capability, or they may be more receptive to advice because they face more complex HR challenges than small firms. Alternatively, this appears to suggest the advice provided through these networks

needs to take into account the possibility that firms in these two size categories require very different advice and guidance.

Third, there is also a positive relationship between IiP accreditation and the uptake of HPWPs, suggesting that medium-sized firms act upon the information and advice gained through the preparation for IiP accreditation. This also suggests that medium-sized firms may have specific HRM concerns and they are keen to search for solutions and take action quickly to address these concerns.

However, none of the internal factors demonstrated significant associations with the uptake of HPWPs in medium-sized firms, suggesting that the extent of the uptake of HPWPs in medium-sized firms is more likely to be influenced by external factors than internal ones and thus medium-sized firms tended to adopt HPWPs mainly in response to external pressures and advice.

#### Large firms

Similar to the findings in small firms, there were both external and internal factors that were associated with the uptake of HPWPs in large firms, though some of these factors are different from those in small firms.

There was a relationship between IiP recognition and the uptake of HPWPs in large firms, suggesting that obtaining IiP accreditation encourages large firms to introduce more HPWPs, or alternatively, large firms that have a greater use of HPWPs are more likely to be accredited as IiP.

There was also a relationship between participating in two or more business networks and the uptake of HPWPs in large firms. Such a result appears to be fairly similar to the one reported in small firms, though the association between multi-membership and the uptake of HPWPs in small firms is only marginally significant. However, compared to small (4.1 per cent) and medium-sized firms (5 per cent), more large firms (13.6 per cent) tended to participate in more

than two of the different business networks asked about. Therefore, large firms seem to be more willing to adopt certain HPWPs after reflecting on the information/advice collected from multiple sources, probably because they can afford to act appropriately on the information/advice they received and have the managerial capability to do so.

Turning to internal factors, it is noteworthy that very large firms with 50,000 employees tended to adopt more HPWPs than other large firms. This finding lends support to the earlier conjecture developed in Chapter 3 that these very large firms require more HPWPs to coordinate a much more complex HRM system which possibly involves developing sophisticated HR policies to cope with managing larger business divisions and workplaces (Chang et al., 2009; Parry et al., 2008).

There is also a relationship between family ownership and HPWPs, suggesting that large firms that are not family-owned were more likely to adopt more HPWPs than large family-owned firms. Although family ownership is more a feature of the literature on small and medium-sized firms (Ram et al., 2007), the results suggest this affects HPWPs only in large firms. One possible interpretation is that non-family-owned large firms can route the available resources to various requirements such as strategic development in HR resources agreed at the board level, while in family-owned large firms family members tend to dominate decision-making particularly at senior levels and thus resources may be allocated to the fields/business units that best serve the interest of the owner(s) rather than issues that can strategically affect corporate development.

As expected, and similar to the finding in small firms, union recognition was associated with HPWPs in large firms, suggesting that unions exert pressure on large firms to introduce more HPWPs (Jackson et al., 1989).

In all, the results from the assessment of the factors associated with the use of HPWPs across different firm sectors again highlight that medium-sized firms are distinct from small and large firms. While the uptake of HPWPs in small and large firms appeared to be associated with both

external and internal factors, the use of HPWPs in medium-sized firms was mainly associated with external factors. The strong relationship between external factors and the use of HPWPs would imply that medium-sized firms are particularly sensitive to aspects of their external environment.

## 6.3 The difference in the HPWP-performance link between small and medium-sized firms

The debates on the HPWP-performance link reviewed in Chapter 2 suggest that, although individual HPWPs are able to improve performance (Pfeffer, 1994), a much stronger association should be expected between HPWP bundles and performance (MacDuffie, 1995), and an extensive use of these practices as a complex HRM system will generate even greater effects on firm performance (Bowen and Ostroff, 2004; Huselid, 1995).

The review of the debate on HRM in SMEs nonetheless indicates that it is unlikely for small or medium-sized firms to follow a systematic approach towards HPWPs (Sisson, 1993) and the use of specific HPWP(s) are generally not associated with high performance in either small (Storey, 2004) or medium-sized firms (Storey, 2002). However, other studies report that the use of HPWPs increased in small and medium-sized firms (Bacon et al., 1996; Forth et al., 2006) and some studies also report HPWPs are associated with some aspects of performance in small firms (Faem et al., 2005; Sels et al., 2006; Way, 2002). Hence, these studies suggest that the HPWP-performance link is indeed universal.

To test this, Chapter 5 explored the HPWP-performance link from a range of perspectives and the findings contradict some of the extant literature. These findings are discussed below.

### Small firms

In small firms, little evidence was found of an association between individual HPWPs and performance. Only 2 out of the 18 HPWPs were positively associated with performance: an

internal labour market was associated with higher labour productivity and induction was linked to better quality of product or service. This would seem to suggest that small firms rarely benefit from using a single HPWP. Such a finding contradicts de Grip and Sieben's (2005) study reporting that training improved labour productivity in small firms and lends support to other studies that report the use of specific HPWPs such as training do not lead to improved performance in small firms (Storey and Westhead, 1997; Storey, 2004). Hence, it would appear that investing in HPWPs could be costly in small firms compared to large firms where economies of scale spread the cost across a large workforce.

However, findings in Chapter 5 also show that a specific bundle of HPWPs that seek to develop opportunities for employees to participate in work and develop their skills and abilities at work is associated with enhanced financial performance in small firms, suggesting that consistent HPWPs aiming to provide employees with the opportunities to perform are likely to provide small firms with the leverage to outperform other firms. Moreover, given that none of the HPWPs constituting the opportunity bundle are individually related to financial performance, the co-existence of these practices appears to create a reinforcing effect and eventually results in a significant improvement in economic returns. In addition, the test of 'internal fit' suggests that small firms also tend to benefit more from adopting different bundles of HPWPs (i.e. the AMOC framework) than using individual HPWPs. It appears synergy may occur between these bundles, though internal fit was not related to the distal performance measure of financial performance. A possible explanation could be that small firms are more likely to have specific HRM concerns and so using a set of closely defined HPWPs helps to focus on those concerns; however, the cost of investing in such HPWPs may outweigh the benefits these practices could generate and thus fail to result in improved financial performance in small firms (Way, 2002).

In contrast to Sels et al. (2006) who found that HPWP intensity leads to an overall increase in financial performance in small firms, the findings reported here show only a mild relationship between the overall use of HPWPs and financial performance in small firms. Although higher labour productivity has been reported in small firms with strong HRM systems, the relationship

between HPWP intensity and financial performance, as well as quality of product or service, is weak. This appears to suggest, although the use of HPWPs is associated with certain aspects of performance in small firms, the lack of economies of scale would make it costly for small firms to adopt such practices and thus reduce the chance of achieving higher economic returns.

### Medium-sized firms

Aside from a few HPWPs consistently associated with poor performance on several performance measures and a few other HPWPs that are simultaneously associated with both improved performance and poor performance on different measures, there are only two HPWPs that are associated only with improved performance in medium-sized firms. Hence, the relationship between individual HPWPs and performance is weak in medium-sized firms as well as in small firms.

Similar to the findings in small firms, where HPWP bundles are concerned, reinforcing effects between closely-defined HPWPs are also reported in medium-sized firms. Most of the associations however suggest HPWP bundles would result in poor performance. Apart from the lower absence rate reported in medium-sized firms using the motivation bundle, all the other associations nevertheless indicate that using a set of HPWPs seeking to motivate the workforce by appraising performance, paying them according to appraisal outcomes and offering opportunities for promotions, appear to result in lower labour productivity. One possible interpretation could be that these practices do not overcome HRM problems in medium-sized firms, probably problems in co-ordination and control (Baron et al., 2007; Greiner, 1972), and therefore may not increase employee motivation.

In stark contrast to the finding in small and large firms, there was no relationship between HPWP intensity and performance in medium-sized firms. Such a finding is not beyond expectation as the negative associations established by a few HPWPs may well cancel out the positive effect generated by a few other HPWPs in medium-sized firms. There are two possible explanations. One is that medium-sized firms are less likely to have economies of scale compared to large firms and this makes it costly for medium-sized firms to use HPWPs. The

larger workforce in medium-sized firms compared to small firms nonetheless suggests that more HPWPs are required to deal with HRM problems such as co-ordination and control and thereby prompt such firms to move toward a bureaucratic logic (Baron et al., 2007:60). Hence, medium-sized firms would appear to be stuck in the middle: more HPWPs are required compared to small firms but they involve high costs compared to large firms, and thereby no significant improvement in performance appears as a result in medium-sized firms.

That said, it could also be the case that medium-sized firms may adopt HPWPs under external pressure and the requirement to demonstrate professional management practices in order to attract additional funding from external financers. Adopting HPWPs mainly as a result of external pressure may distract managers from addressing internal HR problems and lead to no significant impact on performance in medium-sized firms.

#### Large firms

Although more HPWPs (6 out of 18) were positively associated with different performance measures in large firms, the incidence of the individual HPWPs associated with improved performance remains low, suggesting that the HPWP-performance link in large firms is not universal where individual HPWPs are considered in isolation.

Evidence suggests that large firms barely benefit from the use of the various HPWP bundles either, indicating that adopting closely combined HPWPs to target specific HRM concerns does not contribute to performance in large firms.

However, significantly higher levels of performance were found in large firms with a strong HRM system. This clearly demonstrates that a strong HRM system is more likely to have a greater impact on performance (Bowen and Onsoff, 2004) in large firms than either individual HPWPs used in isolation or in various combinations. Hence, the results seem to suggest that, compared to small firms, large firms tend to adopt different HPWPs across various HRM issues in order to build a sophisticated HRM system to facilitate the operation of an

organisation with a much more complex structure compared to small and medium-sized firms.

It is therefore the overall HRM system that matters in large firms.

To summarize, the findings in relation to the assessment of the HPWP-performance link suggest the overall use of HPWPs is strongly associated with performance in large firms; in small firms, a specific bundle of HPWPs that seeks to enhance the opportunity for employees to participate in management decision-making is associated with improved financial performance. In contrast, neither the overall use of HPWPs nor any specific bundle of HPWPs within the AMOC framework is associated with performance in medium-sized firms. In medium-sized firms, some HPWPs are associated with better performance, some are associated with poor performance and a few others are simultaneously associated with both improved and poor performance.

Overall, the findings in relation to the HPWP-performance link in small, medium and large firms suggest that using HPWPs, no matter how these practices are combined, does not necessarily lead to higher performance. Hence the HPWP-performance link is not universal.

The results in this chapter also suggest that studies of the HPWP-performance link in small and medium-sized firms should assess the use of HPWPs from a range of perspectives (i.e. individual HPWPs, several closely defined HPWPs used in combination/HPWP bundles and HPWP intensity), given that the manner of using such practices varies greatly between these firms as demonstrated in Chapter 3 and thus the pattern of such links could well differ between these firms.

Inevitably, the analysis in this thesis is subject to the caveats common to cross-sectional analyses of this nature, and the results here cannot be interpreted as causal. By way of example, the findings in Chapter 3 that financial performance is higher in situations where small firms have used the opportunity bundle does not establish that the opportunity bundle itself has caused high performance, or whether small firms with higher financial performance have taken

the opportunity to invest in this specific HPWP bundle. To ascertain causality, longitudinal data is ideally required.

Interpreting the results should also take into consideration the potential for common-method variance, as the same respondent provided data for both the dependent and the independent variables. In the case of the HPWP-performance link debate, common methods bias is likely to be associated with the phenomenon that respondents who claim to have a greater use of HPWPs are likely to rate performance more highly. However, HPWP bundles are largely associated with poor performance in medium-sized firms and the greater use of HPWPs (i.e. HPWP intensity) is not associated with improved performance. This could indicate that the positive associations between HPWP intensity and higher performance (i.e. higher labour productivity, better quality of product or service and enhanced financial performance) in small firms may be more the result of genuine performance effects rather than common method variance.

# 6.4 Analytically splitting small and medium-sized firms as two distinct groups

The findings in this thesis suggest that the nature of HRM in small firms is different from that in medium-sized firms, and the conventional approach of combining these firms into a single unit of analysis obscures important differences in the relationship between HPWPs and performance in small and medium-sized firms. For example, no evidence was found to suggest that off-the-job training is linked to better performance in small firms (see Chapter 5) in line with Storey (2004); encouraging small firms to invest in more training as a way to improve their performance could therefore be misguided. Nevertheless, in medium-sized firms off-the-job training is significantly associated with lower labour turnover, high labour productivity and better quality of product or service (see Chapter 5). Such contrasting results are instructive, as dividing small and medium-sized firms into two distinct groups provides a better

understanding of the impact of training on firm performance – it matters in medium-sized firms but not in small firms.

Analytically splitting small and medium-sized firms into two distinct groups also provides a good opportunity to appreciate why the HPWP-performance link varies between small, medium and large firms. The evidence presented here suggests that medium-sized firms are likely to have particular characteristics that differentiate them from both small and large firms, and additionally, a distinct pattern of HPWP-performance link in medium-sized firms has been identified. Looking at this link in medium-sized firms specifically, this thesis has demonstrated that some HPWPs were consistently and positively associated with performance, some were consistently but negatively associated with performance and some others were simultaneously associated with both better performance and poor performance. However, there is no similar evidence of such variation in the impact of different HPWPs in either small or large firms. It is clear, therefore, that analysing small and medium-sized firms separately will help in the systematic study of the relationship between HPWPs and performance in small, medium and large firms independently. Analysing small and medium-sized firms separately also provides insights into the different factors associated with the uptake of HPWPs.

The finding that there is a distinct relationship between HPWPs and performance in medium-sized firms compared to small and large firms suggests this issue would certainly prove a fruitful avenue for future research. In particular, it may be worth exploring in greater depth those HPWPs negatively associated with performance in medium-sized firms. Further analysis using WERS 2004 or case studies that focus on medium-sized firms in the future could generate new findings with regard to the nature of HRM in medium-sized firms. Such analyses, therefore, would assist researchers in both academic and government policy circles in understanding not only the differential impact of HPWPs on performance that appears to occur between small and medium-sized firms, but also to understand why using specific HPWPs could harm performance in medium-sized firms and thereby enabling analysis and policy in the future to be tailored more appropriately to the specific sector.

The findings in this thesis have relevance beyond Britain. Looking at the official websites where different countries' governments publish their national business strategies and policies, for example, Japan, Australia, the United States, Canada, China and the EU, small and medium-sized firms are routinely categorized into one single sector and are unanimously targeted by the same policy or government-sponsored initiatives. As stated in this thesis, such a combination may confound government attempts to improve performance in the SME sector. The results presented in this thesis suggest that any country that has policies specifically targeting the SME sector should be somewhat cautious with regard to the difference in the potential impact of practices such as HPWPs on performance between small and medium-sized firms.

### Bibliography

- ACAS (2001) ACAS Business Plan 2001-2002. London: ACAS.
- ACAS (2010) ACAS Annual Report 2009-2010. London: ACAS.
- Agarwal, R. and Audretsch, D.B. (1999) The two views of small firms in industry dynamics: a reconciliation. *Economics Letters*, 62(2): 245-51.
- Appelbaum, E., Bailey, T., Berg, P. and Kallberg, A. (2000) *Manufacturing Advantage*. Ithaca, N.Y.: ILR Press.
- Argón-Sánchez, A., Barba-Aragón, I. And Sanz-Valle, R. (2003) Effects of training on business results. *International Journal of Human Resource Management*, 14(6): 956-80.
- Arthur, J.B. (1992) The link between business strategy and industrial relations systems in American steel mini-mills. *Industrial and Labour Relations Review*, 45: 488-506.
- Arthur, J.B. (1994) Effects of human resource systems on manufacturing performance and turnover. *Academy of Management Journal*, 37: 670-87.
- Arthur, J.B. (1999) Explaining variation in human resource practices in U.S. steel minimills. In
   P. Cappelli, ed. Employment practices and business strategy, 11-41, Oxford: Oxford
   University Press.
- Bacon, N., Ackers, P., Storey, J. and Coates, D. (1996) It's a Small World: Managing human Resources in Small Business. *International Journal of Human Resource Management*, 7(1): 82–100.
- Bacon, N. and Hoque, K. (2005) HRM in the SME Sector: Valuable Employees and Coercive Networks. *International Journal of Human Resource Management*, 16(11): 1976-99.
- Banker, R.D., Field, J.M., Schroeder, R.G. and Sinha, K.K. (1996) Impact of work teams on manufacturing performance: A longitudinal field study. *Academy of Management Journal*, 39: 867-90.
- Baron, R. A. (2003) Human resource management and entrepreneurship: Some reciprocal benefits of closer links. *Human Resource Management Review*, 13: 253-56.
- Baron, J. N. and Hannan, M. T. (2002) Organizational blueprints for success in high-tech startups: Lessons from the Stanford project on emerging companies. *California Management Review*, 44(3): 8–36.

- Baron, J.N., Hannan, M.T., Hsu, G. and Koçcak, Ö. (2007) In the company of women: gender inequality and the logic of bureaucracy in start-up firms. *Work and Occupations*, 34(1): 35-66.
- Batstone, E. and Gourtay, S. (1986) *Unions, Unemployment and Innovation*. Oxford: Basil Biackwell.
- Becker, B. and Gerhart, B. (1996) The impact of human reduce management on organizational performance: progress and prospects. *Academy of Management Journal*, 39(4): 779-801.
- Becker, B. E., and Huselid, M. A. (1998) High performance work systems and firm performance: A synthesis of research and managerial implications. In G. R. Ferris, ed. Research in personnel and human resource management, (16): 53-101. Stamford, CT: JAI Press.
- Becker, B. and Huselid, M. (2006) Strategic Human Resources Management: Where Do We Go From Here? *Journal of Management*, 32(6): 898-925.
- Becker, G.S. (1964) Human Capital, 2<sup>nd</sup> ed. New York: Columbia University Press.
- Beer, M., Spector, B., Lawrence, P., Quinn Mills, D. and Walton, R. (1985) *Human Resource Management: A General Manager's Perspective*. Glencoe, I11: Free Press.
- Bennett, R.J. and. Robson, P.A. (2003) Changing Use of External Business Advice and Government Supports by SMEs in the 1990s. *Regional Studies*, 37(8): 795–811.
- Benson, G.S., Young, S.M. and Lawler, E.E. (2006) High-involvement work practices and analysts forecasts of corporate earnings. *Human Resource Management*, 45(4): 519-37.
- Birdi, K., Clegg, C., Patterson, M., Robinson, A., Stride, C.B., Wall, T.D. and Wood, S. (2008)

  The impact of human resource and operational management practices on company productivity: a longitudinal study. *Personnel Psychology*, 61: 467-501.
- BIS (2010) Financing business growth: The Government's response to *Financing a private* sector recovery. [Online] London: BIS (Published in Oct 2010) Available at http://www.bis.gov.uk/assets/biscore/corporate/docs/f/10-1242-financing-business-growth-response.pdf [accessed 10 November 2010].
- BIS (2009) SME Statistics for the UK and regions 2008. [Online] London: BIS (Published 14 Oct 2009) Available at http://stats.bis.gov.uk/ed/sme [accessed 17 January 2010].

- BIS (2008) Enterprise: unlocking the UK's talent. [Online] London: BIS (Published March 2008). Available at http://www.bis.gov.uk/files/file44992.pdf [accessed 27 October 2010].
- Blau, P.M (1970) A formal theory of differentiation in organizations. *American Sociological Review*, 35(2): 201-18.
- Blyton, P., and Turnbull, P. (1998) The Dynamics of Employee Relations (2<sup>nd</sup> edn). London: Macmillan.
- Bohlander, G., Snell, S and Sherman, A. (2001) *Managing Human Resources* (12<sup>th</sup> edn). Cincinnati, OH: South-Western College Publishing.
- Bowen, D.E. (1986) Managing customers as human resources in service organizations. *Human Resource Management*, 25: 372-84.
- Bowen, D. and Ostroff, C. (2004) Understanding HRM-firm performance linkages: The role of 'strength' of the HRM system. *Academy of Management Review*, 29: 203-21.
- Bowles, M.L. and Coates, G. (1993) Image and substance: the management of performance as rhetoric or reality. *Personnel Review*, 22(2): 3-21.
- Boxall, P. and Purcell, J. (2000) Strategic human resource management: where have we come from and where should we be going? *International Journal of Management Reviews*, 2(2): 183-203.
- Boxall, P and Purcell, J. (2003) Strategy and human resource management (1<sup>st</sup> edn).

  Hampshire, England: Palgrave-Macmillan.
- Boxall, P and Purcell, J. (2008) Strategy and human resource management (2<sup>nd</sup> edn).

  Hampshire, England: Palgrave-Macmillan.
- Boyett, J.H., and Conn, H.P. (1991) Workplace 2000: The revolution reshaping American business. New York: Penguin.
- Broschak, J.P., Davis-Blake, A. and Block, E.S. (2008) Nonstandard, not substandard: the relationship among work arrangements, work attitudes, and job performance. *Work and Occupations*, 35(1): 3-43.
- Brown, M. and Heywood, J.S. (2005) Performance appraisal systems: determinants and change. British Journal of Industrial Relations, 43(4): 659-79.

- Bryson, A., Forth, J. and Kirby, S. (2005) high-involvement management practices, trade union representation and workplace performance in Britain. *Scottish Journal of Political Economy*, 52(3): 451-92.
- Campbell-Hunt, C. (2000) What have we learned about generic competitive strategy? A metaanalysis. *Strategic Management Journal*, 21(2): 127-54.
- Cappelli, P. (1999) Employment practices and business strategy. Oxford: Oxford University Press.
- Cardon, M.S. and Stevens, C.S. (2004) Managing human resources in small organizations: What do we know? *Human Resource Management Review*, 14: 295–323.
- Cardy, R. L. and Dobbins, G. H. (1994) *Performance Appraisal: Alternative Perspectives*.

  Cincinnati, OH: South Western Publishing Company.
- Carless, S. A. (2005) Person-job fit versus person-organization fit as predictors of organizational attraction and job acceptance intentions: A longitudinal study. *Journal of Occupational and Organizational Psychology*, 78: 411-29.
- Carroll, M., Marchington, M., and Earnshaw, J. (1999) Recruitment in small firms: processes methods and problems, *Industrial Relations*, 21(3): 236-50.
- Carter, N.M., Stearns, T.M., Reynolds, P.D. and Miller, B. (1994) New venture strategies: theory development with an empirical base. *Strategic Management Journal*, 15: 21-41.
- Clark, K.B. (1984) Unionisation and firm performance: the impact on profits, growth and productivity. *American Economic Review*, 74: 893-919.
- Cassell, C., Nadin, S., Gray, M. and Clegg, C. (2002) Exploring human resource management practices in small and medium sized enterprises. *Personnel Review*, 31(5/6): 671-93.
- Chang, Y.Y., Mellahi, K. and Wilkinson, A. (2009) Control of subsidiaries of MNCs from emerging economies in developed countries: the case of Taiwanese MNCs in the UK.

  The International Journal of Human Resource Management, 20(1): 75-95.
- Cohen, A. (2007) Commitment before and after: An evaluation and reconceptualization of organizational commitment. *Human Resource Management Review*, 17: 336-54.
- Combs, J., Liu, Y., Hall, A. and Ketchen, D. (2006) How much do high-performance work practices matter? A meta-analysis of their effects on organizational performance. Personnel Psychology, 59: 501-28.

- Cooke, W. (1994) Employee participation programs, group-based incentives and company performance. *Industrial and Labor Relations Review*, 47: 594-609.
- Coyle-Shapiro, J.A-M., Morrow, P.C., Richardson, R., and Dunn, S.R. (2002) Using profit sharing to enhance employee attitudes: a longitudinal examination of the effects on trust and commitment. *Human Resource Management*, 41(4): 423-39.
- Cully, M. (1998) A Survey in Transition: The Design of the 1998 Workplace Employee Relations Survey, London: Department of Trade and Industry.
- Curran, J. (1996) Small Business Strategy in Warner. M (ed) *International Encyclopedia of Business and Management*, London and NY: International Thompson Press, Routledge.
- Datta, D.K., Guthrie, J.P. and Wright, P.M. (2005) Human resource management and labor productivity: does industry matter? *Academy of Management Journal*, 48(1): 135-45.
- Davis, A.E. and Kalleberg, A.L. (2006) Family-friendly organizations? Work and family programs in the 1990s. *Work and Occupations*, 33: 191-223.
- de Grip, A. and Sieben, I. (2005) The effects of human resource management on small firms' productivity and employees' wages. *Applied Economics*, 37: 1047-54.
- de Kok, J. and Uhlaner, L.M. (2001) Organization context and human resource management in the small firm. *Small Business Economics*, 17(4): 273-91.
- Delaney, J.T. and Huselid, M.A. (1996) The impact of human resource management practices on perceptions of organizational performance. *Academy of Management Journal*, 39(4): 949-69.
- Delaney, J.T. and Huselid, M.A. (1996) The impact of human resource management practices on perceptions of organizational performance. *Academy of Management Journal*, 39(4): 949-69.
- den Hartog, D.N., Boselie, P. and Paauwe, J. (2004) Performance management: a model and research agenda. *Applied psychology: An International Review*, 53(4): 556-69.
- Delery J.E. (1998) Issues of fit in strategic human resource management: Implications for research. Human Resource Management Review, 8: 289-309.
- Delery, J. and Doty, H. (1996) Modes of theorizing in strategic human resource management: tests of universalistic, contingency and configurational performance predictions.

  \*\*Academy of Management Journal\*, 39: 802-35.

- Delery, J., Gupta, N., Shaw, J., Jenkins, G., and Ganster, M. (2000) Unionization, compensation, and voice effects on quits and retention. *Industrial Relations*, 39(4): 625-45.
- Dickens, L. (1986) Equal Opportunity—A More Encouraging Approach?. Employee Relations, 8(4): 27-32.
- Drummond, I. and Stone, I. (2006) Exploring the potential of high performance work systems in SMEs. *Employee Relations*, 29(2): 192-207.
- Department of Trade and Industry. (2005) Workplace Employment Relations Survey: Cross-section, 2004 [computer file]. Colchester: The Data Archive [distributor], SN: 5294.
- Duberley, J.P. and Walley, P. (1995) Assessing the adoption of HRM by small and medium-sized manufacturing organizations. *The International Journal of Human Resource Management*, 4(4): 891-909.
- Dundon, T., Grugulis, I., and Wilkinson, A. (2001) New management techniques in small and medium-sized enterprises. In Redman, T., and A. Wilkinson, ed. *Contemporary Human Resource Management: text and cases*, 432-63, Financial Times: Prentice Hall.
- Dyer, L. and Reeves, T. (1995) Human resource strategies and firm performance: What do we know and where do we need to go? *The International Journal of Human Resource Management*, 6: 656-70.
- Edelman, L.F., Brush, C.G. and Manolova, T. (2005) Co-alignment in the resource-performance relationship: strategy as mediator. *Journal of Business Venturing*, 20: 359-83.
- Edwards, P. (1987) Managing the Factory. Oxford: Basil Blackwell.
- Enterprise Directorate (2006) SME definitions. [Online] Available at http://www.dti.gov.uk/bbf/financial-reporting/acc-audit-developments/page16361.html [accessed 17 January, 2008].
- Erickson, C.L. and Jacoby, S.M. (2003) The Effect of Employer Networks on Workplace Innovation and Training. *Industrial and Labor Relations Review*, 56(2): 203-24.
- Ewing, K. and Hock, A. (2003) *The Next Step: Trade Union Recognition in Small Enterprises*.

  London: TUC Publications.

- Faems, D., Sels, L., de Winne, S. and Maes, J. (2005) The effect of individual HR domains on financial performance: evidence from Belgian small businesses. *International Journal of Human Resource Management*, 16(5): 676-700.
- Fernie, S. and Metcalf, D. (1995) Participation, contingent pay, representation and workplace performance: evidence from Great Britain. *British Journal of Industrial Relations*, 33(3): 379-415.
- Findlay, P., McKinlay, A., Marks, A. and Thompson, P. (2000) Flexible when it suits them: the use and abuse of teamwork skills. In Procter, S. and F.Mueller, ed. *Teamworking*, 222-43, London: Macmillan.
- Fletcher, C. (2001) Performance appraisal and management: The developing research agenda. *Journal of Occupational and Organizational Psychology*, 73:473–87.
- Forde, C., Slater, G. & Spencer, D.A. (2006) It's the taking part that counts? Participation, performance and external labour market conditions. *Industrial Relations*, 61(2): 296-320.
- Forth, J., Bewley, H. and Bryson, A. (2006) Small and medium-sized enterprises: findings from the 2004 workplace employment relations survey. London: Routledge.
- Forth, J. and McNabb, R. (2008) Workplace performance: a comparison of subjective and objective measures in the 2004 Workplace Employment Relations Survey. *Industrial Relations Journal*, 39(2): 104-23.
- Frormann, D. (2006) Medium-sized business in Germany. International Journal of Entrepreneurship and Innovation Management, 6(1/2): 18-23.
- Gerhart, B. (2005) Human resources and business performance: findings, unanswered questions, and an alternative approach. *Management Revue*, 16(2): 174-85.
- Gerhart, B., Wright, P.M., McMahan, G.C. and Snell, S.A. (2000) Measurement error in research on human resources and firm performance: how much error and does it influence effect size estimates. *Personnel Psychology*, 53: 803-34.
- Gibb, A.A. (2000) SME policy, academic research and the growth of ignorance, mythical concepts, myths, assumptions, rituals and confusions. *International Small Business Journal*, 18: 13-35.

- Glass, J. and Fujimoto, T. (1995) Employer characteristics and the provision of family responsive policies. *Work and Occupations*, 22: 380-411.
- Golhar, D.Y. and Deshpade, S.P. (1997) HRM practices of large and small Canadian manufacturing firms. *Journal of Small Business Management*, 35(3): 30-8.
- Gollan, P.J. and Wildinson, A. (2007) Contemprorary developments in information and consultation. The International Journal of Human Resource Management, 18(7): 1133-44.
- Gomez-Mejia, L.R., Balkin, D.B. and Cardy, R.L. (2001) *Managing Human Resources* (3<sup>rd</sup> edn). Upper Saddle River, NJ: Prentice Hall.
- Goss, D., Adam-Smith, D. and Gilbert, A. (1994) Small firms and HRM: exceptions that prove the rule? *Journal of Small Business and Enterprise Development*, 1(2): 2-8.
- Green, F. (1993) The impact of trade union membership on training in Britain. *Applied Economics*, 25: 1033-43.
- Greiner, L.E. (1972) Evolution and revolution as organisations grow. *Harvard Business Review*, 50(4): 37-46.
- Griffin, R. (1988) Consequences of quality circles in an industrial setting. Academy of Management Journal, 31: 338-58.
- Guest, D. (1987) Human resource management and industrial relations. *Journal of Management Studies*, 24(5): 503-21.
- Guest, D.E. (1997) Human resource management and performance: a review and research agenda. The International Journal of Human Resource Management, 8(3): 263-76.
- Guest, D.E. (2001) Human resource management: when research confronts theory.

  International Journal of Human Resource Management, 12(7): 1092-106.
- Guest, D.E. and Conway, N. (2007) Human Resource Management, employee attitudes and workplace performance: an examination of the linkages using the 2004 workplace Employment Relations Survey. A report for the Department for Business, Enterprise and Regulatory Reform.
- Guest, D.E., Michie, J., Conway, N. and Sheehan, M. (2003) Human resource management and corporat performance in the UK. *British Journal of Industrial Relations*, 41(2): 291-314.

- Guthrie, J.P. (2001) High-involvement work practices, turnover, and productivity: Evidence from Newzealand. *Academy of Management Journal*, 44(1): 180-90.
- Hair, J.F., Black, W.C., Babin, B.J., nderson, R.E. and Ttham, R.L. (2006) *Multivariate Data Analysis* (6<sup>th</sup> edn). New Jersey: Pearson Prentice Hall.
- Harley, B. (2001) Team membership and the experience of work in Britain: an analysis of the WERS98 data. Work, Employment & Society, 15(4): 721-41.
- Harris, L.C., and Ogbonna, E. (2001) Strategic human resource management, market orientation, and organizational performance. *Journal of Business Research*, 51: 157-66.
- Hattie, J. (1985) Methodology review: Assessing unidimensionality of tests and items. *Applied Psychological Measurement*, 9: 139-64.
- Hendry, C., Arthur, M.B. and Jounes, A.M. (1995) Strategy through people: adaptation and learning in the small-medium enterprise. London: Routledge.
- Heneman, R.L. (1992) Merit pay: linking pay increases to performance ratings. Reading, MA:

  Addison Wesley Publishing Company.
- Heneman, H.G. and Berkley, R.A. (1999) Applicant attraction practices and outcomes among small businesses. *Journal of Small Business Management*, 37(1): 53-74.
- Heneman, R., Tansky, J., and Camp, S. (2000, Fall) Human resource management practices in small and medium sized enterprises: unanswered questions and future research perspectives. *Entrepreneurship, Theory and Practice*, 11-26.
- Hill, R. and Stewart, J. (1999) Human Resource Development in Small Organizations. *Human Resource Development International*, 2(2): 103-23.
- Hill, S. (1991) Why quality circles failed but total quality management might succeed. British Journal of Industrial Relations, 29(4): 541-68.
- Hitt, M.A., Bierman, L., Shimizu, K. and Kochhar, R. (2001) Direct and moderating effects of human capital on strategy and performance in professional service firms: a resource-based perspective. *Academy of Management Journal*, 44(1): 13-28.
- Holliday, R. (1995) Investigating Small Firms: Nice Work? London: Routledge.
- Hoque, K. (2000) Human Resource Management in the Hotel Industry: Strategy, innovation and performance. London: Routledge.

- Hoque, K. and Bacon, N. (2006) The antecedents of training activity in British small and medium-sized enterprises. Work, Employment and Society, 20(3): 531-52.
- Hoque, K. and Bacon, N. (2008) Investors in People and training in the British SME sector.

  Human Relations, 61(4): 451-82.
- Hoque, K. and Kirkpatrick, I. (2003) Non-standard employment in the management and professional workforce: training, consultation and gender implications. *Work Employment & Society*, 17: 667-89.
- Hoque, K. and Noon, M. (2001) Counting angels: a comparison of personnel and HR specialists. *Human Resource Management Journal*, 11(3): 5-22.
- Hunter, L., Beaumont, P. and Sinclair, D. (1996) A "Partnership" Route to Human Resource Management?. *Journal of Management Studies*, 33(2): 235-57.
- Huselid, M. A. (1995) The impact of human resource management practices on turnover, productivity, and corporate financial performance. *Academy of Management Journal*, 38(3): 635-72.
- Ichniowski, C., Kochan, T., Levin, D., Olson, C. and Strauss, G. (1996) What works at work: overview and assessment. *Industrial Relations*, 35(3): 299-333.
- Ichniowski, C., Kochan, T.A., Levine, D.I., Olson, C. and Strauss, G. (2000) What works at work: overview and assessment. In C. Ichiniowski, D.I. Levine, C. Olson, and G. Strauss ed. *The American Workplace: Skills, Compensation and Employee Involvement*, 1-29, Cambridge: Cambridge University Press.
- Ichniowski, C., Shaw, K. and Prennushi, G. (1997) The effects of human resource management practices on productivity: a study of steel finishing lines. *American Economic Review*, 87: 291-313.
- Jackson, S.E., Schuler, R.S. and Rivero, J.C. (1989) Organizational characteristics as predictors of personnel policies. *Personnel Psychology* (Winter), 727-86.
- Jime'nez-Jime'nez, D. and Sanz-Valle, R. (2005) Innovation and human resource management fit: an empirical study. *International Journal of Manpower*, 26(4): 364-81.
- Julien, P.A. (1995) New technologies and technological information in small business. *Journal of Business Venturing*, 10(6): 459-75.

- Kallegerg, A.L. (2003) Flexible firms and labor market segmentation: effects of workplace restructuring on job and workers. *Work and Occupations*, 30(2): 154-75.
- Kalleberg, A.L., Reskin, B.F., and Hudson, K. (2000) Bad jobs in America: standard and nonstandard employment relations and job quality in the United States. *American Sociological Review*, 65: 256-78.
- Kandola, R. and Fullerton, J. (1994) Managing the Mosaic, London: CIPD.
- Keizer, A.B. (2009) Transformations in- and outside the internal labour market: institutional change and continuity in Japanese employment practices. *International Journal of Human Resource Management*, 20(7): 1521-35.
- Kersley, B. and Martin, C. (1997) Productivity growth, participation and communication. Scottish Journal of Political Economy, 44(5): 485-501.
- Kim, Jae-On and Charles W. Mueller (1978) Factor Analysis: Statistical methods and practical issues. Thousand Oaks, CA: Sage Publications, Quantitative Applications in the Social Sciences Series, No. 14.
- Kinnie, N., Purcell, J., Hutchinson, S., Terry, M., Collinson, M. and Scarbrough, H. (1999)

  Employemnt reltions in SMEs: maket-driven or customer-shaped? *Employee Relations*, 21(3): 218-35.
- Kitson, M. and Wilkinson, F. (2003) Labour mobility, training and labour force flexibility. in A. Cosh and A. Hughes, ed. *Enterprise Challenged: Policy and Performance in the British SME Sector 1999–2002*, Cambridge: Centre for Business Research.
- Knocke, D. and Kalleberg, A.L. (1994) Job Training in U.S. Organisations. *American Sociological Review*, 59: 537-46.
- Kotey, B., and Slade, P. (2005) Formal human resource management practices in small growing firms. *Journal of Small Business Management*, 43(1): 16-24.
- Lam, L.W., and White, L.P. (1998) Human resource orientation and corporate performance.

  Human Resource Development Quarterly, 9(4): 351-64.
- Lawler, E. (1971) Pay and organizational effectiveness. New York: McGraw-Hill.
- Lawler, E. (1992) The Ultimate Advantage: Creating the High-Involvement Organization. San Francisco: Jossey-Bass.

- Lawler, E. E. III, Mohrman, S. A., and Ledford, G. E. Jr. (1998) Strategies for high performance organizations: The CEO report. San Francisco: Jossey-Bass Publishers.
- Lazear, E. (1998) The Economics of Personnel. New York: John Wiley and Sons.
- Legge, K. (1995) Human Resource Management: Rhetorics and Realities, London: Macmillan.
- Lepak, D. P., and Snell, S. A. (2002) Examining the human resources architecture: The relationships among human capital, employment and resource configurations. *Journal of Management*, 28(4): 517-43.
- Leung, A. (2003) Different ties for different needs: recruitment practices of entrepreneurial firms at different developmental phases. *Human Resource Management*, 42(4): 303-20.
- Levne, D., and Tyson, L.D. (1990) Participation, productivity, and the firm's environment. In A.S. Blinder, ed. *Paying for Productivity*, 183-244. Washington, DC: Brookings Institution.
- Litz, R.A. and Stewart, A.C. (2000) Research Note: Trade Name Franchise Membership as a Human Resource Management Strategy: Does Buying Group Training Deliver "True Value" for Small Retailers? *Entrepreneurship Theory and Practice*, 25: 125-35.
- Lynch, L.M. and Black, S.E. (1998) Beyond the incidence of employer-provided training.

  Industrial and Labor Relations Review, 52: 64-81.
- MacDuffie, J. P. (1995) Human resource bundles and manufacturing performance: organizational logic and flexible production systems in the world auto industry.

  Industrial and Labor Relations Review, 48: 197-221.
- Machin, S. (2004) Factors of convergence and divergence in union membership. British Journal of Industrial Relations, 42(3): 423-38.
- Machin, S. and Stewart, M. (1990) Unions and the financial performance of British private sector establishments. *Journal of Applied Econometrics*, 5(4): 327-50.
- Machin, S. and Stewart, M. (1996) Trade unions and financial performance. Oxford Economic Papers, 48(2): 213-41.
- MacMahon, J. (1996) Employee relations in small firms in Ireland: an exploratory study of small manufacturing firms. *Employee Relations*, 18(5): 66-80.
- Marchington, M. (2000) Teamworking and employee involvement: terminology, evaluation and context. In Procter, S. and F.Mueller, ed. *Teamworking*, London:Macmillan.

- Marchington, M., and Grugulis, I. (2000) 'Best practice' human resource management: perfect opportunity or dangerous illusion? *International Journal of Human Resource Management*, 11(6): 1104-24.
- Marks, M., Mirvis, P., Hackett, E. and Grady, J. (1986). Employee participation in a quality circle program: impact on quality of work life, productivity, and absenteeism. *Journal of Applied Psychology*, 71: 61-9.
- Marlow, S. (2000) Investigating the use of emergent strategic human resource management activity in the small firms. *Journal of Small Business and Enterprise Development*, 7(2): 135-48.
- Marlow, S. (2006) Human resource management in smaller firms: A contradiction in terms? Human Resource Management Review, 16: 467-77.
- Marsden, D. and Richardson, R. (1994) Performing for pay? The effects of 'merit pay' on motivation in a public service. *British Journal of Industrial Relations*, 32(2): 243-61.
- Maume, D. (1999) Glass ceilings and glass escalators: occupational segregation and race and sex differences in managerial promotions. *Work and Occupations*, 26: 483-509.
- Mayson, S., and Barrett, R. (2006) The 'science' and 'practice' of HRM in small firms. *Human Resource Management Review*, 16: 447-55.
- Meyer, J.P. and Allen, N.J. (1997) Commitment in the Workplace: theory, research, and application. London: Sage.
- Michie, J., Zubanov, N. and Sheehan, M. (2008) Human resource management practices, organizational outcomes and performance: an analysis of WERS 2004 data. Employment relations occasional papers. Department for Business Enterprise & Regulatory Reform.
- Michie, J. and Sheehan, M. (2006) Business strategy and corporate performance in small firms.

  Unpublished research paper funded by the ESRC as part of their 'Human resource management and performance in small & medium sized enterprises' project (Grant RES-000-22-1142).
- Michie, J., and Sheehan, M. (2005) Business strategy, human resources, labour market flexibility and competitive Advantage. *International Journal of Human Resource Management*, 16(3): 445-64.

- Michie, J. & Sheehan, M. (2001) Labour market flexibility, human resource management and corporate performance. *British Journal of Management*, 12(4): 287-306.
- Miles, R.E., and Snow, C.C. (1978) Organizational Strategy, Structure and Process. New York: McGraw Hill.
- Milkovich, G.T., Newman, J.M. and Milkovich, C. (2005) *Compensation* (8th edn). New York: McGraw-Hill/Irwin.
- Mirvis, P.H. and Lawler, E.E. (1977) Measuring the financial impact of employee attitudes. *Journal of Applied Psychology*, 62(1): 1-8.
- Mirvis, P.H., Sales, A.L. and Hackett, E.J. (1991) The Implementation and Adoption of New Technology in Organizations: The Impact on work, people, and culture. *Human Resource Management*, 30(1): 113-39.
- Montemayor, E.F. (1996) Congruence between pay policy and competitive strategy in high-performing firms. *Journal of Management*, 22(6): 889-908.
- Nooteboom, B. (1993) Firm size effects on transaction costs. Small Business Economics, 5: 283-95.
- Nunnally, J. (1967) Psychometric Theory. New York: McGraw-Hill.
- OECD (2005) SME and Entrepreneurship Outlook 2005. Organisation for Economic Cooperation and Development, Paris.
- Parry, P., Dickmann, M. and Morley, M. (2008) North American MNCs and their HR policies in liberal and coordinated market economies. *The International Journal of Human Resource Management*, 19(11): 2024-40.
- Patton, D., Marlow, S. and Hannon, P. (2000) The relationship between training and small firm performance; research frameworks and lost quests. *International Small Business Journal*, 19(1): 11-27.
- Pereira, G. and Osburn, H. (2007) Effects of participation in decision making on performance and employee attitudes: a quality circles meta-analysis. *Journal of Business & Psychology*, 22(2): 145-53.
- Piekkola, H. (2005) Performance-related pay and firm performance in Finland. *International Journal of Manpower*, 26(7/8): 619-35.

- Pfeffer, J. (1994) Competitive advantage through people: unleashing the power of the workforce. Boston: Harvard University Press.
- Pfeffer, J. (1998) The Human Equation: Building Profits by Putting People First. Boston, MA: Harvard Business School Press.
- Pil. F.K. and MacDuffie, J.P. (1999) Organizational and environmental factors influencing the use and diffusion of high-involvement work practices. In P. Cappelli, ed. *Employment practices and business strategy*, 81-106, Oxford: Oxford University Press.
- Porter, M.E. (1980) Competitive Strategy. Free Press, New York, NY.
- Porter, M. E. (1998) Competitive advantage: Creating and sustaining superior performance.

  New York: Free Press.
- Preuss, G.A. and Lautsch, B.A. (2002) The effect of formal versus informal job security on employee involvement programs. *Industrial Relations*, 57(3): 517-39.
- Proctor, S. and Burridge, M. (2008) Teamworking and performance: the extent and intensity of teamworking in the 1998 UK workplace employee relations survey (WERS98). *The International Journal of Human Resource Management*, 19(1): 153-68.
- Purcell, J. and Rinnie, N. (2007) HRM and business performance. In J. Boxall and J. Purcell, ed. *Handbook of human resource management*, 533-49, Oxford: OUP.
- Rainnie, A. (1989) Industrial relations in small firms. London: Routledge.
- Ram, M. (2000) Investors in People in small firms: Case study evidence from the business services sector. *Personnel Review*, 29(1): 69-91.
- Ram, M. (1994) Managing to survive. London: Routledge.
- Ram, M. and Edwards, P. (2003) Praising Caesar not burying him: what we know about employment relations in small firms. Work, Employment & Society, 17(4): 719-30.
- Ram, M., Edwards, P. and Jones, T. (2007) Staying Underground Informal Work, Small Firms, and Employment Regulation in the United Kingdom. *Work and Occupations*, 34(3): 318-44.
- Ramsay, H., Scholarios, D. and Harley, B. (2000) Employees and high-performance work systems: testing inside the black box. *British Journal of Industrial relations*, 38(4): 501-31.

- Raymond, L. and St-Pierre, J. (2005) Antecedents and performance outcomes of advanced manufacturing systems sophistication in SMEs. *International Journal of Operations & Production Management*, 25(6): 514-33.
- Reid, G. (2007) The Foundations of Small Business Enterprise, London: Routledge.
- Renwick, D. and Gennard, J. (2001) Grievance procedures and discipline. In T. Redman and A. Wilkinson, ed. *Contemporary Human Resource Management: text and cases*, 168-92, Financial Times: Prentice Hall.
- Roberts, I., D. Sawbridge, and G. Bamber (1992) Employee Relations in Smaller Enterprises.

  In B. A. Towers, ed. *Handbook of Industrial Relations Practice*, London, UK: Kogan, 240–57.
- Rousseau, D.M. and Tijoriwala, S.A. (1998) Assessing psychological contracts: Issues, alternatives and measures. *Journal of Organizational Behavior*, 19: 679-95.
- Rutherford, M.W., Buller, P.F. and McMullen, P.R. (2003) Human resource management problems over the life cycle of small to medium-sized firms. *Human Resource Management*, 42(4): 321–35.
- Salas, E., Cannon-Bowers, J. A., Rhodenizer, L., and Bowers, C. A. (1999) Training in organizations. Research in Personnel and Human Resources Management, 17: 123-61.
- Scarbrough, H., and Kinnie, N. (2003) Barriers to the development of teamworking in UK firms. *Industrial Relations Journal*, 34(2): 135-49.
- Scase, R. (2003) Employment relations in small firms. In P. Edwards, ed. (2<sup>nd</sup> edn.) *Industrial relations*, 470-88, Oxford: Blackwell.
- Schuler, R.S. and Jackson, S.E. (1987) Linking competitive strategies with human resource management practices. *The Academy of Management EXECUTIVE*, 1(3): 207-19.
- Sels, L., Winne, S.D., Delmotte, J., Maes, J., Faems, D. and Forrier, A. (2006) Linking HRM and small business performance: an examination of the impact of HRM intensity on the productivity and financial performance of small businesses. *Small Business Economics*, 26: 83-101.
- Shen, J., Chanda, A., D'Netto, B. and Monga, M. (2009) Managing diversity through human resource management: an international perspective and conceptual framework. The International Journal of Human Resource Management, 20(2): 235-51.

- Sisson, K. (1993) In Search of HRM. British Journal of Industrial Relations, 31(2): 201-10.
- Smallbone, D., Cumbers, A., Syrett, S. and Leigh, R. (1999) The Single European Market and SMEs: A Comparison of its Effects in the Food and Clothing Sector in the UK and Portugal. *Regional Studies*, 33(1): 51-62.
- Small Business Service (2002) Small Business and Government The Way Forward. London: HMSO.
- Small Business Service (2003) Small and Medium Enterprise Definition. [Online] Available at: http://www.sbs.gov.uk/sbsgov/action/layer?topicId=7000000237andr.s=sl [Accessed 5<sup>th</sup> March 2007].
- Spalter-Roth, R. and Deitch, C. (1999) "I don't feel downsized; I feel out-of-work: gender, race, ethnicity, and the unequal costs of displacement. *Work and Occupations*, 26: 446-82.
- Storey, D. (1994) Understanding the Small Business Sector. London: International Thomson Business Press.
- Storey, D. (1998) Medium-size firms "waste money on training". *Education + Training*, 40(4): 163-65.
- Storey, D. (2002) Education, training and development policies and practices in medium-sized companies in the UK: do they really influence firm performance? *Omega*, 30: 249-64.
- Storey, D.J. and Westhead, P. (1997) Management training in small firms- a case of market failure? *Human Resource Management Journal*, 7(2): 61-71.
- Storey, D.J., Saridakis, G. Sen-Gupta. S., Edwards, P.K. and Blackburn, R.A. (2010) Linking HR formality with employee job quality: The role of firm and workplace size. *Human Resource Management*, 49(2): 309-25.
- Strauss, G. (1998) An overview. In F. Heller E. Pusic, G. Strauss and B. Wilpert, ed. Organisational Participation: Myth and Reality, 8-39, Oxford: Oxford University Press.
- Sundstrom, E., Demeuse, K.P., and Futrell, D. (1990) Work teams-applications and effectiveness. *American Psychologist*, 45: 120-33.
- Taylor, P. and Hirst, J. (2001) Facilitating effective change and continuous improvement: the mortgage express way. *Journal of Change Management*, 2(2): 67-71.

- Tharenou, P., Saks, A.M., and Moore, C. (2007) A review and critique of research on training and organizational-level outcomes. *Human Resource Management Review*, 17: 251-73.
- Vallas, S.P. (2003) Rediscovering the color line within work organizations: the 'knitting of racial groups' revisited. *Work and Occupations*, 30: 379-400.
- Wager, T.H. (1998, April) Determinants of human resource management practices in small firms: Some evidence from Atlantic Canada. *Journal of Small Business Management*, 13-23.
- Walters, G.D. (2007) Using Poisson class regression to analyze count data in correctional and forensic psychology: a relatively old solution to a relatively new problem. *Criminal Justice and Behavior*, 34: 1629-74.
- Walton, R. E. (1985) From control to commitment in the workplace. *Harvard Business Review*, 63(2): 77-84.
- Wall, T.D., Michie, J., Patterson, M., Wood, S.J., Sheehan, M., Clegg, C.W. and West, M. (2004) On the validity of subjective measures of company performance. *Personnel Psychology*, 57: 95-118.
- Wall, T.D. and Wood, S.J. (2005) The romance of human resource management and business performance, and the case for big science. *Human Relations*, 59(4): 429-62.
- Way, S. A. (2002) High performance work systems and intermediate indicators of firm performance within the US small business sector. *Journal of Management*, 28: 765-85.
- Welsh, J.A. and White, J.F. (1981) A small business is not a little big business. *Harvard Business Review*, July-August, 18-32.
- Westhead, P. and Storey, D. J. (1996) Management Training and Small Firm Performance: Why is the Link so Weak?. *International Small Business Journal*, 14(4): 13-24.
- Westhead, P. (1995) Survival and employment growth contrasts between types of owner-managed high-technology firms. *Entrepreneurship: Theory and Practice*, 20(1): 5-28.
- White, L. and Rogers, S. J. (2000) Economic circumstances and family outcomes: A review of the 1990s. *Journal of Marriage and the Family*, 62: 1035-51.
- Whitfield, K. and Poole, M. (1997) Organizing employment for high performance: theories, evidence and policy. *Organization Studies*, 18(5): 745-64.
- Wilkinson, A. (1999) Employment relations in SMEs. Employee Relations, 21(3): 206-16.

- Wilkinson, A. (2001) Empowerment. In T. Redman and A. Wilkinson, ed. *Contemporary Human Resource Management: text and cases*, 336-76, Financial Times: Prentice Hall.
- Wilkinson, A., Dundon, T. and Grugulis, I. (2007) Information but not consultation: exploring employee involvement in SMEs. *The International Journal of Human Resource Management*, 18(7): 1279-97.
- Wilkinson, A., Dundon, T., Marchington, M. and Ackers, P. (2004) Changing Patterns of Employee Voice. *Journal of Industrial Relations*, 46(3): 298-322.
- Williamson, I. O. (2000, Fall) Employer legitimacy and recruitment success in small businesses. *Entrepreneurship Theory and Practice*, 27-42.
- Wilson, E.M., and Iles, P.A. (1999) Managing Diversity An Employment and Service delivery Challenge. The International Journal of Public Sector Management, 12(1): 27-49.
- Wood S. (1999). Human resource management and performance. *International Journal of Management Reviews*, 1: 367-413.
- Wood, S. and de Menezes, L. (1998) High commitment management in the UK: evidence from the workplace industrial relations survey, and employers' manpower and skills practices survey. *Human Relations*, 51(4): 485-515.
- Wood, S. and de Menezes, L. (2008) Comparing perspectives on high involvement management and organizational performance across the British economy. *International Journal of Human Resource Management*, 19(4): 639-83.
- Wright, P.M. and Boswell, W.R. (2002) Desegregating HRM: a review and synthesis of micro and macro human resource management research. *Journal of Management*, 28(3): 247-76.
- Wright, P. M., Dunford, B. B. and Snell, S. A. (2001) Human resources and the resource based view of the firm. *Journal of Management*, 27(6): 701-21.
- Wright, P.M., Gardner, T.M., Moynihan, L.M. and Allen, M.R. (2005) The relationship between HR practices and firm performance: Examining causal order. *Personnel Psychology*, 58: 409-46.
- Wright, P.M, and Snell, S.A. (1991) Toward an integrative view of strategic human resource management. Human Resources Management Review, 1: 203-25.

- Wu, N. (2010) Antecedents of HRM practices in small and medium-sized firms. Included in the refereed proceedings of and presented at the 24th Association of Industrial Relations Academics of Australia and New Zealand (AIRAANZ) Conference, 3-5 February 2010, Sydney.
- Wu, N. (2009) Small and medium-sized firms: are HRM practices the same. Presented at the 2009 British Universities Industrial Relations Association (BUIRA) Annual Conference, 2-4 July 2009, Cardiff.
- Wu, N. (2009) The relationship between HRM and performance in small firms. Presented at the 5<sup>th</sup> International Conference on Performance and Reward, 1 April 2009, Manchester.
- Youndt, M.A., Snell, S.A., Dean, J.E. and Lepak, D.P. (1996) Human resource management, manufacturing strategy, and firm performance. *Academy of Management Journal*, 39(4): 836-66.
- Youndt, M. A. and Snell, S. A. (2004) Human resource configurations, intellectual capital, and organizational performance. *Journal of Managerial Issues*, 16(3): 337-60.

	Small firms	Medium firms	Large firms
HPWPs			
HPWP intensity	3.575	4.922	6.625
Sophisticated recruitment	.283	.296	.469
Induction	.356	.453	.604
Off-the-job training	.250	.510	.528
Internal labour market	.137	.314	.265
Performance-related pay	.214	.190	.394
Profit-related pay	.097	.206	.167
Performance appraisal	.474	.573	.759
Teamwork	.233	.278	.290
Team briefing	.211	.297	.24
Consulting committee	.041	.106	.12
Employee attitude survey	.172	.222	.55
Quality circle	.166	.071	.22
Functional flexibility	.201	.152	.25
Benefits	.171	.373	.62
Flexible working/family-friendly	.179	.235	.23
Equal opportunity	.096	.249	.24
Grievance procedures	.169	.332	
Job security	.126	.065	.11
HPWP bundles			
Ability bundle	.243	.375	
Motivation bundle	.223	.422	
Opportunity bundle	.082	.149	
Commitment bundle	.052	.139	.2
Performance measures	3.039	2.670	) 2.7
Absenteeism	2.522		
Labour turnover	3.644		_
Labour productivity	4.189		_
Quality of product or service	3.430		
Financial performance	3.430	, 3.02	-
<u>Industry sectors</u>	.199	.12	3 .6
Manufacture	.80	·	_
Non-manufacture	.00		
Degree of market competition	.67		
Investors in People	.110	626	.i.

	Small firms	Medium firms	Large firms
No largest customer	.011	.059	.044
Largest customer <5%	.365	.412	.510
Largest customer ≥5-25%	.302	.266	.185
Largest customer ≥26%	.323	.263	.26
Not a member of any employers' network	.482	.529	.405
Member of 1 employers' network	.388	.259	.374
Member of 2 employers' network	.130	.212	.22
<u>Firm size<sup>1</sup></u>			
5-9 employees	.523	.454	.40
10-24 employees	.390	.161	.29
25-49 employees	.087	.123	.13
50-99 employees	.000	.372	.00
100-149 employees	.000	.343	.00
150-249 employees	.000	.285	.00
250-1,999 employees	.000	.000	.34
2,000-9,999 employees	.000	.000	.29
10,000-49,999 employees	.000	.000	.21
50,000 employees	.000	.000	.14
<u>Firm age</u>	126	071	.08
0-4 years	.126	.071	.08
5-9 years	.181	.202	.23
10-19 years	.271 .422	.558	.52
20+ years	.422	.556	.52
Family ownership	.370	.488	.79
Not family owned	.072	.230	.11
Family owned/owner not involved	.558	.282	.08
Family owned/owner involved	.556	.202	
and the land one should be a	.842	.177	.00
Single independent workplace	.034	.057	.32
Union recognition HR specialist	.103	.153	33

Note:

The size categorisation for small firms (i.e. fewer than 50 employees) is based on workplace size. See justification for such a use in Chapter 3 (footnote 10 on page 62).